# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, DC 20549

## FORM 8-K

CURRENT REPORT
Pursuant to Section 13 or 15(d)
of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): September 11, 2017

## THE COOPER COMPANIES, INC.

(Exact name of registrant as specified in its charter)

Delaware (State or other jurisdiction of incorporation) 1-8597 (Commission File Number) 94-2657368 (IRS Employer Identification No.)

6140 Stoneridge Mall Road, Suite 590, Pleasanton, California 94588 (Address of principal executive offices)

(925) 460-3600 (Registrant's telephone number, including area code)

Check the appropriate box below if the Form 8-K is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:					
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)				
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)				
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))				
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c)				
	Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter). Emerging growth company				
	If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. $\Box$				

#### Item 1.01. Entry into a Material Definitive Agreement.

Purchase Aareement

On September 11, 2017, The Cooper Companies, Inc. (the "Company") announced that CooperSurgical, Inc. ("CooperSurgical), a wholly-owned subsidiary of the Company, entered into an Asset Purchase Agreement (the "Purchase Agreement") pursuant to which CooperSurgical will acquire the global rights and business of the Paragard Intrauterine Device (IUD) business of Teva Pharmaceutical Industries Limited ("Teva"), subject to the terms and conditions of the Purchase Agreement (the "Purchase").

CooperSurgical will pay \$1.1 billion in cash, subject to adjustment in accordance with an inventory adjustment mechanism.

The completion of the Purchase is contingent upon standard closing conditions, including regulatory approval in the United States and no material adverse change in the business to be acquired. The Company expects that the Purchase will be completed prior to the end of the current calendar year. The foregoing description of the Purchase Agreement is qualified in its entirety by reference to the full text of the Purchase Agreement which is filed hereto as Exhibit 2.1 and incorporated herein by reference.

In connection with the Purchase, the Company secured a financing commitment letter as further described in Item 7.01 below. Neither the Company nor any of its subsidiaries or respective affiliates has any material relationship with Teva or any of its affiliates other than in respect of the Purchase Agreement and the other ancillary agreements entered into in connection with the Purchase.

#### Item 7.01. Regulation FD Disclosure.

In connection with the Purchase, the Company has arranged a commitment for a bridge loan of \$1.1 billion through DNB Capital LLC and DNB Markets, Inc. (collectively, "DNB") to support funding of the acquisition. DNB's obligations under the commitment letter are subject to customary conditions. The Company anticipates using the bridge loan or alternative financing to fund all or a portion of the Purchase.

On September 11, 2017, the Company issued a press release announcing the signing of the Purchase Agreement and the bridge loan commitment with DNB. A copy of this release is attached hereto as Exhibit 99.1. The Company has also prepared an investor presentation with respect to the transaction, a copy of which is attached as Exhibit 99.2 hereto and incorporated by reference.

Internet addresses in the release are for information purposes only and are not intended to be hyperlinks to other The Cooper Companies, Inc. information.

## Item 9.01. Financial Statements and Exhibits.

## (d) Exhibits

No.	<u>Description</u>
2.1*	Asset Purchase Agreement, dated as of September 11, 2017, by and between CooperSurgical, Inc. and Teva Pharmaceutical Industries Ltd.
99.1	Press Release dated September 11, 2017, of The Cooper Companies, Inc.
99.2	Investor Presentation of The Cooper Companies, Inc.

\* Pursuant to Item 601(b)(2) of Regulation S-K, certain exhibits and schedules to this agreement have been omitted. The Company hereby agrees to furnish supplementally to the Securities and Exchange Commission, upon its request, any or all of such omitted exhibits or schedules.

## SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

THE COOPER COMPANIES, INC.

By: /s/ Albert G. White, III

Albert G. White, III Executive Vice President, Chief Financial Officer and Chief Strategy Officer

Dated: September 11, 2017

## EXHIBIT INDEX

No.	<u>Description</u>
2.1*	Asset Purchase Agreement, dated as of September 11, 2017, by and between CooperSurgical, Inc. and Teva Pharmaceutical Industries Ltd.
99.1	Press Release dated September 11, 2017, of The Cooper Companies, Inc.
00.2	Investor Presentation of The Cooper Companies Inc.

\* Pursuant to Item 601(b)(2) of Regulation S-K promulgated by the SEC, certain exhibits and schedules to this agreement have been omitted. The Company hereby agrees to furnish supplementally to the SEC, upon its request, any or all of such omitted exhibits or schedules.

## ASSET PURCHASE AGREEMENT

between

TEVA PHARMACEUTICAL INDUSTRIES LTD.

and

COOPERSURGICAL, INC.

Dated as of September 11, 2017

## TABLE OF CONTENTS

DTICLE 1 CALE	AND PURCHASE OF ASSETS, ASSUMPTION OF LIABILITIES	Page 1
Section 1.1	Sale and Purchase	1
Section 1.1	Closing	3
Section 1.3	Purchase Price	3
Section 1.3	Withholding	5
Section 1.5	Transfer Taxes and Other Costs	6
Section 1.6	Allocation of the Consideration	6
RTICLE 2 REPRE	ESENTATIONS AND WARRANTIES OF SELLER	7
Section 2.1	Organization	7
Section 2.2	Corporate and Governmental Authorization	7
Section 2.3	Non-Contravention	8
Section 2.4	Financial Statements	8
Section 2.5	Absence of Material Adverse Effect	8
Section 2.6	Material Contracts	8
Section 2.7	Title to Transferred Assets	9
Section 2.8	Real Property	9
Section 2.9	Intellectual Property	10
Section 2.10	Litigation	10
Section 2.11	Compliance with Laws; Licenses and Permits	10
Section 2.12	Environmental Matters	11
Section 2.13	Employees, Labor Matters, etc.	11
Section 2.14	Employee Benefit Plans and Related Matters	12
Section 2.15	Regulatory Matters	12
Section 2.16	Tax Matters	13
Section 2.17	Finders' Fees	13
Section 2.18	No Other Representations and Warranties	13
	ESENTATIONS AND WARRANTIES OF BUYER	14
Section 3.1	Corporate Status	14
Section 3.2	Corporate and Governmental Authorization	14
Section 3.3	Non-Contravention	14
Section 3.4	Sufficiency of Funds	15
Section 3.5	Solvency	15
Section 3.6	Litigation	15
Section 3.7	Finders' Fees	15
Section 3.8	Information Provided	15
Section 3.9	No Additional Representations; Inspection	15
	AIN COVENANTS	16
Section 4.1	Conduct of the Business	16
Section 4.2	No Undue Interference	16
Section 4.3	Access to Information; Confidentiality; Books and Records	17
Section 4.4	Governmental Approvals	18
Section 4.5	Shared Contracts	21
Section 4.6	Further Assurances	21
Section 4.7	Employees and Employee Benefits	21
Section 4.8	Supplemental Disclosure	23

i

Section 4.9	Publicity	23		
Section 4.10	Insurance	23		
Section 4.11	Assurances	24		
Section 4.12	Bulk Transfer Laws	24		
Section 4.13	Payments from Third Parties; Correspondence	24		
Section 4.14	Retained Trademarks	25		
Section 4.15	Mutual Non-Solicitation	25		
Section 4.16	Scope of Transition Services	25		
Section 4.17	Claims Administration	25		
Section 4.18	Financial Statements	27		
Section 4.19	Environmental Report	28		
Section 4.20	Trademarks and Domain Names	28		
ARTICLE 5 TAX MA	ATTERS	28		
Section 5.1	Tax Allocation	28		
Section 5.2	Miscellaneous	28		
ADDICE DA CONDI	TIONS DECEMBER	20		
	TIONS PRECEDENT	28		
Section 6.1	Conditions to Obligations of Buyer and Seller	28		
Section 6.2	Conditions to Obligations of Buyer	29 29		
Section 6.3	Conditions to Obligations of Seller	29		
ARTICLE 7 TERMINATION				
Section 7.1	Termination	30		
Section 7.2	Effect of Termination	30		
ARTICLE 8 INDEMNIFICATION				
Section 8.1	Survival	31 31		
Section 8.2	Indemnification by Seller	31		
Section 8.3	Indemnification by Buyer	31		
Section 8.4	Limitations on Indemnity	32		
Section 8.5	Notification of Claims; Third Party Claims	32		
Section 8.6	Exclusive Remedy	34		
·				
ARTICLE 9 DEFINI	Certain Terms	34 34		
Section 9.1 Section 9.2	Construction	43		
Section 9.2	Construction	43		
ARTICLE 10 MISCE		44		
Section 10.1	Notices	44		
Section 10.2	Amendment; Waivers, etc.	45		
Section 10.3	Expenses	45		
Section 10.4	Governing Law, etc.	45		
Section 10.5	Successors and Assigns	46		
Section 10.6	Entire Agreement	47		
Section 10.7	Severability	47		
Section 10.8	Counterparts; Effectiveness; Third Party Beneficiaries	47		
Section 10.9	Specific Performance	47		
Section 10.10	Waiver of Conflicts; Deal Communications	47		

Annex 9.1(a) **Transferred Assets** Annex 9.1(b) **Excluded Assets** Annex 9.1(c) Annex 9.1(d) Assumed Liabilities Excluded Liabilities

Exhibit A:

Form of Bill of Sale, Assignment and Assumption Form of Trademark Assignment Form of Transition Services Agreement Exhibit B: Exhibit C: Exhibit D: Form of Trademark License Agreement

#### ASSET PURCHASE AGREEMENT

This **ASSET PURCHASE AGREEMENT**, dated as of September 11, 2017 (this "**Agreement**"), is made by and between CooperSurgical, Inc., a Delaware corporation ("**Buyer**") and Teva Pharmaceutical Industries Ltd., a company organized under the laws of Israel ("**Seller**"). Capitalized terms used herein shall have the meanings assigned to such terms in the text of this Agreement or in <u>Section 9.1</u>.

#### RECITALS:

**WHEREAS**, Seller, directly and indirectly through certain of its Affiliates, owns, licenses or otherwise holds certain rights to manufacture, promote, market, distribute and/or sell the Paragard intrauterine copper contraceptive device, having the SKU set forth in <u>Schedule 1.3</u>, in the United States (the "**Products**" and such business in the corresponding territories, the "**Business**"); and

**WHEREAS**, Seller desires to sell (or cause to be sold), and Buyer desires to purchase or cause certain of its Affiliates to purchase, certain assets related to the Business and Buyer is willing to assume or cause certain of its Affiliates to assume certain liabilities related to the Business, in each case, upon the terms and subject to the conditions set forth herein.

**NOW, THEREFORE**, the parties agree as follows:

#### ARTICLE 1

#### SALE AND PURCHASE OF ASSETS, ASSUMPTION OF LIABILITIES

#### Section 1.1 Sale and Purchase.

- (a) <u>Transferred Assets</u>. Subject to the terms and conditions hereof, at the Closing, Seller will, and will cause the Selling Affiliates to, sell, convey, assign and transfer to Buyer, and Buyer will purchase, acquire and accept, as a going concern, the Transferred Assets, in each case free and clear of all Liens other than Permitted Liens.
- (b) Excluded Assets. Anything to the contrary herein notwithstanding, Buyer is not purchasing pursuant to this Agreement or any of the transactions contemplated hereby any right, title or interest in any Excluded Asset.
- (c) <u>Assumed Liabilities</u>. At the Closing, Buyer shall assume or cause certain of its Affiliates to assume the Assumed Liabilities and shall agree to satisfy and discharge when due the liabilities and obligations of Seller and its Affiliates in respect of the Assumed Liabilities. After the Closing, Buyer shall pay or cause certain of its Affiliates to pay all Assumed Liabilities as and when such Assumed Liabilities become due.
- (d) Excluded Liabilities. Neither Buyer nor any of its Affiliates shall assume or be obligated to pay, perform or otherwise discharge any Excluded Liability. Seller and/or its Affiliates, as the case may be, will remain liable to pay, perform and discharge when due, all Excluded Liabilities.
- (e) <u>Business Transfer Documents</u>. To the extent required under applicable Law or as reasonably deemed necessary by either of the parties hereto, to effect the transactions contemplated hereunder, the parties shall execute and deliver, or cause their respective Affiliates to execute and deliver, such asset and/or business transfer agreements, bills of sale, deeds (which, with respect to the Transferred

Real Property, will be a bargain and sale deed with covenant against grantor's acts), assignments, assumptions and other documents and instruments of sale, conveyance, assignment, novation, transfer and assumption (the "Business Transfer Documents") as are necessary to effect any transfer of the Transferred Assets or the Employees at the Closing or such other time for transfer as contemplated by Section 1.1(g) or Section 4.5 below or any assumption of the Assumed Liabilities at the Closing. The Business Transfer Documents shall be in form and substance reasonably agreed to by the parties and as is usual and customary in the applicable jurisdiction; provided that the parties agree and acknowledge that the Business Transfer Documents are intended solely to formalize the terms and conditions of this Agreement in order to comply with any applicable Law and shall be, in all respects, consistent with the terms and conditions set forth in this Agreement. In the event of any inconsistency between this Agreement and a Business Transfer Document, this Agreement shall control.

(f) <u>Designation of Affiliates</u>; <u>Performance of Obligations by Affiliates</u>. To the extent that any of the Transferred Assets are under the control of any of Seller's Affiliates, Seller shall cause such Affiliates to promptly take such legal action as may be necessary to consummate the transfer to Buyer of such Transferred Assets under terms and conditions which are consistent with and subject to the terms of this Agreement. Any obligation of Seller under or pursuant to this Agreement may be satisfied, met or fulfilled, in whole or in part, at Seller's sole and exclusive option, either by Seller directly, or by any Affiliate of Seller that Seller causes to satisfy, meet or fulfill such obligation, in whole or in part. Without otherwise limiting the foregoing, Seller shall cause Teva Pharmaceuticals USA, Inc. ("Teva USA") to transfer the Transferred Real Property to Buyer, and Teva Women's Health, LLC to transfer the Transferred Real Property Leases to Buyer. Notwithstanding the foregoing, this <u>Section 1.1(f)</u> shall not be construed to relieve Seller from any of its obligations under this Agreement.

(g) Transferred Assets Subject to Third-Party Consent. To the extent that the sale, conveyance, assignment or transfer or attempted sale, conveyance, assignment or transfer to Buyer of any Transferred Asset is prohibited by any applicable Law or would require any governmental or third-party consent, authorization, approval or waiver and such consent, authorization, approval or waiver shall not have been obtained prior to the Closing, this Agreement shall not constitute a sale, conveyance, assignment or transfer thereof. During the period commencing on the date of the Closing and continuing until 180 calendar days after the Closing, the parties shall use commercially reasonable efforts to cooperate with each other to obtain promptly such consents, authorizations, approvals or waivers; provided that neither Seller nor its Affiliates shall be required to pay any consideration or other amounts, or to commence, defend or participate in any Litigation or offer or grant any accommodation (financial or otherwise) to any third party, in connection with the foregoing. Unless otherwise provided in any Ancillary Agreement, until the earlier of (i) such 180 calendar day period and (ii) the date such consent, authorization, approval or waiver is obtained, Seller and Buyer shall enter into an arrangement whereby Buyer or its applicable Affiliates are provided the rights, benefits and obligations of such Transferred Asset and Buyer shall reimburse Seller for all costs and amounts incurred by Seller or any Selling Affiliate in connection with such arrangement, which arrangement may include Seller or a Selling Affiliate using Inventory purchased by Buyer hereunder to satisfy obligations of Seller or a Selling Affiliate under existing Contracts. If such consent, authorization, approval or waiver is obtained, Seller shall or shall cause the Selling Affiliates to assign, convey and transfer any such Transferred Asset to Buyer at no additional cost. If such consent, authorization, approval or waiver is not obtained notwithstanding the parties' compliance with the foregoing obligations, Seller will be deemed to have fulfilled its obligations under this Agreement and under no circumstances shall the Purchase Price be reduced or Seller or its Affiliates be subject to any liability on account of the failure to obtain any such consent, authorization, approval or waiver. Buyer further agrees that no representation, warranty or covenant of Seller contained in this Agreement shall be breached or deemed breached, and no condition to Buyer's obligations to close the transactions contemplated by this Agreement, other than the condition set out in Section 6.2(b), shall be deemed not satisfied as a result of (x) the failure to obtain any such

consent, authorization, approval or waiver or as a result of any resulting default or termination or (y) any lawsuit, action, claim, investigation or proceeding commenced or threatened by or on behalf of any Person arising out of or relating to the failure to obtain any consent, authorization, approval or waiver or any resulting default or termination. Notwithstanding anything herein to the contrary, Section 4.5 and not this Section 1.1(g) shall govern the transfer of any Shared Contracts.

(h) <u>Buyer's Recording and Similar Responsibilities</u>. Notwithstanding the foregoing provisions of this <u>Section 1.1</u>, it shall be Buyer's responsibility (i) to prepare the applicable Trademark assignments, which shall be in the form attached to this Agreement as <u>Exhibit B</u>, and to record such assignments following execution thereof by Seller or a Selling Affiliate at the Closing and (ii) to bear the fees and other costs in accordance with <u>Section 1.5</u> and any Taxes arising from such activities.

Section 1.2 <u>Closing</u>. The closing of the sale and purchase of the Transferred Assets and the assumption of the Assumed Liabilities (the "**Closing**") shall take place at 10:00 a.m. Eastern Standard Time through the execution and exchange, via .pdf copies of originally signed documents, of the documents and agreements contemplated herein on the date that is four (4) Business Days after the conditions set forth in <u>Article 6</u> have been satisfied or waived (other than those conditions that by their terms are to be satisfied at the Closing but subject to the satisfaction or waiver of those conditions at such time), unless another time, date or place is agreed to in writing by the parties. Buyer specifically acknowledges that time is of the essence because Seller's intention to exit the Business is or will become known to its employees, customers, suppliers and others having dealings with Seller. The date on which the Closing actually occurs is referred to hereinafter as the "**Closing Date**." At the Closing:

- (a) Seller shall deliver or cause to be delivered to Buyer (unless previously delivered), the following:
  - (i) the officer's certificate referred to in Section 6.2(a);
  - (ii) duly executed counterparts of the Ancillary Agreements; and
- (iii) a certificate of Teva USA, prepared in accordance with Treas. Reg. §1.1445-2(b)(2), certifying under penalties of perjury that Teva USA is not a "foreign person" within the meaning of Section 1445(f)(3) of the Code and the Treasury Regulations thereunder.
  - (b) Buyer shall deliver or cause to be delivered to Seller, or as designated by Seller, one or more of Seller's Affiliates (unless previously delivered), the following:
    - (i) the Purchase Price;
    - (ii) the officer's certificate referred to in Section 6.3(a); and
    - (iii) duly executed counterparts of the Ancillary Agreements.

#### Section 1.3 Purchase Price.

(a) <u>Purchase Price</u>. At the Closing, Buyer shall pay or cause to be paid to Seller, or Seller's applicable Subsidiary or assignee, by wire transfer of immediately available funds to one or more accounts designated by Seller at least four (4) Business Days prior to the Closing Date, an amount equal to the Purchase Price plus the Buyer Portion of PDUFA Fees.

#### (b) Purchase Price Adjustment.

- (i) Within five (5) Business Days following the Closing Date, Seller shall prepare and deliver to Buyer a statement (the "Inventory Statement") setting forth Seller's determination of the value of the Inventory transferred to Buyer as of the Closing (as of 11:59 P.M., local time, on the Closing Date) (the "Closing Inventory"), Seller's determination of Prepaid Inventory transferred to Buyer as of the Closing (as of 11:59 P.M., local time, on the Closing Date) and a calculation of the amount, if any, payable pursuant to clause (g) of this Section 1.3. In connection with the preparation of the Inventory Statement, Buyer shall (A) assist, and shall cause its Affiliates to assist, Seller, its accountants, advisors and other representatives, reasonable access during normal business hours to the personnel, properties, books and records of the Business to the extent relevant to the preparation of the Inventory Statement (including any taking and preparing with Buyer's participation of physical counts of Inventory). For purposes of this Section 1.3, the value of the Inventory transferred to Buyer will be determined by reference to the price for each Product and other type or item of Inventory as set forth on Schedule 1.3 to this Agreement shall include only Inventory that (A) has been approved and released by Seller or such applicable Selling Affiliate appointed qualified person in compliance with applicable Laws, (B) has an unexpired shelf life of not less than twelve (12) months, or such lesser period as may be agreed by Buyer and Seller on a Product-by-Product basis, (C) has not been damaged, recalled and/or incorrectly packaged or labeled and (D) consists of active pharmaceutical ingredients, spare parts, raw materials, containers, packaging and packaging supplies and work-in-process.
- (ii) For the purposes of clarification only, Seller is retaining all Pre-Closing Accounts Payable and Pre-Closing Accounts Receivable and the only Purchase Price adjustment after the Closing will be the adjustment of the Inventory and Prepaid Inventory pursuant to this <u>Section 1.3</u>.
- (c) <u>Dispute Notice</u>. The Inventory Statement shall become final, binding and conclusive upon Seller and Buyer on the 30<sup>th</sup> day following Buyer's receipt of the Inventory Statement (or the earlier date, if any, on which Buyer confirms to Seller in writing its agreement with the Inventory Statement), unless prior to such 30<sup>th</sup> day Buyer delivers to Seller a written notice (a "**Dispute Notice**") stating that Buyer believes the Inventory Statement contains mathematical errors (including quantity of Inventory transferred to Buyer or the amount of Prepaid Inventory), or was not prepared by reference to the price for each Product as set forth on <u>Schedule 1.3</u> to this Agreement or otherwise was not prepared in accordance with Section 1.3(b)(i) and specifying in reasonable detail each item that Buyer disputes (each, a "**Disputed Item**"), the amount in dispute for each Disputed Item and the reasons supporting Buyer's positions. Buyer shall not challenge the Inventory Statement on any other basis, and Buyer shall be deemed to have agreed with all other items and amounts contained in the Inventory Statement delivered pursuant to <u>Section 1.3(b)(i)</u>.
- (d) <u>Resolution Period</u>. If Buyer delivers a Dispute Notice, then Seller and Buyer shall seek in good faith to resolve the Disputed Items during the 15-calendar day period beginning on the date Seller receives the Dispute Notice (the "**Resolution Period**"). If Seller and Buyer reach agreement with respect to any Disputed Items, Seller shall revise the Inventory Statement to reflect such agreement.
- (e) <u>Independent Accountant</u>. If Buyer and Seller are unable to resolve all of the Disputed Items during the Resolution Period, then Buyer and Seller shall jointly engage and submit the unresolved Disputed Items (the "**Unresolved Items**") to the Independent Accountant; *provided* that if Buyer and Seller do not appoint an Independent Accountant within ten (10) calendar days after the end of the Resolution Period, they shall request the American Arbitration Association to appoint as the Independent Accountant a partner in the New York office of a nationally recognized independent

registered public accounting firm that has not had a material relationship with Buyer or Seller within the preceding two years, and such appointment shall be final, binding and conclusive on Buyer and Seller. The Independent Accountant shall act as an arbitrator to determine, based solely on presentations by Buyer and Seller and not by independent review, only the Unresolved Items still in dispute and shall be limited to those adjustments, if any, required to be made for the Inventory Statement to comply with the provisions of this Agreement. The parties shall agree, promptly after the Independent Accountant has been appointed, on procedures governing the resolution of any dispute by the Independent Accountant; *provided* that if the parties fail to agree on such procedures, the dispute resolution procedures of the American Arbitration Association shall govern. Buyer and Seller shall use their reasonable commercial efforts to cause the Independent Accountant to issue its written determination regarding the Unresolved Items within thirty (30) calendar days after such items are submitted for review. The Independent Accountant shall make a determination with respect to the Unresolved Items only and in a manner consistent with this Section 1.3(e) and the price for each Product as set forth on Schedule 1.3 to this Agreement, and in no event shall the Independent Accountant's determination of the Unresolved Items be for an amount that is outside the range of Buyer's and Seller's disagreement. Each party shall furnish to the Independent Accountant such work papers and other documents and information pertaining to the Unresolved Items as are in its possession or under its control as the Independent Accountant may reasonably request. The determination of the Independent Accountant shall be final, binding and conclusive upon Buyer and Seller absent manifest error, and Seller shall revise the Inventory Statement to reflect such determination upon receipt thereof. The fees, expenses and costs of the Independent Accountant

- (f) <u>Access to Information</u>. Each party shall use its reasonable best efforts to provide promptly to the other party all information as such other party shall reasonably request in connection with review of the Inventory Statement or the Dispute Notice, as the case may be, and shall otherwise cooperate in good faith with such other party to arrive at a final determination of the Inventory Statement.
  - (g) Final Adjustment. Within five (5) Business Days after the Inventory Statement is finalized pursuant to clause (c), (d) or (e) of this Section 1.3:
  - (i) if Closing Inventory is more than the product of (A) 105% *multiplied by* (B) the Inventory Target, then Buyer shall pay to Seller an amount equal to the Closing Inventory *minus* the Inventory Target; or
  - (ii) if Closing Inventory is less than the product of (A) 95% *multiplied by* (B) the Inventory Target, then Seller shall pay to Buyer an amount equal to the Inventory Target *minus* the Closing Inventory; *plus*
  - (iii) if the amount of Prepaid Inventory set forth on the Inventory Statement is greater than zero (0), then Buyer shall pay such amount of Prepaid Inventory to Seller.
- (h) Method of Payment. Any amount paid in respect of adjustments to the Purchase Price made pursuant to this Section 1.3 shall be (i) made by wire transfer of immediately available funds to an account designated by the receiving party and (ii) treated as an adjustment to the Purchase Price, including for tax reporting purposes.

Section 1.4 <u>Withholding</u>. Buyer shall make all payments under this Agreement free and clear of all deductions and withholdings in respect of Taxes, unless any such deduction or withholding is required by Law in effect at the time of payment. If any Taxes are required by Law in effect at the time of payment to be deducted from or in respect of any amount payable under this Agreement: (a) such amount payable shall be increased as necessary to ensure that, after all required deductions for Taxes are

made (including deductions applicable to any increases to any amount payable under this Section 1.4), Seller or any applicable Selling Affiliate receives the amount it would have received had no such deductions been made, (b) Buyer shall make such deductions and timely pay the full amount deducted to the Governmental Authority in accordance with applicable Law and (c) Buyer shall deliver to Seller an original or certified copy of a receipt evidencing such payment, except that clause (a) of this sentence shall not apply to any withholding required under Section 1445 of the Code or the rules and regulations promulgated thereunder to the extent any U.S. real property interests are allocated to a non-U.S. Selling Affiliate on Schedule 1.6. Notwithstanding the foregoing, Buyer will provide written notice to Seller promptly upon any determination that withholding may be required on any payments under this Agreement and each party shall take such steps as the other party may reasonably request and otherwise cooperate with the other party to reduce or eliminate such deduction or withholding.

#### Section 1.5 Transfer Taxes and Other Costs.

- (a) All Transfer Taxes payable on or in connection with the transfer of the Transferred Assets to Buyer and the transactions contemplated by this Agreement shall be borne and paid solely by Buyer when due in compliance with applicable Transfer Tax Laws; provided that if Seller is required by applicable Law to pay any Transfer Taxes, then Seller shall pay such Transfer Taxes, and Buyer shall, subject to receipt of reasonably satisfactory evidence of Seller's payment thereof, promptly reimburse Seller in dollars (or, at Seller's request, reimburse a Selling Affiliate in local currency), whether or not such Transfer Taxes were correctly or legally imposed by the applicable Governmental Authority.
- (b) Buyer shall prepare and file all necessary Tax Returns and other documentation required to be filed by it with respect to all Transfer Taxes, and, if required by applicable Law, the parties will, and will cause their Affiliates to, join in the execution of any such Tax Returns and other documentation.
- (c) The determination of whether any exemption from (or reduction in) Transfer Taxes is available with respect to the consummation of the transactions contemplated by this Agreement shall be made by Seller in its reasonable discretion. In making such determination, Seller shall take into account any duly completed exemption certificates delivered to it from Buyer no later than ten (10) calendar days prior to the Closing.
- (d) All costs and fees associated with transferring to Buyer or one of its Affiliates the Transferred IP, the Transferred IT and Regulatory Registrations for the Products conveyed to Buyer at the Closing shall be borne and paid solely by Buyer when due; *provided* that if any such amount shall be incurred by Seller, Buyer shall, subject to receipt of satisfactory evidence of Seller's payment thereof, promptly reimburse Seller.
- (e) All costs and expenses associated with removing and moving any Transferred Asset to a location designated by Buyer shall be borne and paid solely by Buyer when due; *provided* that if any such amount shall be incurred by Seller, Buyer shall, subject to receipt of satisfactory evidence of Seller's payment thereof, promptly reimburse Seller.

Section 1.6 <u>Allocation of the Consideration</u>. The parties have agreed that Buyer will prepare an initial allocation of the Purchase Price, the Assumed Liabilities and other relevant items among the Transferred Assets in accordance with the methodologies set forth in <u>Schedule 1.6</u>, for Seller's review, comment and approval. Buyer will accept all comments of Seller on the allocation of the Purchase Price, the Assumed Liabilities and other relevant items. In the event an adjustment to the Purchase Price is made pursuant to <u>Section 1.3</u> or otherwise under this Agreement (and any refunds and/or other payments

are made in connection therewith), the allocation of the Purchase Price shall be revised by Buyer (subject to the review, comment and approval of Seller) to allocate such adjustment to the Transferred Assets based upon the item to which such adjustment is attributable. The parties acknowledge and agree that (a) the Purchase Price, the Assumed Liabilities and other relevant items shall be allocated among the Selling Affiliates and across such assets or jurisdictions on Schedule 1.6 as Seller determines in its reasonable discretion, (b) except as required in connection with a determination within the meaning of Section 1313(a) of the Code (or equivalent provision of state, local, or foreign Law), (i) this allocation shall be binding on the parties for federal, state, local, foreign and other Tax reporting purposes, (ii) no party will assert or maintain a Tax position inconsistent with this allocation and (iii) the applicable Tax Returns to be filed by any of the parties or their subsidiaries shall reflect this allocation and (c) for financial accounting purposes, Buyer may allocate the Purchase Price, the Assumed Liabilities and other relevant items among the Transferred Assets in accordance with a valuation study performed by an independent third party engaged by Buyer.

#### ARTICLE 2

#### REPRESENTATIONS AND WARRANTIES OF SELLER

Except as set forth in the Seller Disclosure Letter, Seller represents and warrants to Buyer as follows:

Section 2.1 <u>Organization</u>. Each of Seller and the Selling Affiliates is duly organized, validly existing and in good standing (to the extent such concept is recognized in the relevant jurisdiction) under the Laws of its respective jurisdiction of formation, and has full corporate or other applicable legal power and authority to carry on its business in the places where such businesses are now being conducted.

#### Section 2.2 Corporate and Governmental Authorization.

- (a) Seller has full corporate power and authority to enter into this Agreement and each of the Selling Affiliates have, or will have at Closing, full corporate or other applicable legal power and authority to enter into the Ancillary Agreements to which it is to be a party and to perform its obligations hereunder and thereunder (as the case may be). This Agreement has been, and the Ancillary Agreements to which the Selling Affiliates are to be a party will be by Closing, duly authorized and approved by all necessary corporate action.
- (b) Assuming the due authorization, execution and delivery of this Agreement by Buyer, this Agreement constitutes a legal, valid and binding obligation of Seller, enforceable against Seller in accordance with its terms, except as enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditors' rights generally and by general equity principles. Assuming the due authorization, execution and delivery of the Ancillary Agreements by Buyer, each Ancillary Agreement to be executed by any Selling Affiliate, when delivered hereunder, will be duly and validly executed and delivered, and will constitute a legal, valid and binding obligation of such Selling Affiliate, enforceable in accordance with its terms, except as enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditors' rights generally and by general equity principles.
- (c) The execution and delivery of this Agreement by Seller and the execution and delivery of the Ancillary Agreements by any Selling Affiliate party thereto and the consummation of the transactions contemplated hereby or thereby require no action by or in respect of, or filing with, any Governmental Authority, other than (i) compliance with any applicable requirements of the HSR Act and the other Competition Laws of the jurisdictions set forth in Section 2.2(c)(i) of the Seller Disclosure

Letter, (ii) notice and transfer filings with the regulatory bodies set forth in <u>Section 2.2(c)(ii)</u> of the Seller Disclosure Letter and (iii) any actions or filings under Laws (other than Competition Laws) the absence of which would not reasonably be expected, individually or in the aggregate, to materially adversely affect the ability of Seller or any Selling Affiliate to perform its obligations hereunder or thereunder.

Section 2.3 Non-Contravention. Except as set forth on Section 2.3 of the Seller Disclosure Letter, the execution and delivery of this Agreement by Seller, the execution and delivery of the Ancillary Agreements by any Selling Affiliate party thereto and the consummation of the transactions contemplated hereby or thereby, do not and will not (a) violate in any material respect any provision of the Organizational Documents of Seller or any Selling Affiliate, (b) conflict with, result in the breach of, constitute a default under or result in the termination, cancellation or acceleration (whether after the giving of notice or the lapse of time or both) of any right or obligation of Seller or any Selling Affiliate under any Transferred Contract or Transferred Real Property Lease to which it is a party or to which its assets are subject, or result in the creation or the imposition of any Lien (other than a Permitted Lien) upon any of the Transferred Assets, or result in the cancellation, modification, revocation or suspension of any material authorization from any Governmental Authority in respect of the Transferred Assets or (c) assuming compliance with the matters set forth in Section 2.2(c), violate any Law of any Governmental Authority applicable to Seller or any Selling Affiliate, or any of the Transferred Assets or any Order against Seller or any Selling Affiliate or the Transferred Assets, in each case, in any material respect.

Section 2.4 <u>Financial Statements</u>. The financial statements set forth on <u>Section 2.4</u> of the Seller Disclosure Letter (the "**Financial Statements**") (a) have been prepared in good faith having regard to the purpose for which they were prepared, (b) to the Knowledge of Seller, do not materially misstate the historic revenues, and cost of goods sold attributable to the Business and (c) represent good faith estimates of an appropriate attribution of historic sales and marketing expenses, upon the assumptions and the basis of preparation set forth on <u>Section 2.4</u> of the Seller Disclosure Letter.

Section 2.5 <u>Absence of Material Adverse Effect</u>. Since December 31, 2016, to the date of this Agreement, the Business has been operated in the ordinary course and there have not been any events or occurrences that have, individually or in the aggregate, resulted in a Material Adverse Effect.

#### Section 2.6 Material Contracts.

- (a) Except for the Contracts disclosed in <u>Section 2.6</u> of the Seller Disclosure Letter and the Excluded Contracts, as of the date hereof, there are no Transferred Contracts which constitute:
  - (i) any joint venture, partnership, limited liability company or other similar agreements or arrangements (providing for joint research, development or marketing of any of the Products);
  - (ii) any agreement or series of related agreements, including any option agreement, relating to the acquisition or disposition of any business or assets of any other Person or any material real property (whether by merger, sale of stock, sale of assets or otherwise related to the Business);
  - (iii) any agreement that (A) limits the freedom of Seller or its Affiliates to operate the Business or with any Person or in any area or that would so limit the freedom of Buyer or its Affiliates after the Closing (other than customary exclusive distribution agreements for the Products) or (B) contains exclusivity obligations or restrictions binding on Seller or its Affiliates or that would be binding on Buyer or any of its Affiliates after the Closing;

- (iv) any agreement or series of related agreements for the purchase of materials, supplies, goods, services, equipment or other assets related to the Business that is reasonably expected to involve annual payments on the part of Seller or its Affiliates in excess of \$2,500,000 and is not terminable by Seller or such Affiliate on ninety (90) calendar days' notice or less without premium or penalty (excluding sales orders and purchase orders issued in the ordinary course of business);
- (v) any sales, distribution, agency or other similar agreement providing for the sale by the Business of Products, materials, supplies, goods, services, equipment or other assets which is reasonably expected to involve annual payments to Seller or its Affiliates over the remaining term of the agreement in excess of \$2,500,000 and is not terminable by Seller or such Affiliate on ninety (90) calendar days' notice or less without premium or penalty (excluding sales orders and purchase orders issued in the ordinary course of business); and
- (vi) any agreement under which the Business has (A) granted a Lien (other than a Permitted Lien) on any material Transferred Asset, other than a Lien that will be released as of the Closing or (B) provided for the sale of any material Transferred Asset, or granted any preferential rights to purchase any material Transferred Asset, in each case outside the ordinary course of business.
- (b) Each agreement, commitment, arrangement or plan disclosed in the Seller Disclosure Letter pursuant to this Section 2.6, Section 2.9 and the Transferred Real Property Leases (each, a "Material Contract") is a valid and binding agreement of Seller or the applicable Seller Affiliate party thereto, and, to the Knowledge of Seller, each other party thereto (subject to the effects of applicable bankruptcy, clarification, insolvency, fraudulent conveyance, moratorium, sponsorship or other Laws relating to or affecting creditors' rights generally and to general principles of equity, whether considered at law or in equity) and is in full force and effect, and neither Seller nor any applicable Seller Affiliate or, to the Knowledge of Seller, any other party thereto is in default or breach in any material respect under (or is alleged to be in default or breach in any material respect under) the terms of, or has provided or received any notice of any intention to terminate, any such Material Contract, and, to the Knowledge of Seller, no event or circumstance has occurred since September 11, 2015 that, with notice or lapse of time or both, would constitute an event of default thereunder or result in a termination thereof or would cause or permit the acceleration of or other changes of or to any right or obligation or the loss of any benefit thereunder that has not been cured or waived.

Section 2.7 <u>Title to Transferred Assets</u>. Subject to Permitted Liens, Seller or a Seller Affiliate has good title to, or valid leasehold interests in or leases of, all the material assets of the Business being transferred pursuant to this Agreement (in each case, other than Transferred Real Property and Transferred Real Property Leases, which are the subject of <u>Section 2.8</u>; IP Rights, which are the subject of <u>Section 2.9</u>; and Excluded Assets and assets disposed of since the date hereof in the ordinary course).

Section 2.8 Real Property. Section 2.8 of the Seller Disclosure Letter lists (i) the address and owner of the Transferred Real Property and any leases, subleases, licenses, concessions or other agreements, written or oral, granting to any party or parties the right of use or occupancy of any portion of the Transferred Real Property and (ii) the address, landlord and tenant for each Transferred Real Property Lease. Each applicable Seller Affiliate has good title to or, in the case of a Transferred Real Property Lease, valid leasehold interests in, all the Transferred Real Property and the Transferred Real Property Leases, free and clear of any Liens other than Permitted Liens.

#### Section 2.9 Intellectual Property.

- (a) Seller or a Seller Affiliate owns or has the right to use each item of Transferred IP free and clear of Liens other than Permitted Liens.
- (b) To the Knowledge of Seller, the operation of the Business, other than the use of any IP Rights covered by the Licensed IP Contracts, as currently conducted does not infringe upon or otherwise violate any of the IP Rights of any third Person.
  - (c) To the Knowledge of Seller, no third Person is infringing upon or otherwise violating the Transferred IP.
- (d) To the Knowledge of Seller, none of the Transferred IP is the subject of any written objection or claim with respect to the ownership, validity or enforceability thereof.
- (e) Except with respect to non-exclusive licenses granted to third parties in the ordinary course of business, agreements with distributors entered into in the ordinary course of business, "shrink-wrap", other generally available commercial licenses or Contracts otherwise included in Material Contracts, contemplated by this Agreement, Section 2.9(e) of the Seller Disclosure Letter lists, as of the date of this Agreement, all of the written agreements (i) pursuant to which Seller and its Affiliates obtained the right to use or practice rights under IP Rights owned by third Persons (excluding Copyright rights) that are material to the conduct of the Business or (ii) by which Seller or any of its Affiliates has licensed or otherwise authorized a third party to use any Patent or Trademark included in the Transferred IP (collectively, the "Licensed IP Contracts"), including license agreements, settlement agreements and covenants not to sue.
- Section 2.10 <u>Litigation</u>. As of the date hereof, except as set forth on <u>Section 2.10</u> of the Seller Disclosure Letter, (a) no Litigation relating to the Business is pending against or, to the Knowledge of Seller, threatened in writing against Seller or any Selling Affiliate and (b) neither Seller nor any Selling Affiliate, in connection with the Business, is subject to any outstanding Order.

#### Section 2.11 Compliance with Laws; Licenses and Permits.

- (a) Since January 1, 2016, Seller and the Selling Affiliates have conducted the Business in compliance in all material respects with applicable laws, statutes, ordinances, rules, regulations, judgments, injunctions, orders and decrees ("Laws") and, to the Knowledge of Seller, are not under investigation with respect to any violation of any Laws applicable to the Business.
- (b) Seller (together with the Selling Affiliates) has all material licenses, franchises, permits, certificates, approvals or other similar authorizations issued by applicable Governmental Authorities and affecting, or relating to the operation of the Business (the "Permits"). The Permits are valid and in full force and effect and none of the Permits will be terminated as a result of the transactions contemplated by this Agreement. No proceeding is pending or, to the Knowledge of Seller, threatened regarding the withdrawal, material modification or revocation of any such Permit. As of the date hereof, Seller has not received any written communication from any Governmental Authority threatening to withdraw, materially modify or suspend any Permit. Seller is not in violation of the terms of any Permit in any material respect.
- (c) Neither Seller nor any Selling Affiliate nor any representative of any of them has in relation to the Business, directly or indirectly: (i) made, offered to make or promised to make any payment or transfer of anything of value, directly or indirectly, to (A) anyone working in an official capacity for any Governmental Authority, including any employee of any government-owned or controlled entity or public international organization or (B) any political party, official of a political party

or candidate for political office, in order to obtain or retain business, or secure any improper business advantage, except for the payment of fees required by Law to be paid to Governmental Authorities, (ii) made any unreported political contribution, (iii) made or received any payment that was not legal to make or receive, (iv) engaged in any transaction or made or received any payment that was not properly recorded on its books, (v) created or used any "off-book" bank or cash account or "slush fund", or (vi) engaged in any conduct constituting a violation of the United States Foreign Corrupt Practices Act of 1977, as amended.

- (d) Except as set out on Section 2.11 of the Seller Disclosure Letter, neither the Specially Designated Nationals and Blocked Persons List maintained by the Office of Foreign Assets Control, Department of the Treasury ("OFAC") nor any similar list maintained by OFAC is applicable to Seller or any Selling Affiliate. The September 24, 2001 Executive Order Blocking Property and Prohibiting Transactions with Persons who Commit, Threaten to Commit, or Support Terrorism is also not applicable to any of those Persons.
- (e) Notwithstanding the foregoing, Seller makes no representation or warranty in this <u>Section 2.11</u> with respect to Litigation matters, environmental matters, employee benefit matters, regulatory matters or Tax matters, which matters are exclusively addressed in <u>Sections 2.10</u>, <u>2.12</u>, <u>2.14</u>, <u>2.15</u> and <u>2.16</u>, respectively.

#### Section 2.12 Environmental Matters.

- (a) Except as set forth on Section 2.12 of the Seller Disclosure Letter:
- (i) Seller and the Selling Affiliates, with respect to the Business, are in compliance in all material respects with all applicable Environmental Laws and are in possession of, and in compliance in all material respects with, all Permits required under applicable Environmental Laws;
- (ii) Seller and the Selling Affiliates, with respect to the Business, have not received from any Governmental Authority any written notice alleging a violation of or liability under any Environmental Law, except to the extent the substance of any such notice has been materially resolved; and
- (iii) as of the date hereof, no Litigation is pending or, to the Knowledge of Seller, threatened against Seller and the Selling Affiliates, with respect to the Business, arising under any Environmental Law.
- (b) Notwithstanding any of the representations and warranties contained elsewhere in this Agreement, the representations and warranties contained in this <u>Section 2.12</u> constitute the sole and exclusive representations made by Seller relating to environmental matters and matters arising under Environmental Laws.

#### Section 2.13 Employees, Labor Matters, etc..

(a) As of the date of this Agreement, Section 2.13(a) of the Seller Disclosure Letter sets forth an accurate and complete list of the Employees including identification number, titles, hire dates, annual base salary or hourly wage rate, as applicable, accrued vacation and paid-time-off, commission, bonus or other cash incentive opportunity, principal location, exemption status under applicable Law, notice period for termination, and visa status.

- (b) Section 2.13(b) of the Seller Disclosure Letter lists any Collective Bargaining Agreement that relates to the Employees. There is no pending or, to the Knowledge of Seller, threatened strike, slowdown, picketing or work stoppage by, or lockout of, or other similar labor activity or organizing campaign with respect to Seller and its Affiliates as it relates to any Employee as of the date hereof.
- (c) Seller and its Affiliates (with respect to the Business) are in material compliance with all applicable Laws as it relates to the Employees respecting labor, employment, fair employment practices, terms and conditions of employment, employee classification and wages and hours.

#### Section 2.14 Employee Benefit Plans and Related Matters.

- (a) <u>Section 2.14(a)</u> of the Seller Disclosure Letter lists, as of the date of this Agreement, each Business Employee Benefit Plan and each material employment agreement (excluding any offer letter or employment agreement required by applicable Law or any Collective Bargaining Agreement) in which any Employee of the Business is entitled to participate or to which he or she is a party (the "**Listed Plans**").
- (b) With respect to each Listed Plan that is an equity-based compensation plan in the United States, true and complete copies have been filed with the SEC as of the date of this Agreement. With respect to each Listed Plan that is a defined benefit pension or retiree medical benefit plan in the United States, a summary of material terms thereof has been made available in the Data Room as of the date of this Agreement. Except for Listed Plans as to which copies have been filed with the SEC, with respect to each Listed Plan, true and complete copies of all plan documents (including all amendments and modifications thereof) or a summary of material terms thereof, or a form or sample of each material employment agreement, have been made available in the Data Room as of the date of this Agreement.

#### Section 2.15 Regulatory Matters.

- (a) Except as set forth in Section 2.15(a) of the Seller Disclosure Letter, in the three (3) years prior to the date hereof, there has not been any product recall or market withdrawal or replacement conducted by or on behalf of Seller concerning the Products in the United States or any product recall, market withdrawal or replacement conducted by or on behalf of any third party as a result of any alleged defect in the Products in the United States, nor is any such product recall, market withdrawal or replacement currently under consideration. Seller has made available to Buyer copies of material complaints and notices of alleged defect or adverse reaction in the United States with respect to the Products that have been received in writing by Seller and its Affiliates in the 5 years prior to the date hereof.
- (b) In the three (3) years prior to the date hereof, to the Knowledge of Seller, the Products have been manufactured in compliance in all material respects with applicable Law, including cGMP, and applicable Regulatory Registrations. None of Seller or any of its Affiliates nor, to Knowledge of Seller, any third party engaged by Seller or any of its Affiliates in connection with the manufacture of the Products has received in the three (3) years prior to the date hereof, or as of the date hereof is subject to, a Warning Letter (as described in the FDA's Regulatory Procedures Manual) or has received a Form FDA 483 for which the inspection file remains open with respect to the manufacture of any Product.
- (c) Seller is not party to any corporate integrity agreements, monitoring agreements, consent decrees, settlement orders or similar agreements with or imposed by any Governmental Authority relating specifically to any part of the Business.

(d) During the two (2) years prior to the date hereof, none of Seller or any of its Affiliates or their respective employees or, to the Knowledge of Seller, any consultant who has undertaken activities on behalf of Seller or its Affiliates, has been debarred or deemed subject to debarment pursuant to Section 306 of the FDCA nor, to the Knowledge of Seller, (i) are any such Persons the subject of a conviction described in such section, (ii) have any such Persons been excluded from any government healthcare program including Medicare or Medicaid, or (iii) is any such debarment or exclusion proceeding pending.

#### Section 2.16 Tax Matters

- (a) All material Tax Returns that are required to be filed on or before the date of this Agreement in respect of the Transferred Assets have been filed (taking into account any applicable extensions). All such Tax Returns were correct and complete in all material respects and were prepared and filed in material compliance with all applicable Laws.
- (b) All material Taxes (other than Transfer Taxes) in respect of the Transferred Assets that are due and payable with respect to Pre-Closing Tax Periods have been paid, except for Taxes being contested in good faith through appropriate proceedings or Taxes for which adequate reserves have been established in the Financial Statements and Pre-Closing Accounts Payable.
  - (c) There are no Liens for Taxes upon any of the Transferred Assets, except for Permitted Liens.
- (d) The representations and warranties set forth in this <u>Section 2.16</u> are the sole representations and warranties relating to Taxes and no other representations or warranties contained in this <u>Agreement shall</u> be construed to cover any matter involving Taxes.
- Section 2.17 <u>Finders' Fees</u>. There is no investment banker, broker, finder or other intermediary retained by or authorized to act on behalf of Seller who might be entitled to any fee or commission from Buyer or any of its Affiliates upon consummation of the transactions contemplated by this Agreement.

Section 2.18 No Other Representations and Wartanties. EXCEPT AS EXPRESSLY SET FORTH IN THIS ARTICLE 2, (A) NONE OF SELLER OR ANY OF ITS AFFILIATES IS MAKING OR HAS MADE ANY REPRESENTATION OR WARRANTY, EXPRESS OR IMPLIED, AT LAW OR IN EQUITY, WITH RESPECT TO THIS AGREEMENT, THE ANCILLARY AGREEMENTS, SELLER, THE SELLING AFFILIATES, THE TRANSFERRED ASSETS, THE ASSUMED LIABILITIES, THE BUSINESS, THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT (INCLUDING ANY CONSENTS OR APPROVALS REQUIRED IN CONNECTION THEREWITH) OR ANY INFORMATION PROVIDED OR MADE AVAILABLE TO BUYER IN CONNECTION WITH THE TRANSACTIONS CONTEMPLATED BY THIS AGREEMENT (INCLUDING ANY FORECASTS, PROJECTIONS, ESTIMATES, BUDGETS, PRESENTATIONS CONCERNING THE BUSINESS (INCLUDING WITHOUT LIMITATION, THE CONFIDENTIAL INFORMATION MEMORANDA AND ANY "TEASER" DOCUMENTS), OR DUE DILIGENCE OR OTHER "DATA ROOM" MATERIALS), INCLUDING ANY WARRANTY WITH RESPECT TO MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE, AND ALL OTHER REPRESENTATIONS OR WARRANTIES ARE HEREBY EXPRESSLY DISCLAIMED AND SHALL NOT BE DEEMED TO BE OR TO INCLUDE REPRESENTATIONS OR WARRANTIES OF ANY OF THE FOREGOING PARTIES AND HAVE NOT BEEN RELIED UPON BY BUYER OR ANY OF ITS AFFILIATES IN EXECUTING, DELIVERING AND PERFORMING THIS AGREEMENT AND THE TRANSACTIONS CONTEMPLATED HEREBY; AND (B) EXCEPT AS IS OTHERWISE PROVIDED

IN THIS AGREEMENT, ALL OF THE ASSETS AND LIABILITIES TO BE SOLD, CONVEYED, ASSIGNED, TRANSFERRED OR ASSUMED, AS APPLICABLE, IN ACCORDANCE WITH THIS AGREEMENT, SHALL BE SOLD, CONVEYED, ASSIGNED, TRANSFERRED OR ASSUMED ON AN "AS IS, WHERE IS" BASIS.

#### ARTICLE 3

#### REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer represents and warrants to Seller as follows:

Section 3.1 Corporate Status. Buyer is a corporation duly organized, validly existing and in good standing under the laws of Delaware.

#### Section 3.2 Corporate and Governmental Authorization.

- (a) Buyer has full corporate power and authority to enter into this Agreement and the Ancillary Agreements, to perform its obligations hereunder and thereunder and to consummate the transactions contemplated hereby and thereby. This Agreement has been, and the Ancillary Agreements will be by Closing, duly authorized and approved by all necessary corporate action by Buyer. The performance of Buyer's obligations hereunder and thereunder and the consummation of the transactions contemplated hereby and thereby have been duly authorized by all requisite corporate action of Buyer. Buyer has duly executed and delivered this Agreement and on the Closing Date will have duly executed and delivered the Ancillary Agreements.
- (b) Assuming the due authorization, execution and delivery of this Agreement by Seller, this Agreement constitutes a legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with its terms, except as enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditors' rights generally and by general equity principles. Assuming the due authorization, execution and delivery of the Ancillary Agreements by Seller or such applicable Selling Affiliate, each Ancillary Agreement to be executed by Buyer, when delivered hereunder, will be duly and validly executed and delivered, and will constitute a legal, valid and binding obligation of Buyer, enforceable in accordance with its terms, except as enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or similar Laws affecting creditors' rights generally and by general equity principles.
- (c) The execution and delivery of this Agreement and the Ancillary Agreements by Buyer, and the consummation of the transactions contemplated hereby and thereby, require no action by or in respect of, or filing with, any Governmental Authority other than (i) compliance with any applicable requirements of the HSR Act and the Competition Laws of the jurisdictions set forth in Section 3.2(c)(i) of the Buyer Disclosure Letter, (ii) notice filings with the regulatory bodies set forth in Section 3.2(c)(ii) of the Buyer Disclosure Letter and (iii) any actions or filings under Laws (other than Competition Laws) the absence of which would not be, individually or in the aggregate, materially adverse to Buyer or materially impair the ability of Buyer to consummate the transactions contemplated hereby or thereby.

Section 3.3 Non-Contravention. The execution and delivery of this Agreement and the Ancillary Agreements by Buyer, and the performance of its obligations hereunder and thereunder do not (a) violate in any material respect any provision of the Organizational Documents of Buyer, (b) assuming compliance with the matters referred to in Section 3.2(c), violate any Law of any Governmental Authority applicable to Buyer or any Order against Buyer or (c) require any consent, authorization, approval, waiver or other action by any Person under any provision of any material agreement or other instrument to which Buyer is a party.

Section 3.4 <u>Sufficiency of Funds</u>. Buyer at Closing will have sufficient funds on hand to consummate the transactions contemplated by this Agreement and the Ancillary Agreements and to deliver the Purchase Price, any adjustment thereto and all fees and expenses related to the transactions contemplated by this Agreement and the Ancillary Agreements at Closing, and thereafter to pay and discharge the Assumed Liabilities and to perform its obligations under this Agreement and the Ancillary Agreements. Notwithstanding anything to the contrary contained herein, the parties acknowledge and agree that it shall not be a condition to the obligations of Buyer to consummate the transactions contemplated hereby that Buyer has sufficient funds for payment of the Purchase Price.

Section 3.5 <u>Solvency</u>. As of the date hereof and as of immediately after giving effect to the consummation of the transactions contemplated by this Agreement, Buyer and its Subsidiaries will be Solvent. For purposes of this <u>Section 3.5</u>, "**Solvent**" means, with respect to any Person, that:

- (a) the fair saleable value (determined on a going-concern basis) of the assets of such Person shall be greater than the total amount of such Person's liabilities (including all liabilities, whether or not reflected in a balance sheet prepared in accordance with GAAP and whether direct or indirect, fixed or contingent, secured or unsecured, disputed or undisputed);
  - (b) such Person shall be able to pay its debts and obligations in the ordinary course of business as they become due; and
  - (c) such Person shall have adequate capital to carry on its businesses and all businesses in which it is about to engage.

Section 3.6 <u>Litigation</u>. There is no Litigation pending against, or, to the knowledge of Buyer, threatened against or affecting, Buyer before any court or arbitrator or any Governmental Authority which in any manner challenges or seeks to prevent, enjoin, alter or materially delay the transactions contemplated by this Agreement or any Ancillary Agreement.

Section 3.7 <u>Finders' Fees</u>. There is no investment banker, broker, finder or other intermediary retained by or authorized to act on behalf of Buyer who might be entitled to any fee or commission from Seller or any of its Affiliates upon consummation of the transactions contemplated by this Agreement.

Section 3.8 <u>Information Provided</u>. All information provided by Buyer or on behalf of Buyer to Seller or any of its Affiliates or any of its attorneys for delivery to any Governmental Authority or in connection with any discussions, communications or analyses by or between Buyer, Seller or their attorneys related to Competition Laws was when given and remains as of the date hereof and will be, if given following the date hereof, on the date when given, complete and comprehensive (in the aggregate, with respect to the subject matter in question), true and correct in all material respects.

#### Section 3.9 No Additional Representations; Inspection.

(a) Buyer acknowledges and agrees that, except as expressly set forth in <u>Article 2</u>, (i) none of Seller or any of its Affiliates is making or has made any representation or warranty, express or implied, at law or in equity, with respect to this Agreement, the Ancillary Agreements, Seller, the Selling Affiliates, the Transferred Assets, the Assumed Liabilities, the Business, the transactions contemplated by this Agreement (including any consents or approvals required in connection therewith) or any information

provided or made available to Buyer in connection therewith (including any forecasts, projections, estimates, budgets, presentations concerning the business or due diligence or other "data room" materials), including any warranty with respect to merchantability or fitness for any particular purpose, and all other representations or warranties are hereby expressly disclaimed and shall not be deemed to be or to include representations or warranties of any of the foregoing parties and have not been relied upon by Buyer or any of its Affiliates in executing, delivering and performing this Agreement and the transactions contemplated hereby and (ii) all of the assets and liabilities to be sold, conveyed, assigned, transferred or assumed, as applicable, in accordance with this Agreement, shall be sold, conveyed, assigned, transferred or assumed on an "as is, where is" basis.

(b) Buyer acknowledges and agrees that it (i) has made its own inquiry and investigations into, and, based thereon, has formed an independent judgment concerning, the Transferred Assets, the Assumed Liabilities and the Business, (ii) has been provided with adequate access to such information, documents and other materials relating to the Transferred Assets, the Assumed Liabilities and the Business as it has deemed necessary to enable it to form such independent judgment, (iii) has had such time as Buyer deems necessary and appropriate to fully and completely review and analyze such information, documents and other materials and (iv) has been provided an opportunity to ask questions of Seller with respect to such information, documents and other materials and has received satisfactory answers to such questions. Buyer further acknowledges and agrees that none of Seller or any of its Affiliates has made any representations or warranties, express or implied, as to the accuracy or completeness of such information, documents and other materials other than the representations and warranties contained in this Agreement.

#### **ARTICLE 4**

#### CERTAIN COVENANTS

Section 4.1 <u>Conduct of the Business</u>. From the date hereof until the Closing, except as otherwise contemplated, permitted or required by this Agreement, the Ancillary Agreements or as set forth in <u>Schedule 4.1</u> or otherwise requested or consented to in writing by Buyer, which consent shall not be unreasonably conditioned, withheld or delayed, Seller agrees to (and to cause the Selling Affiliates to) use commercially reasonable efforts to conduct the Business in the ordinary course, and Seller shall not (and shall cause the Selling Affiliates not to):

- (a) sell, assign or transfer any of the Transferred Assets, except in the ordinary course of business (including the sale of Inventory or obsolete, worn-out or excess equipment or assets in the ordinary course of business);
- (b) amend in any material respect or terminate any Transferred Contract outside of the ordinary course of business, except for a Shared Contract pursuant to Section 4.5; or
  - (c) agree or commit to do any of the foregoing.

Section 4.2 <u>No Undue Interference</u>. From the date of this Agreement to the Closing Date, except as otherwise expressly permitted by this Agreement or consented to by Seller in writing, Buyer shall not interfere with the Transferred Assets, the Business or any Employee of the Business in any inappropriate or undue manner. Nothing contained in this Agreement shall give Buyer, directly or indirectly, the right to control or direct the operations of Seller, the Transferred Assets, the Business or any Employee prior to the Closing Date.

#### Section 4.3 Access to Information; Confidentiality; Books and Records.

- (a) From the date hereof until the Closing, subject to applicable Laws, Seller shall (i) give Buyer, its counsel, financial advisors, auditors and other authorized representatives reasonable access to the offices, properties, books and records of the Seller and the Selling Affiliates, (ii) furnish to Buyer, its counsel, financial advisors, auditors and other authorized representatives such financial and operating data and other information of the Seller and the Selling Affiliates as such Persons may reasonably request and (iii) instruct the employees, counsel and financial advisors of Seller and Seller's Affiliates to cooperate with Buyer, in each case, solely to the extent relating exclusively to the Business and in connection with Buyer's preparation to integrate the Business into Buyer's organization following the Closing; provided that this Section 4.3(a) shall not entitle Buyer or its accountants, legal counsel or other representatives to contact any third party doing business with Seller or any Seller Affiliate or access the properties, books or records of any such third party, in each case without Seller's prior written consent.
- (b) From and after the Closing, Buyer shall promptly afford Seller and Seller's respective agents reasonable access to the Transferred Records to the extent reasonably necessary for Seller in connection with any audit, investigation, dispute or Litigation; *provided* that Seller agrees to reimburse Buyer promptly for all reasonable and documented out-of-pocket costs and expenses incurred in connection with any such request.
- (c) Notwithstanding anything to the contrary in Section 4.3(a) and (b) (i) access rights pursuant to Section 4.3(a) and (b) shall be exercised in such manner as not to interfere unreasonably with the conduct of the Business or any other business of the party granting such access, (ii) the party granting access may withhold any document (or portions thereof) or information (A) that is subject to the terms of a non-disclosure agreement with a third party, (B) that may constitute privileged attorney-client communications or attorney work product and the transfer of which, or the provision of access to which, as reasonably determined by such party's counsel, constitutes a waiver of any such privilege or (C) if the provision of access to such document (or portion thereof) or information, as determined by such party's counsel, would reasonably be expected to conflict with applicable Laws and (iii) neither Seller nor any of its Affiliates or representatives shall have any obligation to provide Buyer or its representatives (A) access to any Tax Return filed by Seller or any of its Affiliates, or any related materials, in each case not relating exclusively to the Business, (B) access to any individual personnel or payroll records, in each case not relating exclusively to the Business or (C) access to the properties or assets of Seller or its Affiliates to conduct any subsurface or Phase II environmental investigation or sampling or testing of any building materials, indoor air or environmental medium.
- (d) All information provided to Buyer pursuant to this Section 4.3 prior to the Closing shall be held by Buyer as Confidential Information (as defined in the Confidentiality Agreement, dated as of May 24, 2017, between Seller and Buyer (the "Confidentiality Agreement")) and shall be subject to the Confidentiality Agreement, the terms of which are incorporated herein by reference. The Confidentiality Agreement shall continue in full force and effect until the Closing, at which time it shall automatically terminate. From and after the Closing: (i) Seller, on the one hand, and Buyer, on the other hand, shall, and shall cause their respective Affiliates and representatives to, maintain in confidence this Agreement and the Ancillary Agreements and any written, oral or other information related to the negotiation hereof and thereof, (ii) Seller shall, and shall cause its respective Affiliates and representatives to, maintain in confidence any written, oral or other information relating to the Business obtained by virtue of Seller's ownership of the Business prior to the Closing and (iii) Buyer shall, and shall cause its Affiliates and representatives to, maintain in confidence any written, oral or other information of or relating to Seller and its Affiliates (other than information relating to the Business) obtained by virtue of Buyer's ownership of the Business from and after the Closing, except, in each case, to the extent that the applicable party is required to disclose such information by judicial or administrative process or pursuant to applicable Law or the rules of any securities exchange to which that party is subject or such information can be shown to have been in the public domain through no fault of the applicable party.

Notwithstanding the foregoing, after the Closing, Buyer shall, and shall cause its Affiliates and representatives to, use commercially reasonable efforts to promptly (and in any event within thirty (30) calendar days after the Closing) remove, erase, delete or otherwise destroy all information of or relating to Seller and its Affiliates (other than information relating to the Business) (whether in print, electronic or other forms) in the possession of any Employee of which Buyer acquires possession or control as a result of the transactions contemplated by this Agreement, except that Buyer shall not be required to remove, erase, delete or otherwise destroy information stored in any automatic archiving or backup system.

(e) Subject to Section 4.3(d) Seller and its Affiliates shall have the right to retain copies of all books, data, files, information and records in any media (including, for the avoidance of doubt, Tax Returns and other information and documents relating to tax matters) of the Business relating to periods ending on or prior to the Closing Date (i) relating to information (including employment and medical records) regarding the Employees, (ii) as may be required by any Governmental Authority, including pursuant to any applicable Law or regulatory request or (iii) as may be necessary for Seller or its Affiliates to perform their respective obligations pursuant to this Agreement or any of the Ancillary Agreements, in each case subject to compliance with all applicable privacy Laws. Buyer agrees that, with respect to all original books, data, files, information and records of the Business existing as of the Closing Date, it will (x) comply in all material respects with all applicable Laws relating to the preservation and retention of records, (y) apply preservation and retention policies that are no less stringent than those generally applied by Buyer to its own books and records and (z) for at least ten years after the Closing Date, preserve and retain either (I) all such original books, data, files, information and records or (II) copies, which may be electronic copies, thereof, and thereafter dispose of its last remaining examples, which may be originals or copies, of such books, data, files, information and records only after it shall have given Seller ninety (90) calendar days' prior written notice of such disposition and the opportunity (at Seller's expense) to remove and retain such information.

#### Section 4.4 Governmental Approvals.

- (a) Seller and Buyer shall use reasonable best efforts to cooperate (i) to make, in the most expeditious manner practicable, all filings and applications with and to, and obtain all licenses, permits, consents, approvals, authorizations, qualifications and orders of, applicable Governmental Authorities to consummate the transactions contemplated by this Agreement and (ii) to obtain consents from other Persons, if any, listed on <a href="Schedule 4.4(a)">Schedule 4.4(a)</a>. In furtherance of the foregoing, Buyer agrees to provide such assurances as to financial capability, resources and creditworthiness as may be reasonably requested by any Governmental Authority or other Person whose consent or approval is sought hereunder.
- (b) In furtherance of the provisions set forth in Section 4.4(a), Seller and Buyer shall use reasonable best efforts to (A) file or cause to be filed as promptly as practicable, but in no event later than 10 Business Days following the execution and delivery of this Agreement, with the United States Federal Trade Commission (the "FTC"), the United States Department of Justice (the "DOJ") and any other Governmental Authority, in each case to the extent applicable, all notification and report forms that may be required for the transactions contemplated hereby and thereafter provide as promptly as practicable any supplemental information requested in connection therewith pursuant to the HSR Act or any other Competition Law and (B) include in each such filing, notification and report form referred to in the immediately preceding clause (A) a request for early termination or acceleration of any applicable waiting or review periods, to the extent available under the applicable Competition Law. Buyer shall pay all filing fees required for such filings, notifications and report forms. In connection therewith, Seller and Buyer shall:

- (i) furnish to the other party such necessary information and reasonable assistance as the other party may reasonably request in connection with its preparation of any filing or submission that is necessary under the HSR Act and other Competition Laws;
- (ii) subject to applicable Laws and a party's confidential information, including revenues for products for which NAICS codes are provided and Item 4 (c) and (d) documents, provide outside competition law counsel for the other party with a draft of any filing or submission and a reasonable opportunity to review such draft before making or causing to be made such filing or submission, and consider in good faith the views of outside competition law counsel for such other party regarding such filing or submission;
- (iii) not extend any applicable waiting or review periods or enter into any agreement with a Governmental Authority to delay or not to consummate the transactions contemplated hereby to be consummated on the Closing Date, except with the prior written consent of the other party;
- (iv) not have any substantive contact with any Governmental Authority in respect of any filing or proceeding contemplated by this Section 4.4(b) unless, to the extent practicable, they have engaged in prior consultation with the other party and, to the extent permitted by such Governmental Authority, given the other party the opportunity to participate; and
- (v) to the extent permitted by Laws, keep each other apprised of the status of any material communications with, and any inquiries or requests for additional information from, the FTC, the DOJ and any other applicable Governmental Authority.

Buyer further agrees to utilize its best efforts to take any and all actions to avoid or eliminate each and every impediment that may be asserted by any Governmental Authority with respect to the transactions contemplated by this Agreement so as to enable the Closing to occur as soon as reasonably possible, including:

- (vi) the prompt use of its best efforts to avoid the entry of, or to effect the dissolution of, any permanent, preliminary or temporary injunction or other order, decree, decision, determination or judgment that would delay, restrain, prevent, enjoin or otherwise prohibit consummation of the transactions contemplated by this Agreement, including:
  - (A) the defense through litigation on the merits of any claim asserted in any court, agency or other proceeding by any Person, including any Governmental Authority, seeking to delay, restrain, prevent, enjoin or otherwise prohibit consummation of such transactions;
  - (B) the proffer and agreement by Buyer, no later than sixty (60) calendar days from the date of this Agreement, of its willingness to sell, divest, lease, license or otherwise dispose of, or hold separate pending such disposition, and promptly to effect the sale, divestiture, lease, license, disposal and holding separate of (subject to the proviso set forth herein), any contraceptive and sterilization products, product lines or interests (including any complementary or ancillary products, product lines or interests related to contraceptive or sterilization products) therein of Buyer or any of its Subsidiaries that are not Transferred Assets (together, "Buyer C/S Interests") (and the entry into agreements with, and submission to orders of, the relevant Governmental Authority giving effect thereto); provided, however, that if the representations of Buyer

set forth in <u>Section 3.8</u> are inaccurate, this clause (B) shall apply instead to all assets, rights, product lines, categories of assets or businesses or other operations or interests of Buyer or any of its Subsidiaries that are not Transferred Assets (together, "**Buyer Interests**");

- (C) the proffer and agreement by Buyer of its willingness to take such other actions, and promptly to effect such other actions (and the entry into agreements with, and submission to orders of, the relevant Governmental Authority giving effect thereto), in each case with respect to Buyer C/S Interests or, if the representations of Buyer set forth in Section 3.8 are inaccurate, Buyer Interests, no later than sixty (60) calendar days from the date of this Agreement, in each case if such action should be reasonably necessary or advisable to avoid, prevent, eliminate or remove the actual, anticipated or threatened (x) commencement of any proceeding in any forum or (y) issuance of any order, decree, decision, determination or judgment that would delay, restrain, prevent, enjoin or otherwise prohibit consummation of the transactions contemplated by this Agreement by any Governmental Authority; and
- (vii) the prompt use of its best efforts to take, in the event that any permanent, preliminary or temporary injunction, decision, order, judgment, determination or decree is entered or issued, or becomes reasonably foreseeable to be entered or issued, in any proceeding or inquiry of any kind that would make consummation of the transactions contemplated by this Agreement in accordance with its terms unlawful or that would delay, restrain, prevent, enjoin or otherwise prohibit consummation of the transactions contemplated by this Agreement, any and all steps (including the appeal thereof and the posting of a bond) necessary to resist, vacate, modify, reverse, suspend, prevent, eliminate or remove such actual, anticipated or threatened injunction, decision, order, judgment, determination or decree so as to permit such consummation on a schedule as close as possible to that contemplated by this Agreement.

Nothing in this Section 4.4(b) shall obligate Buyer to agree to any divestitures or other remedy in each case not conditioned on the consummation of the Closing.

- (c) Buyer shall not, nor shall it permit its respective Subsidiaries and Affiliates to, acquire or agree to acquire any rights, assets, business, Person or division thereof (through acquisition, license, joint venture, collaboration or otherwise), if such acquisition, would reasonably be expected to delay or increase the risk of not obtaining any applicable clearance, consent, approval or waiver under antitrust or Competition Laws with respect to the transactions contemplated by this Agreement.
- (d) Seller shall pay, or cause to be paid, the PDUFA Fee, in full, on or before the earlier of (y) the date on which the PDUFA Fee is due under applicable Laws and (z) the Closing Date. As promptly as reasonably practicable following the Closing, Buyer and Seller shall: (i) file Transfer Letters with the applicable Governmental Authority to transfer ownership of the Regulatory Registration for the transferred Products, (ii) take such actions, if any, as are reasonably required, to ensure that the PDUFA Fee paid by Seller for the transferred Products continues to be credited to the transferred Products following the transfer of ownership of the Regulatory Registration for the transferred Products, and (iii) submit all appropriate and necessary documentation with respect to FDA approval by the applicable Governmental Authority of the removal of Trademarks of the Seller or its Affiliates and the inclusion of Buyer's name, corporate logo and National Drug Code on Product labeling. Transfer of title to any Regulatory Registration for the Products shall be effective as of the Closing. As promptly as reasonably practicable following the Closing, Buyer shall apply for and initiate applicable processes to obtain, establish and begin using its own National Drug Code for the Products, and shall promptly notify Seller thereof.

Section 4.5 Shared Contracts. Following the Closing, the parties desire for the benefit of Seller and Buyer, respectively, to have and obtain the rights and benefits under each Shared Contract to the extent related to the continuing business of Buyer and Seller and its Affiliates. The parties shall cooperate with each other to provide Seller and its Affiliates and Buyer and its Affiliates with their applicable rights and benefits under each Shared Contract *first*, by effecting a partial assignment of such Shared Contract or, if unable to effect such partial assignment, *second*, by assisting Buyer in entering into a new Contract or Contracts with the applicable third party on substantially similar terms (a "Separated Contract"). The costs of entering into a new Contract or Contract(s) shall be borne by Buyer. If any Shared Contract cannot be partially assigned or separated into a Separated Contract at Closing, Seller and Buyer shall, and shall cause each of their respective Affiliates to, use their reasonable commercial efforts to cause, for the period after the Closing until such Shared Contract is separated into a Separated Contract or such Separated Contract expires pursuant to its terms, (i) the rights and benefits under each Shared Contract to the extent relating to the Business to be borne by Buyer; (ii) the Liabilities under each Shared Contract to the extent relating to the Business to be borne by Seller; and (iv) the Liabilities under each Shared Contract to the extent relating to any business other than the Business to be borne by Seller.

Section 4.6 <u>Further Assurances</u>. From time to time after the Closing Date, at the request of another party, without further consideration and at the expense of the party so requesting, each of the parties shall execute and deliver to such requesting party, or shall cause to be executed and delivered to such requesting party, such additional instruments or documents, and shall take or cause to be taken such other action, as such requesting party may reasonably request in order to consummate more effectively the transactions contemplated by this Agreement.

#### Section 4.7 Employees and Employee Benefits.

- (a) Prior to, from and after the date of this Agreement until the Closing Date, Buyer shall consult with Seller and obtain Seller's express prior written consent, which Seller shall not unreasonably withhold or delay, before distributing any communications to any Employee whether relating to employee benefits, post-Closing terms of employment or otherwise. Buyer shall provide Seller with advance copies of, and a reasonable opportunity to comment on the form and content of, all such communications.
- (b) To the extent permitted by applicable Law, Seller shall periodically provide a list to Buyer prior to the Closing Date, which shall be included on Schedule 4.7(c), to reflect new hires, leaves of absence, employee transfers and employment terminations and any other material changes to Section 2.13 of the Seller Disclosure Letter and provide copies of such updated lists and information to Buyer.
- (c) Buyer shall provide offers of employment to all Employees listed on Schedule 4.7(c) upon the occurrence of the Closing. Offers pursuant to this Section 4.7(c) shall (i) be for a comparable position at the same or a nearby geographic work location with a commute of no more than thirty (30) miles from Employee's office immediately prior to the Closing Date, (ii) shall provide to each Employee, until December 31, 2018, (y) no less favorable base salary, wage rate and bonus opportunity, as applicable, than as set forth opposite such Employee's name on list provided pursuant to Section 4.7(b) and (z) employee benefits that are comparable to the benefits provided to Buyer's employees in comparable positions, including Buyer's medical plan and 401(k) plan and, for those employees who remain continuously in Buyer's service through the one (1) year anniversary of the Closing Date, Buyer's defined benefit pension plan, (iii) be sufficient to avoid statutory, contractual, common Law or other severance obligations, and (iv) otherwise comply in all respects with applicable Law (including with

respect to compensation and benefits). Buyer shall assume the liability to provide each Employee with the same amount of paid-time off as the Employee had immediately prior to Closing. Employee then shall accrue any further paid-time off in accordance with Buyer's policies. Buyer shall assume the liability to provide severance benefits to each Employee that are no less favorable than the better of (i) those severance or termination benefits applicable to such Employee as of immediately prior to the Closing Date and (ii) those provided under Buyer's severance plan, program, policy or practice (whether contractual or otherwise) on the date of such Employee's termination, in each case including all service with Seller and its Affiliates prior to the Closing Date as service with Buyer and its Affiliates for purpose of calculating entitlement to severance pay and termination benefits thereunder.

- (d) With respect to Employees, Buyer or its Affiliates shall (i) recognize, for all purposes (other than benefit accrual under a defined benefit pension plan) under all plans, programs and arrangements established or maintained by Buyer or its Affiliates for the benefit of the Employees, service with Seller and its Affiliates prior to the Closing Date to the extent such service was recognized under the corresponding Listed Plan covering such Employees, including for purposes of eligibility, vesting and benefit levels and accruals, in each case, except where it would result in a duplication of benefits and (ii) waive any pre-existing condition exclusion, actively-at-work requirement or waiting period under all employee health and other welfare benefit plans established or maintained by Buyer or its Affiliates for the benefit of the Employees, to the greatest extent permitted by those plans, except to the extent such pre-existing condition, exclusion, requirement or waiting period would have applied to such individual under the corresponding Listed Plan. Effective as of the Closing, each Employee shall cease to be an employee of Seller or the applicable Affiliate and shall cease to participate in any Business Employee Benefit Plan or other employee compensation, benefit or welfare plan of Seller or any of its Affiliates as an active employee.
- (e) Buyer shall cause a defined contribution plan with a Code Section 401(k) feature of which it is the plan sponsor to (i) cover all Employees as of immediately after the Closing Date, (ii) recognize service with Seller and its Affiliates prior to the Closing Date for all purposes and (iii) accept a transfer of assets representing the full balance standing to the credit of accounts of all Employees from Seller's (or its applicable Affiliate's) 401(k) plan, but not a transfer of any participant loans or related promissory notes under that plan, within sixty (60) calendar days of the Closing Date and shall cause the transfer to comply with the requirements of applicable Law, including without limitation Code Section 411(d)(6). No later than the date on which those assets are transferred from Seller's (or its applicable Affiliate's) 401(k) plan, Buyer shall offer, or cause to be offered, loans to the individuals listed on Schedule 4.7(e), in amounts not exceeding, for each individual, the lesser of (y) the then-current balance of that individual's loan from Seller's (or its applicable Affiliate's) 401(k) plan and (z) the outstanding balance of that loan set out in Schedule 4.7(e), on the further terms and conditions set out in this Section 4.7(e). Seller shall, no later than 10 Business Days before the asset transfer, notify Buyer in writing of the payment instructions for paying off the outstanding loans under Seller's (or its applicable Affiliate's) 401(k) plan, and Buyer shall disburse its loans to those payment instructions in satisfaction of the outstanding loans under Seller's (or its applicable Affiliate's) 401(k) plan. The interest rate payable under each of Buyer's loans shall be the interest rate set out in Schedule 4.7(e). The maturity date of each of Buyer's loans shall be the final payment date for the relevant individual set out in Schedule 4.7(e), except that Buyer may accelerate the maturity of each loan to the date that is 30 days after the date on which the relevant individual becomes eligible t
- (f) Buyer or its Affiliates shall bear all the liabilities, obligations and costs relating to the termination of Employees hired by Buyer, including any statutory or common law severance or other separation benefits, any contractual or other severance or separation benefits and any other legally mandated payment obligations (including any compensation payable during a mandatory termination

notice period and any payments pursuant to an Order of a court having jurisdiction over the parties). To the extent any Employee does not accept the offer of employment with Buyer, Seller shall provide the Employee with a severance and release agreement and bear the costs associated with the severance and termination.

- (g) In the United States, pursuant to IRS Revenue Procedure 2004-53, Buyer and Seller and their respective Affiliates shall apply the "standard" method for purposes of employee payroll reporting with respect to any Employee.
- (h) The provisions contained in this Agreement with respect to any Employee are included for the sole benefit of the respective parties hereto and shall not create any right in any other person, including any Employee (or dependent or beneficiary of any of the foregoing). Nothing herein shall be deemed an amendment of any plan providing benefits to any Employee.

Section 4.8 <u>Supplemental Disclosure</u>. Seller and Buyer shall have the right, from time to time prior to the Closing, to supplement or amend the Seller Disclosure Letter and the Buyer Disclosure Letter, as the case may be, with respect to any matter hereafter arising (other than as a result of a knowing breach of this Agreement by the party seeking to supplement or amend the Seller Disclosure Letter or the Buyer Disclosure Letter, as the case may be) that, if existing or known at the date of this Agreement, would have been required to be set forth or described in such Seller Disclosure Letter or Buyer Disclosure Letter, as the case may be. Any such supplemental or amended disclosure shall be deemed to have cured any breach of any representation or warranty made in this Agreement for purposes of determining whether or not the conditions set forth in <u>Article 6</u> have been satisfied, but not for purposes of indemnification set forth in <u>Article 8</u>.

Section 4.9 <u>Publicity</u>. No party to this Agreement shall originate any publicity, news release or other similar public announcement, written or oral, whether relating to this Agreement or any documents or transactions contemplated hereby or the existence of any arrangement between the parties, without the prior written consent of the other party (not to be unreasonably withheld) whether or not named in such publicity, news release or other similar public announcement, except (i) Seller and Buyer and their Affiliates may issue a press release in connection with the execution and delivery of this Agreement, and (ii) either party may originate any such publicity, news release or other similar public announcement as may be required by Law or any listing or trading agreement concerning its publicly traded securities; *provided* that in such event under <u>clauses (i)</u> and <u>(ii)</u>, the issuing party shall still be required to consult with the other party, whether or not named in such publicity, news release or other similar public announcement, a reasonable time prior to its release to allow the other party to comment thereon and, after its release, shall provide the other party with a copy thereof; *provided*, *further*, that if such non-issuing party shall not have provided its consent to the issuing party's request for consent and/or consultation pursuant to the foregoing after one (1) Business Day, the issuing party shall be deemed to have fulfilled its obligations pursuant to this provision and shall be free to issue such publicity, news release or other similar public announcement. If Buyer or Seller, based on the advice of its counsel, determines that this Agreement, or any of the Ancillary Agreements, must be publicly filed with a Governmental Authority, then the filing party, prior to making any such filing, shall provide the other party and its counsel with a redacted version of this Agreement (and any other Ancillary Agreement) which it intends to file, and will give due consideration to any comments provide

Section 4.10 <u>Insurance</u>. The coverage under all insurance policies related to the Business and arranged or maintained by Seller or its Affiliates is only for the benefit of Seller and its Affiliates, and not for the benefit of Buyer or the Business. As of the Closing Date, Buyer agrees to arrange for its own insurance policies with respect to the Business and agrees not to seek, through any means, to benefit from any of Seller's or its Affiliates' insurance policies which may provide coverage for claims relating in any way to the Business.

Section 4.11 <u>Assurances</u>. For a period of up to eighteen months after the Closing Date, if either Buyer or Seller becomes aware that any of the Transferred Assets has not been transferred to Buyer or that any of the Excluded Assets has been transferred to Buyer, it shall promptly notify the other and the parties hereto shall, as soon as reasonably practicable and, subject to <u>Section 1.1(e)</u>, ensure that such property is transferred, at Buyer's expense and with any necessary prior third-party consent or approval, to:

- (a) Buyer, in the case of any Transferred Asset which was not transferred to Buyer at the Closing; or
- (b) Seller, in the case of any Excluded Asset which was transferred to Buyer at the Closing.

Section 4.12 <u>Bulk Transfer Laws</u>. Buyer acknowledges that Seller and the Selling Affiliates have not taken, and do not intend to take, any action required to comply with any applicable bulk sale or bulk transfer Laws or similar Laws of any jurisdiction. Buyer hereby waives compliance by Seller and the Selling Affiliates with the provisions of any bulk sale or bulk transfer Laws or similar Laws of any jurisdiction in connection with the transactions contemplated by this Agreement.

#### Section 4.13 Payments from Third Parties; Correspondence.

- (a) Except as expressly provided in any Ancillary Agreement, in the event that, on or after the Closing Date, either party shall receive any payments or other funds due to the other pursuant to the terms of this Agreement or of any Ancillary Agreement, then the party receiving such funds shall promptly forward such funds to the proper party. The parties acknowledge and agree there is no right of offset regarding such payments and a party may not withhold funds received from third parties for the account of the other party in the event there is a dispute regarding any other issue under this Agreement or of any Ancillary Agreement. For purposes of this Agreement, any payments or other funds received by Buyer or any of its Affiliates from a third party in respect of any Product(s) shall first be applied against any outstanding undisputed receivable held by Seller or any applicable Selling Affiliate in respect of such Product(s) for any period prior to Closing and only after all such undisputed receivables have been satisfied shall any payments or other funds received by Buyer or any of its Affiliates from a third party in respect of any Product(s) be deemed in respect of receivables for any period after the Closing.
- (b) From and after the Closing until the one (1) year anniversary of the Closing Date, (i) Seller shall use commercially reasonable efforts to cause to be delivered to Buyer any mail or other communications received by Seller or any of its Affiliates from any Person in respect of any Product with respect to the period after the Closing and (ii) Buyer shall use commercially reasonable efforts to cause to be delivered to Seller or its applicable Selling Affiliate any mail or other communications received by Buyer or its Affiliates from any Person either intended for Seller or any of its Affiliates and not related to any Product or Transferred Asset or regarding any Product or Transferred Asset with respect to the period prior to the Closing. The provisions of this Section 4.13(b) are not intended to, and shall not be deemed to, constitute an authorization by any of Seller or Buyer to permit the other to accept service of process on its behalf and neither Seller nor Buyer are or shall be deemed to be the agent of the other for service of process purposes.

Section 4.14 <u>Retained Trademarks</u>. Except as expressly permitted in the Trademark License Agreement substantially in the form as attached as <u>Exhibit D</u> (the "**Trademark License Agreement**"), Buyer covenants that neither Buyer nor any of its Affiliates shall use in any manner any Trademark or trade dress of Seller or any of its Affiliates (other than the Trademarks included in the Transferred IP), including those set forth on <u>Schedule 4.14</u>.

Section 4.15 <u>Mutual Non-Solicitation</u>. From and after the Closing Date, Seller and Buyer each agree that they shall not, and they shall each cause their respective Subsidiaries and Affiliates to not, for a period of one (1) year after such Closing Date, directly or indirectly, encourage, induce, or solicit (a) in the case of Seller, any Employee to leave employment with Buyer or any of its Affiliates or (b) in the case of Buyer, any employee of Seller or any of its Affiliates who at any time provided services to the Business; provided, that this clause shall not preclude Seller, Buyer or any of their respective Subsidiaries or Affiliates from (a) posting a general solicitation through a public medium or general or mass mailing by or on behalf of Seller, Buyer or any of their respective Subsidiaries or Affiliates, as applicable, that is not targeted at employees of such other party or (b) soliciting any terminated employee of such other party so long as such former employee has been terminated from the employment with Seller, Buyer or any of their respective Subsidiaries or Affiliates, as applicable, for more than six (6) months.

Section 4.16 Scope of Transition Services. Prior to the Closing Date, Seller, Buyer and their respective Affiliates shall reasonably co-operate as to the separation of the Business from Seller's other business and planning Buyer's integration of the Business, so as to reduce the number and scope of services to be provided in the Transition Services Agreement substantially in the form as attached as Exhibit C (the "Transition Services Agreement"). Prior to the Closing Date, Seller, Buyer and their respective Affiliates shall complete the schedules to the Transition Services Agreement, which shall provide for a term not to exceed six (6) months following the Closing Date and, without limiting Buyer's rights under Section 2.2 of the Transition Services Agreement with respect to any Omitted Service (as such term is defined in the Transition Services Agreement), shall be limited to commercial, regulatory, pharmacovigilance/quality and corporate functions and services in a manner and scope of such services and functions that had been provided by the Seller or a Seller Affiliate to the Business in the 12 month period prior to the Closing Date but are not (a) provided by Employees hired by Buyer, (b) provided by independent contractors under Transferred Contracts or (b) included in the Transferred Assets. Without limiting the generality of the immediately preceding sentence, if Seller has not filed with the FDA the annual report for the Product under 21 C.F.R. 314.81(b)(ii), the services to be provided under the Transition Services Agreement will include services related to the preparation of that annual report. Notwithstanding the foregoing to the contrary, nothing in this Section 4.16 shall require or permit Seller, Buyer or their Affiliates to conduct any integration activities to the extent that the same are prohibited by Competition Laws.

Section 4.17 <u>Claims Administration</u>. Buyer shall, at Seller's request, provide claim administration services with respect to Seller Product Claims ("Claim Services"), on the terms and conditions set out in this <u>Section 4.17</u>. "Seller Product Claims" means lawsuits or other claims, regardless of when commenced or made and irrespective of the legal theory asserted, arising from or relating to the design, manufacture, testing, advertising, marketing, distribution or sale of any unit of Product inserted in a patient on or before the Closing Date, including all liabilities and obligations arising from or relating to (x) warranty obligations, (y) infringement, dilution, misappropriation or other violation of IP Rights, or (z) alleged or actual hazard or defect in design, manufacture, materials or workmanship, including any failure to warn or alleged or actual breach of express or implied warranty or representation, in each case with respect to a unit of Product inserted in a patient on or before the Closing Date.

- (a) If Seller desires Claim Services for a Seller Product Claim, Seller shall request those Claim Services by written notice to Buyer (a "Service Request"). Seller shall in its Service Request designate a primary contact person at Seller for Claim Services related to that Seller Product Claim. Buyer and Seller each may, from time to time, by written notice to the other, designate contact persons for Service Requests and other communications concerning Claim Services who differ from their contact persons for other notices under this Agreement.
- (b) The Claim Services for each Seller Product Claim shall, as and to the extent requested by the Seller, include the following services, each to the extent commercially reasonable:
  - (i) Commercial management, coordination and oversight services related to the administration, evaluation and defense of the Seller Product Claim, including working with Seller or its Affiliates, Seller's insurers (if any), and any Advisors to (A) review the Seller Product Claim, (B) evaluate the Seller Product Claim, (C) develop a strategy for responding to the Seller Product Claim, including through negotiation, mediation, litigation and arbitration, (D) execute that strategy and (E) reevaluate and modify that strategy as appropriate; and/or
  - (ii) Support from Buyer personnel with regulatory and technical expertise concerning the Product in connection with the evaluation and defense of the Seller Product Claim to include cooperation in claims or lawsuits including reviewing and responding to allegations in pleadings, motions, or briefs; response to document and other discovery requests; preparation for and appearance at depositions, hearings or trial; consultation with expert witnesses and other claim or litigation related activities as Buyer believes are reasonably necessary to evaluate and defend Seller Product Claims.
- (c) "Advisors" means consultants and legal, accounting and other professional advisors, including expert witnesses, involved in the response to a Seller Product Claim or otherwise involved in the Claim Services. Seller or its insurers may designate Advisors. Buyer may, and shall at Seller's request, designate Advisors, provided that the services provided by Advisors designated by Buyer on its own initiative do not unreasonably duplicate services provided by the Advisors designated by Seller or its insurers. To help avoid duplication of services, Buyer shall notify Seller prior to engaging Advisors.
- (d) Each Service Request shall describe in reasonable detail, to the extent known by Seller at the time of delivery of the Service Request, the Seller Product Claim for which Seller is requesting Claim Services and the Claim Services requested by Seller. Seller may, by written notice to Buyer from time to time, discontinue or change the scope of the Claim Services that Buyer is providing with respect to any Seller Product Claim.
- (e) The Claim Services shall be subject to the reasonable oversight and direction of Seller in all respects, subject to compliance with applicable law and regulation. Seller shall have the right to approve, or establish limitations on, the settlement of each Seller Product Claim in Seller's discretion.
- (f) Seller shall reimburse Buyer for all documented out-of-pocket expenses incurred by Buyer in the course of providing the Claim Services, including all fees and expenses of all Advisors. Buyer will invoice Seller monthly for those expenses, and Seller shall pay those expenses to Buyer, by wire transfer of immediately available U.S. dollars or other means acceptable to Buyer, within 30 days of the date of Buyer's invoice.
- (g) The Claim Services may, to the extent requested by Seller, include commercial oversight of providers of legal, accounting or other professional services, but the Claim Services will not include legal, accounting or other professional services.

- (h) Buyer shall make personnel or Advisors with the requisite expertise reasonably available to provide the Claim Services.
- (i) Seller shall cooperate reasonably with Buyer's performance of the Claim Services, including promptly informing Buyer of developments in the Seller Product Claims for which Buyer is providing Claim Services and providing relevant information reasonably requested by Buyer.
- (j) Seller Product Claims are Excluded Liabilities, for which neither Buyer nor its Affiliates shall have any liability or obligation, and nothing in this <u>Section 4.17</u>, nor the performance of any Claim Services by Buyer or its Affiliates, shall result in the imposition of any liability or obligation on Buyer or its Affiliates arising from, concerning or in connection with any Seller Product Claim. Buyer's indemnification rights under <u>Section 8.2</u> shall apply to all Seller Product Claims.
- (k) BUYER CANNOT GIVE ANY ASSURANCE CONCERNING THE OUTCOME OF ANY SELLER PRODUCT CLAIM, AND SELLER HEREBY ASSUMES ALL RESPONSIBILITY FOR ANY SUCH OUTCOME. BUYER HEREBY DISCLAIMS ALL REPRESENTATIONS AND WARRANTIES CONCERNING THE CLAIM SERVICES, AND BUYER SHALL HAVE NO LIABILITY IN CONNECTION WITH ANY SELLER PRODUCT CLAIM OR THE OUTCOME OF ANY SELLER PRODUCT CLAIM.
- (l) Seller shall defend, indemnify and hold harmless the Buyer Indemnitees from and against, and pay or reimburse the Buyer Indemnitees for, all Losses arising from or relating to the Claim Services.

Section 4.18 Financial Statements. Seller shall prepare and deliver to Buyer unaudited financial information of the Business, including income statements for the relevant period and balance sheets as of the last day of the relevant period, prepared on a basis consistent with the assumptions and basis of presentation of the special purpose financial statements for the Business for the fiscal years ended December 31, 2015 and 2016 that were audited and delivered to Buyer on September 1, 2017 (the "Audited Financials"), for (1) each of Seller's 2017 fiscal quarters that ended prior to the Closing Date (the "Quarterly Financials") and (2) the period beginning on the day after the last day of the immediately preceding fiscal quarter and ending on the Closing Date (the "Stub Period Financials"). Seller shall deliver the Quarterly Financials no later than January 31, 2018; provided, however, that if the Closing Date is on or after December 31, 2017, Seller shall deliver the financial information for its fiscal quarter ended December 31, 2017 no later than sixty (60) days after the Closing Date is on or before October 31, 2017, no later than December 11, 2017, or (z) if the Closing Date is after October 31, 2017, no later than sixty (60) days after the Closing Date is on or before October 31, 2017, no later than December 11, 2017, or (z) if the Closing Date is after October 31, 2017, no later than sixty (60) days after the Closing Date. Subject to the other provisions of this Section 4.18. Seller shall provide such financial statements and information (audited or unaudited, as the case may be) related to the pre-Closing Business, at such times, as Buyer reasonably requires to timely comply with its obligations under Rules 3-05 and 1-02(w) of Regulation S-X and cooperate reasonably with Buyer in connection with those statements and information. Buyer shall reimburse Seller and its Affiliates for all documented out-of-pocket expenses incurred by Seller and its Affiliates in connection with fulfilling their obligations in

Section 4.19 Environmental Report. Seller shall, on or before the Closing Date, authorize ERM Consulting & Engineering, Inc. ("ERM") to permit Buyer to rely on the Phase I Environmental Site Assessment of the Owned Real Property, dated August 30, 2017, prepared by ERM, and hereby authorizes Buyer to make such arrangements with ERM as are required for that reliance, at Buyer's expense.

Section 4.20 <u>Trademarks and Domain Names</u>. Without limiting the generality of Seller's obligations under <u>Section 4.1</u>, from the date of this Agreement through the Closing Date, Seller shall maintain the Trademarks and domain names included in the Transferred Assets and pay all associated fees and expenses. Seller shall, on or before the Closing Date, to the extent it has not already done so, for each Trademark included in the Transferred Assets that is owned by Teva Women's Health LLC, duly record with the applicable Governmental Authorities with which that Trademark is registered the change in that entity's name resulting from its conversion from a corporation to a limited liability company.

### **ARTICLE 5**

## TAX MATTERS

Section 5.1 Tax Allocation. In the case of any Taxes that are imposed on a periodic basis and are payable for a Straddle Period, the portion of such Tax that relates to the portion of the Straddle Period ending on the Closing Date shall (a) in the case of any Taxes other than Taxes based upon or related to income, gains or receipts, sales or use, employment, withholding or similar items, be deemed to be the amount of such Tax for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of calendar days from the beginning of such Straddle Period through the Closing Date and the denominator of which is the number of calendar days in the entire Straddle Period, and (b) in the case of any Tax based upon or related to income, gains or receipts, sales or use, employment, withholding or similar items, be deemed equal to the amount of Tax which would be payable if the relevant Tax period ended on the Closing Date. Any credits relating to a Straddle Period shall be taken into account as though the relevant Tax period ended on the Closing Date. All determinations necessary to give effect to the foregoing allocations shall be made in a manner consistent with past practice. Any Tax refunds or credits for overpayment for a Straddle Period shall be determined in a manner consistent with this Section 5.1, and any such amounts that relate to a Pre-Closing Tax Period (or are otherwise described in Annex 9.1(b)(x)) shall be for the account of Seller. Buyer shall, or shall cause its Affiliates, to promptly remit to Seller the amount of any such refund or credit (or any other amounts described in Annex 9.1(b)(x)).

Section 5.2 <u>Miscellaneous</u>. Each of Buyer and Seller shall provide the other with such information and records, and make such of its officers, directors, employees and agents available, as may reasonably be requested by such other party in connection with the preparation of any Tax Return or the conduct of any Tax Proceeding of the Transferred Assets or the Business for any Pre-Closing Tax Period or a Straddle Period.

#### ARTICLE 6

## CONDITIONS PRECEDENT

Section 6.1 <u>Conditions to Obligations of Buyer and Seller</u>. The obligations of Buyer and Seller to consummate the transactions contemplated hereby shall be subject to the fulfillment at or prior to the Closing of the following conditions, which conditions shall not be capable of waiver:

- (a) No Injunction, etc.. Consummation of the transactions contemplated hereby shall not have been materially restrained, enjoined or otherwise prohibited by an Order of any Governmental Authority or made illegal by any applicable Law.
- (b) <u>Governmental Approvals</u>. (i) All waiting periods (and extensions thereof) applicable to the consummation of the transactions contemplated by this Agreement under applicable Competition Laws, as listed on <u>Schedule 6.1(b)(i)</u> of this Agreement, shall have expired or been terminated and (ii) all authorizations, clearances, consents and approvals required to be obtained under applicable Competition Laws or otherwise from any applicable Governmental Authority to permit the parties to consummate the transactions contemplated by this Agreement, as listed on <u>Schedule 6.1(b)(ii)</u> of this Agreement, shall have been obtained and be in full force and effect.
- Section 6.2 <u>Conditions to Obligations of Buyer</u>. The obligation of Buyer to consummate the transactions contemplated hereby shall be subject to the fulfillment at or prior to the Closing of the following additional conditions:
- (a) Representations; Performance. The representations and warranties of Seller contained in Article 2 of this Agreement (other than the Fundamental Representations) (without giving effect to any limitations as to "materiality" or "Material Adverse Effect" set forth therein) shall be true and correct at and as of the Closing Date with the same effect as though made at and as of such time (except for representations that are as of a specific date which representations shall be true and correct in all material respects as of such date), except where all failures to be so true and correct would not reasonably be expected, individually or in the aggregate, to have a Material Adverse Effect. Each of the Fundamental Representations shall be true and correct in all material respects at and as of the Closing Date with the same effect as though made at and as of such time (other than representations and warranties made as of a specified date, which shall be true and correct as of the date specified). Seller shall have in all material respects duly performed and complied with all agreements, covenants and conditions required by this Agreement to be performed or complied with by Seller at or prior to the Closing. Seller shall have delivered to Buyer a certificate, dated as of the Closing Date, signed by a duly authorized officer of Seller to the effect set forth above in this Section 6.2(a).
  - (b) the Persons listed on Schedule 6.2(b) shall have consented to the assignment to Buyer of the agreements listed in Schedule 6.2(b);
  - (c) Since the date of this Agreement, there has been no event or occurrence that would have a Material Adverse Effect.
  - (d) The actions set forth in Section 1.2(a) shall have been completed or will be completed at the Closing.
- Section 6.3 <u>Conditions to Obligations of Seller</u>. The obligation of Seller to consummate the transactions contemplated hereby shall be subject to the fulfillment at or prior to the Closing of the following additional conditions:
- (a) <u>Representations</u>; <u>Performance</u>. The representations and warranties of Buyer contained in this Agreement and in any certificate or other writing delivered pursuant hereto shall be true and correct in all material respects at and as of the Closing Date with the same effect as though made at and as of such time. Buyer shall have in all material respects duly performed and complied with all agreements, covenants and conditions required by this Agreement to be performed or complied with by Buyer at or prior to the Closing. Buyer shall have delivered to Seller a certificate, dated as of the Closing Date, signed by a duly authorized officer of Buyer to the effect set forth above in this <u>Section 6.3(a)</u>.
  - (b) The actions set forth in Section 1.2(b) shall have been completed or will be completed at such Closing.

## ARTICLE 7

## TERMINATION

- Section 7.1 Termination. This Agreement may be terminated at any time prior to the Closing Date:
  - (a) by the written agreement of Buyer and Seller;
  - (b) by either Buyer or Seller by notice to the other party, if:
- (i) the Closing shall not have been consummated on or before January 1, 2018 (the "End Date"); provided, that in the event the condition set forth in Section 6.1(b) shall not have been satisfied on or prior to the End Date, but all other conditions set forth in Article 6 shall have been satisfied, or are capable of being satisfied (or have been waived by the party then entitled to give such waiver) on or prior to the End Date, then the End Date shall be extended without further action by the parties hereto to January 31, 2018; provided, further, that the right to terminate this Agreement pursuant to this Section 7.1(b)(i) shall not be available to any party whose breach of any provision of this Agreement results in the failure of the Closing to be consummated by such time; or
- (ii) (A) there shall be any Law that makes consummation of the Closing illegal or otherwise prohibited or (B) any judgment, injunction, order or decree of any Governmental Authority having competent jurisdiction enjoining Buyer or Seller from consummating the Closing is entered and such judgment, injunction, order or decree shall have become final and nonappealable;
- (c) by Buyer by notice to Seller, if a breach of any representation or warranty or failure to perform any covenant or agreement on the part of Seller set forth in this Agreement shall have occurred that would cause the condition set forth in <u>Section 6.2(a)</u> not to be satisfied, and such breach is incapable of being cured by the End Date; provided, however, that Buyer shall not have the right to terminate this Agreement pursuant to this <u>Section 7.1(c)</u> if Buyer is then in material breach or violation of its representations, warranties or covenants contained in this Agreement; or
- (d) by Seller by notice to Buyer, if a breach of any representation or warranty or failure to perform any covenant or agreement on the part of Buyer set forth in this Agreement shall have occurred that would cause the condition set forth in Section 6.3(a) not to be satisfied, and such breach is incapable of being cured by the End Date; provided, however, that Seller shall not have the right to terminate this Agreement pursuant to this Section 7.1(d) if Seller is then in material breach or violation of its representations, warranties or covenants contained in this Agreement.
- Section 7.2 <u>Effect of Termination</u>. If this Agreement is terminated pursuant to <u>Section 7.1</u>, this Agreement shall become void and of no effect without liability of any party (or any of its directors, officers, employees, stockholders, Affiliates, agents, successors or assigns) to the other party except as provided in this <u>Section 7.2</u>; *provided* that no such termination (nor any provision of this Agreement) shall relieve any party from liability for any Losses (including claims for Losses based on the consideration that would have otherwise been payable to Seller) for Fraud or for willful breach of any covenant hereunder. The provisions of <u>Section 4.3(d)</u>, this <u>Section 7.2</u>, <u>Section 9.1</u>, <u>Section 9.2</u>, <u>Section 10.4</u> shall survive any termination hereof pursuant to <u>Section 7.1</u>.

## **ARTICLE 8**

## INDEMNIFICATION

Section 8.1 <u>Survival</u>. Except for the representations and warranties contained in <u>Section 2.16</u>, which shall not survive the Closing, all representations and warranties made in this Agreement shall survive the Closing Date for a period of twelve (12) months after the Closing Date; *provided* that the representations and warranties of Seller set forth in <u>Sections 2.1</u> (Organization), <u>2.2(a)</u> (Authority) and <u>2.17</u> (Finders' Fees) (collectively, the "**Fundamental Representations**"), the representations and warranties of Seller set forth in <u>Section 2.18</u> (No Other Representations and Warranties), and the representations and warranties of Buyer set forth in <u>Sections 3.1</u> (Corporate Status), <u>3.2(a)</u> (Authority), <u>3.7</u> (Finders' Fees) and <u>3.9</u> (No Additional Representations; Inspection) shall survive until the expiration of the applicable statute of limitations. The respective covenants and agreements of Seller and Buyer contained in this Agreement required to be performed or complied with prior to the Closing (each, a "**Pre-Closing Covenant**") shall survive the Closing Date for a period of twelve (12) months from the Closing Date, and all other respective covenants and agreements of Seller and Buyer contained in this Agreement (each, a "**Post-Closing Covenant**"), as well as Pre-Closing Covenants to the extent they contemplate performance following the Closing, shall survive the Closing Date hereunder indefinitely or for such lesser period of time as may be specified therein, but not to exceed the applicable statute of limitations in the event of a breach of such covenant or other agreement. Notwithstanding the foregoing, except as set forth in <u>Section 7.2</u>, no representation, warranty, covenant or agreement made in this Agreement shall survive any termination of this Agreement.

Section 8.2 <u>Indemnification by Seller</u>. From and after the Closing, and subject to this <u>Article 8</u>, Seller shall defend, indemnify and hold harmless Buyer, and each of its Subsidiaries and Affiliates, and their respective officers, directors, employees, agents, successors and assigns (collectively, the "**Buyer Indemnitees**") from and against, and pay or reimburse the Buyer Indemnitees for all Losses resulting from (a) any inaccuracy in or breach of any representation or warranty by Seller in or pursuant to this Agreement, (b) any breach or default in performance by Seller of any Pre-Closing Covenant of Seller or (d) any Excluded Liability. Notwithstanding that a claim for Losses may fall into multiple categories of this <u>Section 8.2</u>, Buyer Indemnitees may recover such Losses only one time.

Section 8.3 <u>Indemnification by Buyer</u>. From and after the Closing, and subject to this <u>Article 8</u>, Buyer shall defend, indemnify and hold harmless Seller and each of its Subsidiaries and Affiliates and their respective officers, directors, employees, agents, successors and assigns (collectively, the "Seller Indemnitees") from and against, and pay or reimburse the Seller Indemnitees for, any and all Losses resulting from (a) any inaccuracy in or breach of any representation or warranty by Buyer in or pursuant to this Agreement, (b) any breach or default in performance by Buyer of any Pre-Closing Covenant of Buyer or any Post-Closing Covenant of Buyer, (c) the ownership and operation of the Transferred Assets and the Business after the Closing and (d) the Assumed Liabilities. Notwithstanding that a claim for Losses may fall into multiple categories of this <u>Section 8.3</u>, Seller Indemnitees may recover such Losses only one time.

Section 8.4 Limitations on Indemnity. Buyer and Seller agree, for themselves and on behalf of the Buyer Indemnitees and the Seller Indemnitees that:

- (a) No Buyer Indemnitee shall be entitled to indemnification under Section 8.2(a) in respect of any Loss incurred or suffered by such Buyer Indemnitee until such time as the aggregate of all Losses that Buyer Indemnitees may have under Section 8.2(a) exceeds \$5,500,000 (the "Indemnity Threshold"), and then only for the aggregate amount of all Losses in excess of the Indemnity Threshold. The aggregate liability of Seller in respect of claims for indemnification pursuant to Section 8.2(a) will not exceed \$55,000,000 (the "Cap"). The Cap shall not apply to Losses under Section 8.2(d). Notwithstanding anything to the contrary set forth herein, the provisions of this Section 8.4(a) will not apply in respect of claims for breach of any Fundamental Representation; provided, that in no event shall the aggregate liability of Seller in respect of claims for indemnification pursuant to this Agreement exceed the Purchase Price actually received by Seller.
- (b) No Buyer Indemnitee shall be entitled to indemnification, to sue for damages or to assert any other right or remedy under this Agreement with respect to any Loss, cause of action or other claim to the extent it (i) is a potential Loss that may be asserted rather than an actual Loss that has, in fact, been incurred by such party or (ii) is a Loss with respect to which such party or any of its Affiliates has taken action (or caused action to be taken) to accelerate the time period in which such matter is asserted or payable.
- (c) No Buyer Indemnitee shall be entitled to recover any amount relating to any matter arising under one provision of this Agreement to the extent such Buyer Indemnitee has already recovered such amount with respect to such matter pursuant to such provision or any other provisions of this Agreement.
- (d) With respect to each indemnification obligation in this Agreement: (i) each such obligation shall be calculated on an After-Tax Basis; (ii) all Losses shall be net of any Eligible Insurance Proceeds; (iii) in no event shall an Indemnifying Party have liability to the Indemnified Party for any special, incidental, indirect, speculative, treble or punitive damages, loss of business reputation or opportunity, lost revenue, income or profits, diminution in value or similar items, except if and to the extent any such damages are recovered against an Indemnified Party pursuant to a Third Party Claim; and (iv) the parties shall treat any indemnification payment made under this Agreement as an adjustment to the Purchase Price.
- (e) If any portion of Losses to be reimbursed by the Indemnifying Party may be covered, in whole or in part, by third-party insurance coverage, the Indemnified Party shall promptly give notice thereof to the Indemnifying Party. If the Indemnifying Party so requests, the Indemnified Party shall use its commercially reasonable efforts to collect the maximum amount of insurance proceeds thereunder, in which event all such proceeds actually received, net of costs reasonably incurred by the Indemnified Party in seeking such collection, shall be considered "Eligible Insurance Proceeds". In any case where an Indemnified Party recovers from a third party any Eligible Insurance Proceeds and/or any other amount in respect of any Losses for which an Indemnifying Party has actually reimbursed such Indemnified Party pursuant to this Article 8, such Indemnified Party shall promptly pay over to the Indemnifying Party such Eligible Insurance Proceeds and/or the amount so recovered (after deducting therefrom the amount of expenses incurred by it in procuring such recovery), but not in excess of the sum of (i) any amount previously paid by the Indemnifying Party to or on behalf of the Indemnified Party in respect of such claim and (ii) any amount expended by the Indemnifying Party in pursuing or defending any claim arising out of such matter.
- (f) If the Indemnified Party receives any payment from an Indemnifying Party in respect of any Losses pursuant to Section 8.2 or 8.3 and the Indemnified Party could have recovered all or a part of such Losses from a third party (a "Potential Contributor") based on the underlying claim asserted against the Indemnifying Party, the Indemnified Party shall assign such of its rights to proceed against the Potential Contributor as are necessary to permit the Indemnifying Party to recover from the Potential Contributor the amount of such payment.

- (g) Seller shall have no liability or obligation to indemnify any Buyer Indemnitee with respect to the breach of any representation, warranty, covenant or agreement, based on any facts or circumstances that were reasonably knowable to such Buyer Indemnitee by reason of relevant information having been posted to the Data Room or otherwise provided to such Buyer Indemnitee by or on behalf of Seller.
- (h) Any Indemnified Party shall take all commercially reasonable steps to mitigate any Losses incurred by such party upon and after becoming aware of any event or condition that would reasonably be expected to give rise to any indemnification rights hereunder.

## Section 8.5 Notification of Claims; Third Party Claims.

- (a) A Person that may be entitled to be indemnified under this Agreement (the "Indemnified Party") shall promptly notify the party or parties liable for such indemnification (the "Indemnifying Party") in writing of any claim in respect of which indemnity may be sought under this Article 8, describing in reasonable detail the facts and circumstances with respect to the subject matter of such claim; provided, however, that the failure to provide such notice shall not release the Indemnifying Party from any of its obligations under this Article 8 except to the extent the Indemnifying Party is prejudiced by such failure. The parties agree that (i) in this Article 8 they intend to shorten (in the case of the limited survival periods specified in Section 8.1) the applicable statute of limitations period with respect to certain claims, (ii) notices for claims in respect of a breach of a representation, warranty, covenant or agreement (other than a Post-Closing Covenant) must be delivered prior to the expiration of any applicable survival period specified in Section 8.1 for such representation, warranty, covenant or agreement, (iii) notices for claims in respect of a breach of a Post-Closing Covenant must be delivered prior to the date that is six months after the last day of the effective period of such Post-Closing Covenant and (iv) any claims for indemnification for which notice is not timely delivered in accordance with this Section 8.5 shall be expressly barred and are hereby waived.
- (b) Upon receipt of notice of a claim for indemnity from an Indemnified Party pursuant to this <u>Section 8.5</u> in respect of a pending or threatened claim or demand by a third party that the Indemnified Party has determined has given or could reasonably give rise to a right of indemnification under this Agreement (such claim or demand being a "Third Party Claim" and including a pending or threatened claim or demand asserted by a third party against the Indemnified Party), the Indemnifying Party may, by notice to the Indemnified Party delivered within 20 Business Days of the receipt of notice of such Third Party Claim, assume the defense and control of such Third Party Claim, with its own counsel and at its own expense, but shall allow the Indemnified Party a reasonable opportunity to participate in the defense of such Third Party Claim with its own counsel and at its own expense. The Indemnified Party may take any actions reasonably necessary to defend such Third Party Claim prior to the time that it receives notice from the Indemnifying Party as contemplated by the preceding sentence. The Indemnifying Party shall not, without the prior written consent of the Indemnified Party (which shall not be unreasonably withheld), consent to a settlement, compromise or discharge of, or the entry of any judgment arising from, any Third Party Claim, unless such settlement, compromise or discharge does not involve any finding or admission of any violation of Law or admission of any wrongdoing by the Indemnified Party and the Indemnifying Party shall (i) pay or cause to be paid all amounts arising out of such settlement or judgment concurrently with the effectiveness of such settlement, (ii) not encumber any of the material assets of any Indemnified Party or agree to any restriction or condition that would apply to or materially adversely affect any Indemnified Party and (iii) obtain, as a condition of any settlement or other resolution, a complete and unconditional release of each Indemnified Party from any and all liability

in respect of such Third Party Claim. The Indemnified Party shall not settle, compromise or consent to the entry of any judgment with respect to any claim or demand for which it is seeking indemnification from the Indemnifying Party or admit to any liability with respect to such claim or demand without the prior written consent of the Indemnifying Party, which the Indemnifying Party shall not unreasonably withhold or delay if it has not assumed the defense of such claim or demand.

- (c) Notwithstanding anything to the contrary in this <u>Article 8</u> (including <u>Sections 8.2</u> and <u>8.3</u>), no Indemnifying Party shall have any liability under this <u>Article 8</u> for any Losses arising out of or in connection with any Third Party Claim that is settled or compromised by an Indemnified Party without the prior written consent of such Indemnifying Party.
- (d) In the event any Indemnifying Party receives notice of a claim for indemnity from an Indemnified Party pursuant to this <u>Section 8.5</u> that does not involve a Third Party Claim, the Indemnifying Party shall notify the Indemnified Party within 20 Business Days following its receipt of such notice whether the Indemnifying Party disputes its liability to the Indemnified Party under this <u>Article 8</u>. The Indemnified Party shall reasonably cooperate with and assist the Indemnifying Party in determining the validity of any such claim for indemnity by the Indemnified Party.

Section 8.6 Exclusive Remedy. Anything to the contrary in this Agreement notwithstanding, Seller and Buyer hereby agree that following the Closing, subject to Section 10.9, the sole and exclusive remedy of a party for any breach or inaccuracy of any representation, warranty, covenant or agreement contained in this Agreement shall be the indemnification rights set forth in Article 5 and this Article 8, regardless of the legal theory under which any liability or obligation may be sought to be imposed, whether sounding in contract or tort, or whether at law or in equity, or otherwise. In furtherance of the foregoing and subject to Section 10.9 and the indemnification provisions set forth in Article 5 and this Article 8, (a) Buyer hereby waives, from and after the Closing Date, any and all rights, claims and causes of action any Buyer Indemnitee may have against Seller or any of its Affiliates, or their respective directors, officers, employees, Affiliates, controlling Persons, agents or representatives, successors or assigns and (b) Seller hereby waives, from and after the Closing Date, any and all rights, claims and causes of action any Seller Indemnitee may have against Buyer or any of its Affiliates, or their respective directors, officers, employees, Affiliates, agents or representatives, successors or assigns, in either case to the extent such rights, claims and causes of action relate to this Agreement and arise under or are based upon any Federal, state, provincial, local or foreign statute, law, ordinance, rule or regulation or otherwise. The parties hereto agree that the provisions of this Agreement relating to indemnification, and the limits imposed on Buyer's and the Buyer Indemnitees' rights and remedies with respect to this Agreement and the transactions contemplated hereby (including this Article 8) were specifically bargained for between sophisticated parties and were specifically taken into account in the determination of the amounts to be paid to Seller hereunder. Without limiting the generality o

## ARTICLE 9

### DEFINITIONS

Section 9.1 Certain Terms. The following terms have the respective meanings given to them below:

"Advisors" has the meaning set forth in Section 4.17(c).

- "Affiliate" means, with respect to any Person, any other Person directly or indirectly controlling, controlled by or under common control with such Person.
- "After-Tax Basis" means that, in determining the amount of the payment necessary to indemnify any party against, or reimburse any party for, Losses, the amount of the Losses shall be determined net of any Tax benefit derived (or reasonably expected to be derived) by the Indemnified Party (or any Affiliate thereof) in the fiscal year of, or the fiscal year of, the Indemnified Party during which the Losses were incurred as the result of sustaining or paying such Losses (including as the result of facts or circumstances due to which the Indemnified Party sustained or paid such Losses).
  - "Agreement" has the meaning set forth in the Preamble.
- "Ancillary Agreements" means the (a) Business Transfer Documents, (b) the Bill of Sale, Assignment and Assumption substantially in the form as attached as <a href="Exhibit A">Exhibit A</a>, (c) the Trademark Assignment substantially in the form as attached as <a href="Exhibit B">Exhibit B</a>, (d) the Transition Services Agreement substantially in the form as attached as <a href="Exhibit D">Exhibit D</a>, and (f) all other agreements, documents and instruments executed and delivered in connection with the transactions contemplated by this Agreement.
  - "Assumed Liabilities" means the obligations and liabilities set forth or described on Annex 9.1(c).
  - "Audited Financials" has the meaning set forth in Section 4.18.
  - "Business" has the meaning set forth in the Recitals.
- "Business Day" means any day that is not (a) a Saturday, (b) a Sunday or (c) any other day on which commercial banks are authorized or required by law to be closed in the City of New York.
- "Business Employee Benefit Plan" means each material employee benefit plan (as defined in Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended, whether or not subject thereto), sponsored, contributed to or maintained by Seller or any of its Affiliates in which any Employee participates, excluding any plan, program, agreement or arrangement required by applicable Law or regulation (e.g., government mandated severance plans).
  - "Business Transfer Documents" has the meaning set forth in Section 1.1(e).
  - "Buyer" has the meaning set forth in the Preamble.
  - "Buyer C/S Interests" has the meaning set forth in Section 4.4(b)(vi)(B).
- "Buyer Disclosure Letter" means the letter, dated as of the date hereof, delivered by Buyer to Seller prior to the execution of this Agreement and identified as the Buyer Disclosure Letter.
  - "Buyer Indemnitees" has the meaning set forth in Section 8.2.
  - "Buyer Interests" has the meaning set forth in Section 4.4(b)(vi)(B).

"Buyer Portion of PDUFA Fees" means the dollar amount represented by the following formula: the PDUFA Fees multiplied by a fraction, the numerator of which is the number of days from the Closing Date through September 30, 2018 and the denominator of which is 365.

"Cap" has the meaning set forth in Section 8.4(a).

"cGMP" means then-current standards of good manufacturing practice for the manufacture, processing, packaging, testing or holding of a drug to assure that such drug meets (a) the requirements of applicable Law as to safety, identity and strength, and (b) the quality and purity characteristics that such drug purports or is represented to possess, including as set forth by the FDA in 21 C.F.R. Parts 210 and 211.

"Claim Services" has the meaning set forth in Section 4.17.

"Closing" has the meaning set forth in Section 1.2.

"Closing Date" has the meaning set forth in Section 1.2.

"Closing Inventory" has the meaning set forth in Section 1.3(b)(i).

"Code" means the Internal Revenue Code of 1986, as amended.

"Collective Bargaining Agreement" means any collective bargaining or labor agreement or works council agreement to which Seller or any of its Affiliates is a party with respect to the Business.

"Competition Laws" means all Laws that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or lessening of competition through merger or acquisition or restraint of trade.

"Confidentiality Agreement" has the meaning set forth in Section 4.3(d).

"Contracts" has the meaning set forth in clause (xii) of Annex 9.1(a).

"Copyrights" means those rights existing under the copyright laws for those works subject to the copyright laws and copyright registrations and applications for registration thereof, including all renewals and extensions thereof.

"Data Room" means the electronic data room containing documents and materials relating to the Business as constituted as of the Closing Date.

"Deal Communications" has the meaning set forth in Section 10.10(d).

"Dispute Notice" has the meaning set forth in Section 1.3(c).

"Disputed Item" has the meaning set forth in Section 1.3(c).

"DOJ" has the meaning set forth in Section 4.4(b).

"Eligible Insurance Proceeds" has the meaning set forth in Section 8.4(e).

**"Employee**" means each employee of Seller and its Affiliates who performs services exclusively related to the Business or whose name is set forth on <u>Section 2.13</u> of the Seller Disclosure Letter (subject to <u>Section 4.7(c)</u>) under the heading "Included Employee", including in all cases, each such employee who as of the Closing Date is on leave of absence (including medical leave, military leave, workers compensation leave and short-term or long-term disability) or vacation; *provided* that any individual whose name is set forth on <u>Section 2.13</u> of the Seller Disclosure Letter (subject to <u>Section 4.7(c)</u>) under the heading "Excluded Employee" shall not be an "Employee".

"End Date" has the meaning set forth in Section 7.1(b)(i).

"Environmental Law" means any Law, Permit and contract with a Governmental Authority and regulations thereunder and other requirements having the force of Law and all Orders promulgated under those Laws and regulations and related common law theories regulating or relating to pollution, the protection of natural resources or the environment, public health and safety or employee health and safety.

"ERM" has the meaning set forth in Section 4.19.

"Excluded Assets" means the property of Seller and its Affiliates set forth or described on Annex 9.1(b), which property is not to be transferred to Buyer hereunder.

"Excluded Contracts" has the meaning set forth in clause (vi) of Annex 9.1(b).

"Excluded Liabilities" means the liabilities and obligations of Seller and its Affiliates set forth or described on Annex 9.1(d), which are not to be assumed by Buyer hereunder.

"FDA" means the U.S. Food and Drug Administration.

"FDCA" means the Federal Food, Drug and Cosmetic Act of 1938 of the U.S. Food and Drug Administration, as amended.

"Financial Statements" has the meaning set forth in Section 2.4.

"Fraud" means a claim for common law fraud with a specific intent to deceive based on a representation contained in this Agreement; provided that, at the time such representation was made, (a) such representation was materially inaccurate, (b) the party making such representation had actual knowledge of the material inaccuracy of such representation, (c) the party making such representation had the specific intent to deceive the other party, and (d) the other party acted in reliance on such inaccurate representation and suffered financial injury or Loss as a result of such material inaccuracy.

"FTC" has the meaning set forth in Section 4.4(b).

"Fundamental Representations" has the meaning set forth in Section 8.1.

"GAAP" means generally accepted accounting principles in the United States.

"Governmental Authority" means any nation or government, any state or other political subdivision thereof, any entity, authority or body exercising executive, legislative, judicial, regulatory or administrative functions of or pertaining to government, any court, tribunal or arbitrator and any self-regulatory organization.

"Hazardous Substance" means any material or substance regulated as toxic or hazardous, or as a pollutant or containment, under any applicable Environmental Law.

- "HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder, as the same may be amended from time to time.
  - "Indemnified Party" has the meaning set forth in Section 8.5(a).
  - "Indemnifying Party" has the meaning set forth in Section 8.5(a).
  - "Indemnity Threshold" has the meaning set forth in Section 8.4(a).
- "Independent Accountant" means a partner in the New York office of Grant Thornton or, if no partner at such firm is willing or able to serve in such capacity, a partner in the New York office of another nationally recognized independent registered public accounting firm appointed by mutual agreement of Buyer and Seller.
- "Interest Rate" means an interest rate that is calculated daily based on the current "prime rate" (expressed as a rate per annum) as published in *The Wall Street Journal* for each of the days in the applicable period.
- "**Inventory**" means all inventory owned by any Selling Affiliate and used exclusively in the Business or held for sale exclusively to customers of the Business, including the Products and active pharmaceutical ingredients, spare parts, raw materials, containers, packaging and packaging supplies and work-in-process.
  - "Inventory Statement" has the meaning set forth in Section 1.3(b)(i).
  - "Inventory Target" means an amount equal to \$2,000,000.
  - "IP Rights" means (a) Patents, (b) Trademarks, (c) Copyrights and (d) trade secrets, know-how and data and databases.
  - "IRS" means the Internal Revenue Service.
  - "Knowledge of Seller" means the actual knowledge, as of the date hereof, of the Persons specified in Schedule 9.1(b).
  - "Laws" has the meaning set forth in Section 2.11(a).
  - "Licensed IP Contracts" has the meaning set forth in Section 2.9(e).
- "Lien" means, with respect to any property or asset, any mortgage, lien, pledge, charge, security interest, lease, encumbrance or other adverse claim of any kind in respect of such property or asset.
  - "Listed Plan" has the meaning set forth in Section 2.14(a).
- "Litigation" means any action, cease and desist letter, demand, suit, arbitration proceeding, administrative or regulatory proceeding, citation, summons or subpoena of any nature, civil, criminal, regulatory or otherwise, in law or in equity.
- "Losses" means any and all damages, judgments, awards, liabilities, losses, obligations, claims of any kind or nature, fines and costs and expenses (including reasonable fees and expenses of attorneys, auditors, consultants and other agents).

"Material Adverse Effect" means any material adverse change in, or effect on, the assets, financial condition or results of operations of the Business, taken as a whole; provided that any such change or effect resulting from any of the following, individually or in the aggregate, shall not be considered when determining whether a Material Adverse Effect has occurred: (a) any change in economic conditions generally or capital and financial markets generally, including changes in interest or exchange rates, (b) any change in the industry in which the Business operates or in which Products are used or distributed, including increases in operating costs, (c) any change or proposed change in Laws or GAAP, or the enforcement or interpretation thereof, and any actions taken in order to comply with applicable Law, (d) any failure by the Business to achieve any earnings or other financial projections or forecasts, (e) conditions in jurisdictions in which the Business operates or Products are sold, including hostilities, acts of war, sabotage, terrorism or military actions, or any escalation or worsening of any of the foregoing that do not materially disproportionally affect the Business, (f) any change resulting from the negotiation, execution, announcement or consummation of the transactions contemplated by, or the performance of obligations under, this Agreement or the Ancillary Agreements, including any such change relating to the identity of, or facts and circumstances relating to, Buyer and including any actions by customers, suppliers or personnel, (g) any action taken by Buyer and any of its Affiliates, financing sources, agents or representatives, (h) any hurricane, flood, tornado, earthquake or other natural disaster or any other force majeure event, (i) any actions required to be taken or omitted pursuant to this Agreement or the Ancillary Agreements or taken with Buyer's consent or not taken because Buyer withheld, delayed or conditioned its consent, (j) any matter set forth in the Seller Disclosure Letter, (

"Material Contract" has the meaning set forth in Section 2.6(b).

"NAICS" means the North American Industry Classification System.

"OFAC" has the meaning set forth in Section 2.11(d).

"Order" means any order, writ, judgment, injunction, decree, ruling, assessment, stipulation, determination or award entered by or with any court or other Governmental Authority or arbitrator.

"Organizational Documents" means the articles of incorporation, constitution, certificate of incorporation, charter, bylaws, articles of formation, certificate of formation, regulations, operating agreement, certificate of limited partnership, partnership agreement and all other similar documents, instruments or certificates executed, adopted or filed in connection with the creation, formation or organization of a Person, including any amendments thereto.

"Owned Real Property" means the real property owned by Seller located at 825 Wurlitzer Drive, North Tonawanda, New York 14120.

"Patents" means patents (as well as the relevant complementary protection certificates where applicable) and patent applications (including any divisions, continuations, continuations-in-part, provisional applications, reexamined versions or reissues thereof) whether or not patents are issued on any such applications and whether or not any such applications are modified, withdrawn or resubmitted.

"PDUFA Fees" means all user fees and other prescription drug program fees payable to the FDA for the fiscal year ended September 30, 2018 that relate exclusively to the Products or the facility included in the Transferred Real Property (and not to any of Seller's other products, facilities or operations) under the FDCA, as amended by the Prescription Drug User Fee Amendments of 2017, in each case to the extent transferrable to a purchaser of the Products or that facility.

"Permits" has the meaning set forth in Section 2.11(b).

"Permitted Liens" means (a) real estate and personal property Taxes, assessments, governmental levies, fees or charges or statutory liens for current Taxes or other governmental charges with respect to the Transferred Real Property not yet due and payable or due and payable but not delinquent or the amount or validity of which is being contested in good faith by appropriate proceedings, (b) mechanics', carriers', workers', repairers' and similar statutory liens arising or incurred in the ordinary course of business or in connection with construction contracts for amounts that are not delinquent or are being contested in good faith and that would not individually or in the aggregate be materially adverse to the Business, (c) zoning, entitlement, building codes and other land use regulations, ordinances or legal requirements imposed by any Governmental Authorities having jurisdiction over the Transferred Real Property, (d) all rights relating to the construction and maintenance in connection with any public utility of wires, poles, pipes, conduits and appurtenances thereto, on, under or above the Transferred Real Property which would be shown on a current accurate survey of the Transferred Real Property and which, do not materially impair the use of the Transferred Real Property as currently utilized, (e) all matters disclosed in the Seller Disclosure Letter, (f) any state of facts which an accurate survey or inspection of the Transferred Real Property would disclose and which, individually or in the aggregate, do not materially impair the continued use of the Transferred Real Property for the purposes for which it is used for the Business, (g) title exceptions which (i) are disclosed in the Title Policy or (ii) relate to matters occurring subsequent to the date of the Title Policy and would be disclosed in a current title insurance commitment or title insurance policy for any such Transferred Real Property issued by a title company, (h) statutory Liens in favor of lessors arising in connection with any property leased by any Selling Affiliate, (i) other defects, irregularities or imperfections of title, encroachments, easements, servitudes, permits, rights of way, flowage rights, restrictions, leases, licenses, covenants, sidetrack agreements and oil, gas, mineral and mining reservations, rights, licenses and leases, which, in each case, do not materially impair the continued use of the Transferred Real Property for the purposes for which it is used for the Business, (j) Liens, other than those on the Transferred Real Property, that, individually or in the aggregate, do not, and would not reasonably be expected to, materially detract from the value of any of the property, rights or assets of the Business and (k) with respect to Transferred Real Property Leases, any matters that affect the landlord's fee

"Permitted Removal" has the meaning set forth in Section 10.10(g).

"Person" means an individual, corporation, partnership, limited liability company, association, trust or other entity or organization, including a government or political subdivision or an agency or instrumentality thereof.

"Post-Closing Covenant" has the meaning set forth in Section 8.1.

"Post-Closing Tax Period" means any taxable period beginning after the Closing Date and the portion of a Straddle Period beginning after the Closing Date.

"Post Marketing Safety Studies" means any and all post-authorization safety studies (PASS) or similar post-marketing study or trial, conducted or required to be conducted by any Governmental Authority with respect to any of the Products with the aim of identifying, characterizing or quantifying a safety hazard, confirming the safety profile of such Product, or of measuring the effectiveness of risk management measures related to such Product.

- "Potential Contributor" has the meaning set forth in Section 8.4(f).
- "Pre-Closing Accounts Payable" has the meaning set forth in clause (i) of Annex 9.1(d).
- "Pre-Closing Accounts Receivable" has the meaning set forth in clause (i) of Annex 9.1(b).
- "Pre-Closing Covenant" has the meaning set forth in Section 8.1.
- "Pre-Closing Tax Period" means any Tax period ending on or before the Closing Date and, with respect to a Tax period that begins on or before the Closing Date and ends thereafter, the portion of such Tax period ending at the completion of the Closing Date.
- **"Prepaid Inventory"** means, without duplication with Inventory, inventory that has been paid for by Seller or any Selling Affiliate prior to the Closing Date and was intended to be used by Seller or such Selling Affiliate exclusively in the Business that has not been delivered to Seller or such Selling Affiliate at or prior to the Closing Date. For the avoidance of doubt, Prepaid Inventory shall not be deemed Inventory for purposes of the adjustment pursuant to <u>Section 1.3(g)</u>.
  - "Privileged Deal Communications" has the meaning set forth in Section 10.10(d).
  - "Products" has the meaning set forth in the Recitals.
  - "Purchase Price" means \$1,100,000,000.
  - "Quarterly Financials" has the meaning set forth in Section 4.18.
- "Regulatory Registrations" means the premarket notifications or premarket approvals issued by the FDA, European Union Conformity Marking (CE marks) issued by a European Union Notified Body, and all other technical, medical, scientific, labeling and similar licenses, registrations, authorizations, permits, certifications, franchises, variances, exemptions, orders, approvals, amendments and renewals of the Products (including marketing authorizations and labeling approvals) issued by any Governmental Authority of any country and held or pending (including any applications) as of the Closing Date by Seller or any Selling Affiliate or third-party distributors (under rights of reservation of such seller) that are required for the manufacture, commercialization, labeling, distribution, use, storage, import, export, transport, marketing or sale of the Products within any country, including the approved FDA New Drug Application for the Product.
- "Related Documents" means the Business Transfer Documents, the Transition Services Agreement and each other document to be signed and delivered by the parties under this Agreement.
  - "Residual Communication" has the meaning set forth in Section 10.10(g).
  - "Resolution Period" has the meaning set forth in Section 1.3(d).
  - "SEC" means the U.S. Securities and Exchange Commission.
  - "Seller" has the meaning set forth in the Preamble.
- "Seller Disclosure Letter" means the letter, dated as of the date hereof, delivered by Seller to Buyer prior to the execution of this Agreement and identified as the Seller Disclosure Letter.
  - "Seller Indemnitees" has the meaning set forth in Section 8.3.

- "Seller Parties" has the meaning set forth in Section 10.10(a).
- "Seller Product Claims" has the meaning set forth in Section 4.17.
- "Selling Affiliates" means all of the Affiliates of Seller that own any Transferred Assets or have obligations or liabilities in respect of any Assumed Liabilities.
- "Separated Contract" has the meaning set forth in Section 4.5.
- "Service Request" has the meaning set forth in Section 4.17(a).
- "Shared Contract" means each Contract that relates to both (a) the Business or any Transferred Assets and (b) one or more other businesses or products of Seller or any Affiliate of Seller that is identified as such on Schedule 9.1(a)(xii).
  - "Solvent" has the meaning set forth in Section 3.5.
  - "Straddle Period" means any Tax period that includes, but does not end on, the Closing Date.
  - "Stub Period Financials" has the meaning set forth in Section 4.18.
- "Subsidiary" means, with respect to any Person, any entity of which securities or other ownership interests (a) having voting power to elect a majority of the board of directors or other persons performing similar functions or (b) representing more than fifty percent of such securities or ownership interests are at the time directly or indirectly owned by such Person.
- "Tax" means any federal, state, local or foreign income, alternative, minimum, accumulated earnings, personal holding company, franchise, capital stock, profits, windfall profits, gross receipts, sales, use, goods and services, value added, transfer, registration, stamp, premium, excise, severance, real property, personal property, ad valorem, occupancy, license, occupation, employment, payroll, social security, disability, unemployment, workers' compensation, withholding, estimated or other similar tax (including all interest and penalties thereon and additions thereto).
  - "Tax Proceeding" means any audit, request for information, investigation, hearing, litigation, legal action, or judicial contest relating to Taxes.
- "Tax Return" means any federal, state, local or foreign tax return, declaration, statement, report, schedule, form or information return or any amendment to any of the foregoing relating to Taxes.
  - "Taxing Authority" means any Governmental Authority exercising any authority to impose, regulate or administer the imposition of Taxes.
  - "Teva USA" has the meaning set forth in Section 1.1(f).
  - "Third Party Claim" has the meaning set forth in Section 8.5(b).
  - "Title Policy" means Ticor Title Insurance Policy No. 7430632-78651461, dated August 4, 2009, issued to Teva USA.
  - "Trademark License Agreement" has the meaning set forth in Section 4.14.

"Trademarks" means trademarks, service marks, trade names, trade dress, logos, slogans, and other similar designations of source or origin, together with the goodwill associated with any of the foregoing, and all applications, registrations and renewals therefor.

"Transfer Letters" means letters to the applicable Governmental Authority and, if necessary for filing with the FDA, an application form in substantially the same form as described in 21 C.F.R. § 314.72, which are submitted by Buyer and Seller to transfer all rights to a Regulatory Registration for a transferred Product from Seller or its Affiliates to Buyer.

"Transfer Taxes" mean any federal, state, county, local, foreign and other sales, use, transfer, goods and services, value added, conveyance, documentary transfer, stamp duty, recording or other similar Tax, fee or charge imposed on or in connection with the transactions contemplated by or the instruments executed under or in connection with this Agreement or the recording of any sale, transfer, or assignment of property (or any interest therein) effected pursuant to this Agreement.

"Transferred Assets" means the Owned Real Property, assets and other personal property of Seller and its Affiliates exclusively pertaining to the Business set forth or described on Annex 9.1(a), which expressly exclude the Excluded Assets, with such changes between the date of this Agreement and the Closing Date as shall have occurred in transactions not inconsistent with any of Seller's representations, warranties, covenants and agreements set forth herein.

"Transferred Contracts" has the meaning set forth in clause (xii) of Annex 9.1(a).

"Transferred Employee Liabilities" has the meaning set forth in clause (viii) of Annex 9.1(c).

"Transferred IP" has the meaning set forth in clause (vii) of Annex 9.1(a).

"Transferred IT" has the meaning set forth in clause (ix) of Annex 9.1(a).

"Transferred Real Property" has the meaning set forth in clause (i) of Annex 9.1(a).

"Transferred Real Property Leases" has the meaning set forth in clause (i) of Annex 9.1(a).

"Transferred Records" has the meaning set forth in clause (vi) of Annex 9.1(a).

"Transition Services Agreement" has the meaning set forth in Section 4.16.

"Unresolved Items" has the meaning set forth in Section 1.3(e).

"Waiver Request" has the meaning set forth in Section 4.18.

Section 9.2 <u>Construction</u>. The words "hereof", "herein" and "hereunder" and words of like import used in this Agreement shall refer to this Agreement as a whole and not to any particular provision of this Agreement. The words "party" or "parties" shall refer to parties to this Agreement. The captions herein are included for convenience of reference only and shall be ignored in the construction or interpretation hereof. References to Articles, Sections and Exhibits are to Articles, Section and Exhibits of this Agreement unless otherwise specified. All Exhibits and Disclosure Letters annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any capitalized term used in any Exhibit or Disclosure Letter but not otherwise defined therein shall have the meaning given to such term in this Agreement. Any singular term in this Agreement shall be deemed to include the plural, and any plural term the singular. Whenever the words "include",

"includes" or "including" are used in this Agreement, they shall be deemed to be followed by the words "without limitation", whether or not they are in fact followed by those words or words of like import. "Writing", "written" and comparable terms refer to printing, typing and other means of reproducing words (including electronic media) in a visible form. References to any agreement or contract are to that agreement or contract as amended, modified or supplemented from time to time in accordance with the terms hereof and thereof. References to any Person include the successors and permitted assigns of that Person. References from or through any date mean, unless otherwise specified, from and including or through and including, respectively. Any reference to "days" means calendar days unless Business Days are expressly specified. If any action under this Agreement is required to be done or taken on a day that is not a Business Day, then such action shall be required to be done or taken not on such day but on the first succeeding Business Day thereafter.

## **ARTICLE 10**

### MISCELLANEOUS

Section 10.1 Notices. All notices, requests and other communications to any party hereunder shall be in writing (including electronic transmission) and shall be given:

if to Buyer,

CooperSurgical, Inc. 75 Corporate Drive Trumbull, CT 06611

Attention: Robert D. Auerbach, M.D. FACOG, Executive Vice President and Chief Strategy and Medical Officer and

Graceann Pisano, Corporate Counsel

Telephone: 203-601-5200

 $E-mail: robert. auerbach @coopersurgical.com \ and \ rachel. augustine @coopersurgical.com \ and \ graceann. pis an o@coopersurgical.com \ an own \ graceann. pis an o@coopersurgical.com \ an own \ graceann. pis an o@coopersurgical.com \ an own \ graceann. pis an own \ grac$ 

with copies (which shall not constitute notice) to:

The Cooper Companies, Inc. 6140 Stoneridge Mall Road, Suite 590 Pleasanton, California 94588 Attention: Randal Golden, Vice President and General Counsel Telephone: 925-460-3660

Email: rgolden@cooperco.com

Carter Ledyard & Milburn LLP 2 Wall Street New York, NY 10005 Attention: David I. Karabell and Bryan Hall

Telephone: 212-732-3200

E-mail:karabell@clm.com and hall@clm.com

if to Seller,

Teva Pharmaceutical Industries Ltd. 425 Privet Road Horsham, PA 19044

Attention: Ira Schreger, Sr. Vice President and General Counsel, Transactions

Telephone: 215-591-3000

E-mail: ira.schreger@tevapharm.com

with a copy (which shall not constitute notice) to:

Goodwin Procter LLP 100 Northern Avenue Boston, MA 02210

Attention: Kingsley Taft, H. David Henken and Gregg Katz Telephone: (617) 570-1222, (617) 570-1672 and (617) 570-1406

E-mail: ktaft@goodwinlaw.com, dhenken@goodwinlaw.com and gkatz@goodwinlaw.com

or such other address or e-mail address as such party may hereafter specify for the purpose by notice to the other parties hereto. All such notices, requests and other communications shall be deemed received on the date of receipt by the recipient thereof if received prior to 5:00 p.m. on a Business Day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed to have been received on the next succeeding Business Day in the place of receipt.

Section 10.2 <u>Amendment; Waivers, etc.</u>. No amendment, modification or discharge of this Agreement, and no waiver hereunder, shall be valid or binding unless set forth in writing and duly executed by the party against whom enforcement of the amendment, modification, discharge or waiver is sought. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the party granting such waiver in any other respect or at any other time. Neither the waiver by any of the parties hereto of a breach of or a default under any of the provisions of this Agreement, nor the failure by any of the parties, on one or more occasions, to enforce any of the provisions of this Agreement or to exercise any right or privilege hereunder, shall be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. The rights and remedies herein provided are cumulative and none is exclusive of any other, or of any rights or remedies that any party may otherwise have at law or in equity.

Section 10.3 Expenses. Except as otherwise provided herein, all costs, fees and expenses incurred in connection with this Agreement, the Ancillary Agreements and the transactions contemplated hereby and thereby, whether or not consummated, shall be paid by the party incurring such cost or expense.

Section 10.4 Governing Law, etc.

(a) THIS AGREEMENT SHALL BE GOVERNED IN ALL RESPECTS, INCLUDING AS TO VALIDITY, INTERPRETATION AND EFFECT, BY THE LAWS OF THE STATE OF NEW YORK, WITHOUT GIVING EFFECT TO ITS PRINCIPLES OR RULES OF CONFLICT OF LAWS, TO THE EXTENT SUCH PRINCIPLES OR RULES ARE NOT MANDATORILY APPLICABLE BY STATUTE AND WOULD PERMIT OR REQUIRE THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION. Buyer and Seller hereby irrevocably

submit to the jurisdiction of the courts of the State of New York located in Manhattan, New York City, New York and the federal courts of the United States of America also located in Manhattan solely in respect of the interpretation and enforcement of the provisions of this Agreement and in respect of the transactions contemplated hereby. Each of Buyer and Seller irrevocably agrees that all claims in respect of the interpretation and enforcement of the provisions of this Agreement and in respect of the transactions contemplated hereby, or with respect to any such action or proceeding, shall be heard and determined in such a New York State or federal court, and that such jurisdiction of such courts with respect thereto shall be exclusive, except solely to the extent that all such courts shall lawfully decline to exercise such jurisdiction. Each of Buyer and Seller hereby waives, and agrees not to assert, as a defense in any action, suit or proceeding for the interpretation or enforcement hereof or in respect of any such transaction, that it is not subject to such jurisdiction. Each of Buyer and Seller hereby waives, and agrees not to assert, to the maximum extent permitted by law, as a defense in any action, suit or proceeding for the interpretation or enforcement hereof or in respect of any such transaction, that such action, suit or proceeding may not be brought or is not maintainable in such courts or that the venue thereof may not be appropriate or that this Agreement may not be enforced in or by such courts. Buyer and Seller hereby consent to and grant any such court jurisdiction over the person of such parties and over the subject matter of any such dispute and agree that mailing of process or other papers in connection with any such action or proceeding in the manner provided in Section 10.1 or in such other manner as may be permitted by law, shall be valid and sufficient service thereof. Notwithstanding anything in this Section 10.4 to the contrary, each party to this Agreement may bring an action to see

- (b) EACH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY RIGHT SUCH PARTY MAY HAVE TO A TRIAL BY JURY IN RESPECT OF ANY LITIGATION DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT OR THE TRANSACTIONS CONTEMPLATED HEREBY.
- (c) IN CONNECTION WITH ANY DISPUTE HEREUNDER, EACH PARTY HERETO WAIVES ANY CLAIM OF SPECIAL, INCIDENTAL, INDIRECT, SPECULATIVE, TREBLE OR PUNITIVE DAMAGES, LOSS OF BUSINESS REPUTATION OR OPPORTUNITY, LOST REVENUE, INCOME OR PROFITS, DIMINUTION IN VALUE OR SIMILAR ITEMS FROM THE OTHER PARTY HERETO (OR ANY AFFILIATE OF SUCH OTHER PARTY HERETO), EXCEPT THAT THE COURT SHALL HAVE THE POWER TO AWARD ANY RELIEF PROVIDED BY GOVERNING STATUTE (IT BEING UNDERSTOOD THAT THIS WAIVER DOES NOT COVER ANY RIGHT TO INDEMNITY FOR CONSEQUENTIAL, SPECIAL, INCIDENTAL, INDIRECT OR PUNITIVE DAMAGES, LOST PROFITS, DIMINUTION IN VALUE OR SIMILAR ITEMS PAYABLE TO THIRD PARTIES THAT MAY BE IMPOSED OR OTHERWISE INCURRED).
- (d) Neither this Agreement nor any right or obligation of any of the parties under this Agreement shall be governed by the U.N. Convention on Contracts for the International Sale of Goods, and the parties to this Agreement expressly waive or disclaim, as the case may be, any right or obligation they may have under this Agreement pursuant to the U.N. Convention on Contracts for the International Sale of Goods.

Section 10.5 <u>Successors and Assigns</u>. This Agreement shall be binding upon and inure to the benefit of the parties and their respective heirs, successors and permitted assigns; *provided* that neither party to this Agreement may assign any of its rights or obligations under this Agreement, including by sale of stock, operation of Law in connection with a merger or sale of substantially all the assets, without the prior written consent of the other party, except that (i) Seller may, without such consent, assign its rights to, or have its obligations discharged by, an Affiliate and (ii) Buyer may, without such consent,

assign its rights to acquire the Transferred Assets and assume the Assumed Liabilities hereunder, in whole or in part, to one or more of its Affiliates; *provided*, *however*, that no such assignment by Buyer or Seller shall relieve Buyer or Seller of any of its obligations hereunder. An assignment to an Affiliate of Buyer permitted under Section 10.5(ii) shall be deemed null and void as of the time of the assignment if, following such assignment such Affiliate ceases to be an Affiliate of Buyer. Any attempted assignment or transfer in violation of this Section 10.5 shall be null and void.

Section 10.6 Entire Agreement. This Agreement, the Ancillary Agreements (when executed and delivered) and the Confidentiality Agreement constitute the entire agreement and supersede all prior agreements, understandings and representations, both written and oral, between the parties with respect to the subject matter hereof.

Section 10.7 <u>Severability</u>. If any provision, including any phrase, sentence, clause, section or subsection, of this Agreement is determined by a court of competent jurisdiction to be invalid, inoperative or unenforceable for any reason, such circumstances shall not have the effect of rendering such provision in question invalid, inoperative or unenforceable in any other case or circumstance, or of rendering any other provision herein contained invalid, inoperative or unenforceable to any extent whatsoever. Upon any such determination, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.

Section 10.8 <u>Counterparts</u>; <u>Effectiveness</u>; <u>Third Party Beneficiaries</u>. This Agreement may be executed in several counterparts, each of which shall be deemed an original and all of which shall together constitute one and the same instrument. This Agreement shall become effective when each party shall have received a counterpart hereof signed by all of the other parties. Until and unless each party has received a counterpart hereof signed by the other party, this Agreement shall have no effect and no party shall have any right or obligation hereunder (whether by virtue of any other oral or written agreement or other communication). Except as provided under <u>Article 8</u>, no provision of this Agreement is intended to confer any rights, benefits, remedies, obligations or liabilities hereunder upon any Person other than the parties and their respective successors and assigns.

Section 10.9 <u>Specific Performance</u>. The parties agree that irreparable damage would occur if any provision of this Agreement were not performed in accordance with the terms hereof and that the parties shall be entitled to an injunction or injunctions to prevent breaches of this Agreement or to enforce specifically the performance of the terms and provisions hereof in any court specified in <u>Section 10.4</u>, in addition to any other remedy to which they are entitled at law or in equity. The parties hereby waive, in any action for specific performance, the defense of adequacy of a remedy at law and the posting of any bond or other security in connection therewith.

## Section 10.10 Waiver of Conflicts; Deal Communications.

- (a) Each of the parties hereto acknowledges and agrees that Goodwin Procter LLP ("Goodwin") has acted as counsel to Seller (together with the Affiliates of Seller, collectively, the "Seller Parties") in connection with the negotiation of this Agreement and consummation of the transactions contemplated hereby.
- (b) Buyer hereby consents and agrees to Goodwin representing any of the Seller Parties after the Closing, including with respect to disputes in which the interests of the Seller Parties may be directly adverse to Buyer and its Affiliates, and even though Goodwin may have represented a Seller Party in a matter substantially related to any such dispute, or may be handling ongoing matters for Seller. Buyer further consents and agrees to the communication by Goodwin to the Seller Parties in connection with any such representation of any fact known to Goodwin arising by reason of Goodwin's prior representation of Seller or any of its Affiliates.

- (c) In connection with the foregoing, Buyer hereby irrevocably waives and agrees not to assert any conflict of interest arising from or in connection with: (i) Goodwin's prior representation of Seller or any of its Affiliates and (ii) Goodwin's representation of the Seller Parties prior to and after the Closing.
- (d) Buyer further agrees, on behalf of itself and its Affiliates, that all communications in any form or format whatsoever between or among any of Goodwin and/or any Seller Party, or any of their respective managers, directors, officers employees or other representatives that relate in any way to the negotiation, documentation and consummation of the transactions contemplated by this Agreement or any dispute arising under this Agreement (collectively, the "Deal Communications") shall be deemed to be retained and owned collectively by the Seller Parties, shall be controlled by Seller on behalf of the Seller Parties and shall not pass to or be claimed by Buyer or any of its Affiliates. All Deal Communications that are attorney-client privileged (the "Privileged Deal Communications") shall remain privileged after the Closing and the privilege and the expectation of client confidence relating thereto shall belong solely to the Seller Parties, shall be controlled by Seller on behalf of the Seller Parties and shall not pass to or be claimed by Buyer or any of its Affiliates.
- (e) Notwithstanding the foregoing, in the event that a dispute arises between Buyer, on the one hand, and a third party other than a Seller Party, on the other hand, Buyer may assert the attorney-client privilege to prevent the disclosure of the Privileged Deal Communications to such third party; provided, that Buyer may not waive such privilege without the prior written consent of Seller, which Seller shall not unreasonably withhold or delay. In the event that Buyer is legally required by governmental order or otherwise to access or obtain a copy of all or a portion of the Deal Communications, Buyer shall immediately (and, in any event, within two (2) Business Days) notify Seller in writing (including by making specific reference to this Section) so that Seller can seek a protective order and Buyer agrees at the request and expense of Seller to use its reasonable commercial efforts to assist therewith.
- (f) To the extent that files or other materials maintained by Goodwin constitute property of its clients, only the Seller Parties shall hold such property rights and Goodwin shall have no duty to reveal or disclose any such files or other materials or any Deal Communications by reason of any attorney-client relationship between Goodwin and any Seller Party.
- (g) Buyer agrees that it will not (i) access or use the Deal Communications, including by way of review of any electronic data, communications or other information, or by seeking to have any Seller Party waive the attorney-client or other privilege, or by otherwise asserting that Buyer has the right to waive the attorney-client or other privilege or (ii) seek to obtain the Deal Communications from Goodwin. In furtherance of the foregoing, it shall not be a breach of any provision of this Agreement if prior to the Closing any Seller Party, or any of their respective directors, officers employees or other representatives takes any action to protect from access or remove from any Transferred Asset or the premises of the Business (or any offsite back-up or other facilities) any Deal Communications, including without limitation by segregating, encrypting, copying, deleting, erasing, exporting or otherwise taking possession of any Deal Communications (any such action, a "Permitted Removal"). In the event that, notwithstanding any good faith attempts by any Seller Party, or any of their respective directors, officers, employees or other representatives to achieve a Permitted Removal of any Deal Communication, any copy, backup, image, or other form or version or electronic vestige of any portion of such Deal Communication remains accessible to or discoverable or retrievable by Buyer (each, a "Residual").

**Communication**"), Buyer agrees that it will not, and that it will cause its Affiliates and their respective directors, officers, employees or other representatives not to intentionally use or intentionally attempt to use any means for the purpose of accessing, retrieving, restoring, recreating, unarchiving or otherwise gaining access to or viewing any Residual Communication.

**IN WITNESS WHEREOF,** the parties have duly executed this Agreement as of the date first above written.

## TEVA PHARMACEUTICAL INDUSTRIES LTD.

By: /s/ Yitzhak Peterburg

Name: Yitzhak Peterburg Title: Chief Executive Officer

By: /s/ Dror Bashan

Name: Dror Bashan

Title: SVP M&A and Global Alliances

## COOPERSURGICAL, INC.

By: /s/ Robert D. Auerbach

Name: Robert D. Auerbach, M.D. FACOG Title: EVP and Chief Strategy & Medical Officer

## Annex 9.1(a) Transferred Assets

The Transferred Assets consist only of Seller's and its relevant Affiliates' right, title and interest in, to and under the following assets as they exist at the time of the Closing, but in each case specifically excluding any assets constituting Excluded Assets:

- (i) Real Property and Real Property Leases. (A) The real property and any buildings, improvements and fixtures thereon, as set forth on Schedule 9.1(a)(i)(A) (collectively, the "Transferred Real Property") and (B) the real property leases set forth on Schedule 9.1(a)(i)(B) (the "Transferred Real Property Leases");
  - (ii) Inventory; Prepaid Inventory. All Inventory and Prepaid Inventory, in each case at the time of the Closing;
- (iii) Equipment. To the extent transferable, the machinery, equipment, tools and other tangible personal property including furniture, trade fixtures, stores and supplies (but excluding the Inventory and Transferred IT which are identified separately on this Annex 9.1(a)) that is (A) owned by any Selling Affiliate and (B) exclusively used in the Business, whether located at the sites of such Selling Affiliate, at a customer facility or the property of any vendor performing manufacturing, warehousing or other services for the Business, together with the interests of Seller or any Selling Affiliate in respect of any warranties relating thereto;
  - (iv) Goodwill. The goodwill generated by or associated with the Business;
- (v) <u>Permits</u>. To the extent transferable, all permits, licenses and authorizations granted to any Selling Affiliate by a Governmental Authority exclusively in connection with the operation of the Business prior to the Closing Date;
- (vi) <u>Business Records</u>. All books, records, files, data and information in all media exclusively relating to the Business and in the possession of Seller or any Selling Affiliate (but excluding records or files not reasonably separable from documents or databases that do not relate exclusively to the Business), including: (1) vendor lists, (2) customer lists, (3) a list of the distributors for the Products, (4) pricing lists for the Products, (5) testing and clinical data, market research reports, marketing plans and other marketing-related information and materials, (6) quality control, vigilance and regulatory records and (7) other business records (collectively, the "**Transferred Records**");
- (vii) Intellectual Property. (A) the Trademarks exclusively related to the Business, including the Trademarks set forth on  $\underline{Schedule~9.1(a)(vii)(A)}$  and (B) the Patents exclusively related to the Business, including the Patents set forth on  $\underline{Schedule~9.1(a)(vii)(B)}$  (collectively, the "Transferred IP");
  - (viii) Domain Names. The Internet Domain Names exclusively related to the Business, including the Internet domain names set forth on Schedule 9.1(a)(viii);
- (ix) <u>IT Systems</u>. To the extent transferable, (A) information management systems, databases, software programs and source code and (B) computer hardware, in each case, only to the extent set forth on <u>Schedule 9.1(a)(ix)</u> (collectively, the "**Transferred IT**");
  - (x) Listings, etc. Telephone and fax numbers and listings in directories in all media exclusively relating to the Business;

Annex 9.1(a) - 1

(xi) Prepaid Charges. All prepaid charges exclusively related to the Business, including the Buyer Portion of PDUFA Fees;

(xii) <u>Transferred Contracts</u>. All leases, licenses (other than Transferred Real Property Leases, which are identified separately on this <u>Annex 9.1(a)</u>), bids, tenders, purchase orders, consulting agreements, supply agreements, distribution contracts, manufacturing contracts, maintenance contracts, agreements, commitments and other contracts (collectively, "Contracts") relating to the operation and conduct of the Business or any of the Transferred Assets as set forth on <u>Schedule 9.1(a)(xii)</u> (collectively, the "Transferred Contracts"); provided, however, that the Shared Contracts shall be subject to the provisions of <u>Section 4.5</u>; and

(xiii) Regulatory Registrations. All Regulatory Registrations as set forth on Schedule 9.1(a)(xiii).

Annex 9.1(a) - 2

## Annex 9.1(b) Excluded Assets

The Excluded Assets consist of any assets of Seller or any of its Affiliates that do not constitute Transferred Assets as described on Annex 9.1 (a), including without limitation, the following:

- (i) <u>Accounts Receivable</u>. All accounts receivable, notes receivable and similar rights to receive payments of Seller or any of its Affiliates existing on the Closing Date and arising out of the operation or conduct of the Business prior to the Closing Date (the "**Pre-Closing Accounts Receivable**");
- (ii) <u>Cash and Cash Equivalents</u>. All cash, cash equivalents and deposits (including marketable securities and other investment assets and all monies received in respect of the sale of warranty programs) held by Seller or any of its Affiliates on the Closing Date;
- (iii) <u>Hedging or Other Currency Exchange Agreements</u>. All rights to receive payments of Seller or any of its Affiliates pursuant to a hedging or other currency exchange agreement existing on the Closing Date;
- (iv) <u>Benefit Plans</u>. All the assets of and all the assets relating to and all rights under any employee compensation, benefit or welfare plan or any related contract between any Person and Seller or any of its Affiliates (including Business Employee Benefit Plans);
- (v) <u>Certain Records</u>. Any records and files not identified as Transferred Records, including (A) the personnel records maintained by Seller or any of its Affiliates, (B) Tax Returns, (C) records (including accounting records) relating to Taxes paid or payable by Seller or any of its Affiliates and all financial and Tax records relating to the Business that form part of Seller's or any of its Affiliates' general ledger or otherwise constitute accounting records, (D) records prepared in connection with the transactions contemplated by this Agreement or the Ancillary Agreements, including bids received from other Persons and analyses relating to the Business and (E) file copies of the Transferred Records retained by Seller or any of its Affiliates, in each case whether generated before or after the Closing Date;
- (vi) <u>Certain Contracts and Contract Rights</u>. All rights of Seller and its Affiliates under (A) this Agreement and the Ancillary Agreements, (B) any Shared Contracts, (C) any Contracts related to shared services and systems provided by Seller or its Affiliates to the Business, other than Buyer's rights under the Ancillary Agreements and (D) any contracts between or among Seller and any of its Affiliates or between or among Affiliates of Seller, whether arising before or after the Closing Date (collectively, the "**Excluded Contracts**");
- (vii) <u>Insurance</u>. All current and prior insurance policies arranged or maintained by Seller or any of its Affiliates and all rights of any nature with respect thereto, including all rights to insurance recoveries thereunder and to assert claims with respect to any such insurance recoveries, whether arising before or after the Closing Date;
- (viii) <u>Corporate Organizational Records</u>. The organizational documents, qualifications to do business as a foreign corporation, arrangements with registered agents relating to foreign qualifications, taxpayer and other identification numbers, seals, minute books, stock transfer books, blank stock certificates and other documents relating to the organization, maintenance and existence of Seller and each of its Affiliates as a corporation or other entity;
  - (ix) Capital Stock. All shares of capital stock of Seller's Affiliates;

Annex 9.1(b) - 1

- (x) <u>Tax Claims</u>. (A) Refunds and credits, claims for refunds or credits and rights to receive refunds or credits from any Taxing Authority with respect to Taxes arising out of, relating to or in respect of the Business or the Transferred Assets for any Pre-Closing Tax Period and (B) all VAT credits, refunds, or reclaim rights held by Seller or any of its Affiliates on the Closing Date;
- (xi) <u>Real Property</u>. Each of the following: (A) any real property and any buildings, improvements and fixtures thereon (other than the Transferred Real Property); and (B) any leasehold interests, including any prepaid rent, security deposits and options to renew or purchase in connection therewith, of Seller or any of its Affiliates (other than the Transferred Real Property Leases);
  - (xii) Intellectual Property. Except for Transferred IP, all other IP Rights;
  - (xiii) Domain Names. All Internet domain names other than those set forth on Schedule 9.1(a)(viii); and
- (xiv) Excluded IT Systems. All property in the nature of databases, software programs, computer hardware, source code and object code owned or licensed by Seller or any of its Affiliates, in each case that is not otherwise specifically set forth on Schedule 9.1(a)(ix).

Annex 9.1(b) - 2

## Annex 9.1(c) Assumed Liabilities

The Assumed Liabilities consist of any and all liabilities and obligations of Seller or any of its Affiliates arising out of, relating to or otherwise in any way in respect of the Business or any Transferred Asset, whether known or unknown or accrued or contingent at the Closing, whether presently existing or arising after the Closing, other than and subject to the Excluded Liabilities, including the following liabilities and obligations:

- (i) Accounts Payable. All accrued receipts and accounts payable arising out of or relating to the operation or conduct of the Business after the Closing Date;
- (ii) <u>Transferred Contract Liabilities</u>. All liabilities and obligations that accrue after the Closing Date arising out of or relating to the Transferred Contracts, but excluding those in respect of the Pre-Closing Accounts Payable; *provided*, *however*, that the Shared Contracts shall be subject to the provisions of Section 4.5;
  - (iii) Taxes and Transfer Taxes. (A) All Taxes arising out of or relating to the Business or the Transferred Assets for all Post-Closing Tax Periods and (B) all Transfer Taxes;
- (iv) <u>Asset Ownership</u>. All liabilities and obligations that accrue after the Closing Date arising out of or relating to any Transferred Asset or the ownership by Buyer and its Affiliates of any Transferred Asset;
- (v) Product Claims. All liabilities and obligations arising from or relating to (A) lawsuits or other claims, regardless of when commenced or made and irrespective of the legal theory asserted, arising from or relating to the design, manufacture, testing, advertising, marketing, distribution or sale of any unit of Product inserted in a patient after the Closing Date, including all liabilities and obligations arising from or relating to (I) warranty obligations, (II) infringement, dilution, misappropriation or other violation of IP Rights, or (III) alleged or actual hazard or defect in design, manufacture, materials or workmanship, including any failure to warn or alleged or actual breach of express or implied warranty or representation, in each case with respect to a unit of Product inserted in a patient after the Closing Date or (B) the return after the Closing of any Product sold prior to or after the Closing;
- (vi) Environmental Liabilities. All liabilities and obligations that accrue after the Closing Date in respect of the Transferred Real Property, the Business or any Transferred Asset arising under or relating to any Environmental Law or Hazardous Substances;
- (vii) <u>Business Claims</u>. Except as otherwise set forth in this Agreement and except for the matters specifically identified as Excluded Liabilities, all obligations and liabilities in respect of any criminal, civil or administrative suit, action or proceeding, pending or threatened, and claims, whether or not presently asserted, arising out of or relating to the Business or the operation or conduct of the Business after the Closing Date;
- (viii) Employment Matters. Excluding in all cases any equity awards or equity-related compensation, all employment, labor, compensation, pension, employee welfare and employee benefits related liabilities, obligations, commitments, claims, losses and other liabilities contained in Contracts with Employees relating to each Employee (or any dependent or beneficiary of any Employee) that (1) arise as a result of an event or events that occur on or after the Closing Date or (2) Buyer or its Affiliates have specifically agreed to assume pursuant to this Agreement (such liabilities, obligations, commitments, claims and losses, the "Transferred Employee Liabilities");

Annex 9.1(c) - 1

- (ix) <u>Liabilities to Suppliers</u>. Liabilities and obligations to suppliers or other third parties, such as licensors, for materials and services, to the extent relating to the Business and ordered in the ordinary course of business on or prior to the Closing Date, but scheduled to be delivered or provided after the Closing Date;
  - (x) Liabilities to Customers. Liabilities and obligations to customers under purchase orders for Products that have not yet been shipped at the Closing Date; and
- (xi) Post Marketing Safety Studies. All Liabilities and obligations arising from or relating to the performance of all Post Marketing Safety Studies that have been initiated, or committed to, by Seller or any Selling Affiliate prior to the Closing Date.

Annex 9.1(c) - 2

## Annex 9.1(d) Excluded Liabilities

The Excluded Liabilities consist of the following liabilities and obligations of Seller or any of its Affiliates other than to the extent they consist of Assumed Liabilities:

- (i) <u>Accounts Payable.</u> All accrued receipts and accounts payable arising out of the operation or conduct of the Business before the Closing Date (the "**Pre-Closing Accounts Payable**");
- (ii) <u>Taxes</u>. All Taxes (other than Transfer Taxes) of the Selling Affiliates arising out of, relating to or in respect of the Business or the Transferred Assets for all Pre-Closing Tax Periods;
- (iii) Employment Matters. All employment, labor, compensation, pension, employee welfare and employee benefits related liabilities, obligations, commitments, claims and losses relating to each (A) employee of Seller and its Affiliates, including all former employees of the Business (or any dependent or beneficiary of any such employee), other than the Employees and their dependents and beneficiaries, in each case, that arise out of an event or events that occur at any time and (B) Employee (or any dependent or beneficiary of any Employee) that arise as a result of an event or events that occur prior to the Closing Date, except for any such liabilities, obligations, commitments, claims and losses described in clause (2) of the definition of Transferred Employee Liabilities;
- (iv) <u>Agreement Related Liabilities</u>. All liabilities and obligations of Seller and Seller Affiliates under and related to this Agreement and Related Documents including for expenses and fees incidental to the negotiation, preparation, approval or authorization of this Agreement and the Related Documents or consummation of the transaction contemplated by this Agreement and Related Documents;
- (v) <u>Liability for Breach</u>. Any obligation or liability of Seller or any Seller Affiliate under any Transferred Contract arising out of a breach or alleged breach thereof that occurred as of or prior to the Closing and any obligation or liability of Seller or any Seller Affiliate under any contract which is not a Transferred Contract, whether or not arising out of a breach or alleged breach thereof;
- (vi) <u>Liability for Violation of Laws, etc.</u> Any obligation or liability of Seller or any Seller Affiliate (A) arising by reason of any violation or alleged violation of any Laws or any requirement of any Governmental Authority, (B) arising under any Environmental Laws, including those with respect to the operation of any business or the Transferred Assets (including any properties now or previously owned, leased or occupied by Seller or any Seller Affiliate) or (C) except as set forth in <u>clause (v)</u> of <u>Annex 9.1(c)</u> arising by reason of any breach or alleged breach by Seller or any Seller Affiliate of any Contract or Order, in any such case to the extent that obligation or liability results from or arises out of events, facts or circumstances occurring or existing on or prior to the Closing Date, notwithstanding that the date on which any proceeding or claim related thereto is commenced or made is after the Closing;
- (vii) <u>Liability for Injury, etc.</u> Except as set forth in <u>clause (v)</u> of <u>Annex 9.1(c)</u>, any obligation or liability of Seller or any Seller Affiliate arising out of the injury or bodily harm or impairment or potential injury or bodily harm or impairment or death of any individual, or damage to or destruction of any property, whether based on negligence, breach of warranty, error or omission, strict liability, enterprise liability or any other legal or equitable theory arising from or related to activities of the Business, to the extent any of those obligations and liabilities result from or arise out of events, facts or circumstances occurring or existing on or prior to the Closing including claims and litigation disclosed in the Seller Disclosure Letter to this Agreement or otherwise disclosed to Buyer, notwithstanding whether the date on which any proceeding or Claim therefor is commenced or made before or after the Closing;

Annex 9.1(d) - 1

(viii) Excluded Asset Liabilities. Each liability, obligation or commitment that relates exclusively to, or that arises exclusively out of, any Excluded Asset, or that arises out of the distribution to, or ownership by, Seller or any of the Selling Affiliates of any Excluded Asset or associated with the realization of the benefits of any Excluded Asset, whether arising before or after the Closing Date; and

(ix) Other Liabilities. Any other obligation or liability of Seller or any Seller Affiliate not expressly assumed by Buyer under this Agreement.

Annex 9.1(d) - 2

# Exhibit A

# Form of Bill of Sale, Assignment and Assumption

(See Attached)

Exhibit A - 1

# Exhibit B

# Form of Trademark Assignment

(See Attached)

Exhibit B - 1

# Exhibit C

# Form of Transition Services Agreement

(See Attached)

Exhibit C - 1

# Exhibit D

# Form of Trademark License Agreement

(See Attached)

Exhibit D - 1

### BILL OF SALE AND ASSIGNMENT AND ASSUMPTION

WHEREAS, Buyer and Seller have entered into that certain Asset Purchase Agreement dated September [•] 2017 (the "Asset Purchase Agreement"), which is incorporated herein by reference, providing for, among other things, the purchase by Buyer of certain assets of Seller;

WHEREAS, the Asset Purchase Agreement provides, among other things, for Seller to sell, convey, transfer and assign to Buyer, and Buyer to purchase and acquire from Seller all of the Transferred Assets, free and clear of all Liens (other than Permitted Liens) and in exchange for the consideration described in the Asset Purchase Agreement including the assumption by Buyer of the Assumed Liabilities; and

**WHEREAS**, Buyer and Seller now desire to carry out the intents and purposes of the Asset Purchase Agreement by the execution and delivery by Buyer and Seller of this instrument.

NOW THEREFORE, in consideration of the foregoing and the terms and conditions of the Asset Purchase Agreement, the parties hereto agree as follows:

- 1. <u>Bill of Sale Transferred Assets</u>. Upon the terms and subject to the conditions set forth herein and in the Asset Purchase Agreement, Buyer hereby purchases and acquires from Seller, and Seller hereby sells, conveys, transfers and assigns, or will cause to be sold, conveyed, transferred and assigned, free and clear of all Liens (other than Permitted Liens), all of Seller's right, title and interest in, to and under the Transferred Assets, including the Transferred Contracts. Seller hereby covenants to deliver to Buyer such other deeds, bills of sale, endorsements, consents, assignments and other good and sufficient instruments of conveyance and assignment as Buyer and its counsel shall deem reasonably necessary or appropriate to effect, consummate, confirm or evidence the transfer to Buyer of the Transferred Assets and any other transactions contemplated thereby.
- 2. <u>Assumption of Liabilities</u>. Upon the terms and subject to the conditions set forth herein and in the Asset Purchase Agreement, effective at the Closing, Buyer hereby assumes and agrees to pay, perform and discharge as and when due the Assumed Liabilities.
- 3. Excluded Assets. Notwithstanding Paragraph 1 hereof, the Transferred Assets shall exclude the Excluded Assets and Seller shall retain all rights, title and interests to the Excluded Assets.
- 4. <u>Liabilities Not Assumed</u>. Notwithstanding <u>Paragraph 2</u> hereof, the Assumed Liabilities shall exclude Excluded Liabilities and Seller shall retain all such Excluded Liabilities.
- 5. No Third Party Beneficiaries. Nothing in this Agreement, whether expressed or implied, is intended or shall be construed to confer upon or give to any person other than the parties hereto and any Indemnified Party, any rights, remedies or other benefits under or by reason of this Agreement.
- 6. <u>Inconsistent or Conflicting Provisions</u>. If a provision of this Agreement is inconsistent or conflicts with the provisions of the Asset Purchase Agreement, the provisions of the Asset Purchase Agreement shall govern and prevail.

- 7. Counterparts. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original and all of which taken together shall constitute one and the same instrument.
- 8. <u>Governing Law</u>. This Agreement shall be governed and construed by and enforced in accordance with the internal laws of the State of New York, without giving effect to principles of conflicts of laws.

[Remainder of Page Intentionally Left Blank.]

<b>IN WITNESS WHEREOF</b> , the parties have executed this Agreement as of the day and year first above written.		
BUYER:	SELLER:	
Ву:	Ву:	
Name:	Name:	
Title:	Title:	
	Ву:	
	Name:	
	Title:	

#### TRADEMARK ASSIGNMENT

This Trademark Assignment (this "Assignment") is made effective this [ ] day of [ ], 2017, by and among Teva Women's Health, LLC, a Delaware limited liability company, and having a principal place of business at 425 Privet Road, Horsham, PA 19044, and Teva Pharmaceuticals USA, Inc., a Delaware corporation, and having a principal place of business at 425 Privet Road, Horsham, PA 19044 (together, "Assignor"), and CooperSurgical, Inc., a Delaware corporation, and having a principle place of business at 75 Corporate Drive, Trumbull, CT 06611 ("Assignee").

WHEREAS, Assignor possesses certain rights in and to the trademarks, service marks and trade names set forth on <u>Exhibit A</u> attached hereto and incorporated herein by reference (the "<u>Marks</u>");

WHEREAS, Teva Pharmaceutical Industries Ltd. ("<u>Teva</u>") and Assignee are parties to that certain Asset Purchase Agreement, dated September [•], 2017 (the "<u>Purchase Agreement</u>"), pursuant to which Teva agreed to cause Assignor to transfer, sell and convey to Assignee certain of the assets of Assignor or its affiliates, including the Marks and the goodwill of the business symbolized thereby;

WHEREAS, Assignor is an affiliate of Teva;

WHEREAS, Assignor now wishes to assign the Marks to Assignee, and Assignee is desirous of acquiring the Marks from Assignor, together with the goodwill of the business symbolized thereby; and

WHEREAS, the execution and delivery of this Assignment is a condition to Closing under the Purchase Agreement.

NOW, THEREFORE, in consideration of the premises set forth above and in the Purchase Agreement and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged:

Assignor does hereby sell, assign, convey and transfer unto Assignee and its successors, assigns and legal representatives, Assignor's entire right, title and interest in and to the Marks, together with the goodwill of the business symbolized thereby and appurtenant thereto, the same to be held and enjoyed by Assignee, its successors, permitted assigns or legal representatives, together with income, royalties, damages or payments due on or after the date hereof, including, without limitation, all claims for damages or payments by reason of infringement or unauthorized use of the Marks, along with the right to sue for past infringements and collect same for Assignee's sole use and enjoyment.

Assignor does hereby authorize the Director of the United States Patent & Trademark Office, and the empowered official of any country or countries foreign to the United States whose duty it is to record trademark registrations, applications and title thereto, to record the Marks and title thereto as the property of Assignee, its successors, assigns or legal representatives in accordance with the terms of this instrument.

Each of the parties acknowledges and agrees that neither the representations and warranties nor the rights and remedies of the parties under the Purchase Agreement shall be deemed to be enlarged, modified or altered in any way by this Assignment, and, to the extent there shall arise a conflict between any term or condition of this Assignment and any term or condition of the Purchase Agreement, the applicable term or condition of the Purchase Agreement shall control.

Assignee and Assignor also agree that multiple copies of this Assignment may be executed, each of which shall be deemed an original, and each of which shall be valid and binding upon Assignee and Assignor.

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, Assignor and Assignee have caused this Assignment to be executed as a sealed instrument by their duly authorized representatives as of the date first written above.

By: Name:	
Title:	
By:	
Name:	
Title:	
ASSIGNO	R: TEVA PHARMACEUTICALS USA, INC.
By:	
Name:	
Title:	
By:	
Name:	
Title:	
ASSIGNE	E: COOPERSURGICAL, INC.
By:	
Name:	
Title:	

 ${\it ASSIGNOR: } \textbf{TEVA'S WOMEN'S HEALTH, LLC}$ 

Signature page to Trademark Assignment

# Exhibit A

# <u>Marks</u>

COUNTRY	TRADEMARK	OWNER	APPLICATION NO.	APPLICATION DATE	REGISTRATION NO.	REGISTRATION DATE
Canada	PARAGARD	Teva's Women's Health, LLC	0612701	10-Aug-88	TMA362791	10-Nov-89
United States of America	PARAGARD	Teva's Women's Health, LLC	73710526	10-Feb-88	1505716	27-Sep-88
United States of America	PARAGARD logo B&W	Teva's Women's Health, LLC	85952706	6-Jun-13	4594032	26-Aug-14
United States of America	PARAGARD logo in Color	Teva's Women's Health, LLC	85952804	6-Jun-13	4594036	26-Aug-14
United States of America	PARAGARD PEER EXPERT EDUCATION RESOURCES in B&W	Teva's Women's Health, LLC	87274295	20-Dec-16	N/A	N/A
United States of America	PARAGARD PEER EXPERT EDUCATION RESOURCES in Color	Teva's Women's Health, LLC	87274291	20-Dec-16	N/A	N/A
United States of America	PARAGARD WING WOMAN LOGO in B&W	Teva's Women's Health, LLC	87050996	26-May-16	5132320	31-Jan-17
United States of America	PARAGARD WING WOMAN LOGO in Color	Teva's Women's Health, LLC	87051004	26-May-16	5132322	31-Jan-17
United States of America	NO HORMONES, PLEASE	Teva Pharmaceuticals USA, Inc.	87383060	23-March-17	N/A	N/A

[•], 2017

## TEVA PHARMACEUTICAL INDUSTRIES LTD

(Teva)

and

COOPERSURGICAL, INC.

(Buyer)

TRANSITION SERVICES AGREEMENT

# TABLE OF CONTENTS

<b>1.</b> D	EFIN	ITIONS	1
2.		THE SERVICES	4
	2.1	Provision of the Services	4
	2.2	Omitted Service	6
	2.3	Cooperation	7
	2.4	Facilities and Access to Facilities	7
	2.5	Security	8
	2.6	Service Quality	9
	2.7	Subcontractors	10
	2.8	No Legal, Tax or Other Advice	10
3.		CHARGES, INVOICING AND PAYMENT	10
	3.1	Payment	10
	3.2	Sales Taxes and Withholding Taxes	11
4.		FORCE MAJEURE	12
	4.1	Excusing Performance	12
	4.2	Notice of Force Majeure Event	12
	4.3	Resumption	12
5.		PERSONNEL	12
	5.1	No Secondment	12
	5.2	Application of the Transfer Regulations	12
6.		LIABILITY	13
	6.1	Exclusions from Liability	13
	6.2	Limitation of Liability	13
	6.3	Indemnities	13
	6.4	Representations	14
	6.5	Exclusion of Warranties	15
7.		TERM & TERMINATION	15
	7.1	Term of Services	15
	7.2	Termination	15
	7.3	Effect of Termination of a Service	16
	7.4	Effect of Termination of Agreement	16
	7.5	Survival of Terms	16
8.		MISCELLANEOUS	17
	8.1	Notices	17
	8.2	Assignment	18

i

18

8.3 Confidentiality

8.4	Intellectual Property Rights	19
8.5	Workstream Leads	20
8.6	TSA Steering Committee	20
8.7	Disputes	2
8.8	Governing Law	22
8.9	Jurisdiction	22
8.10	Amendments; Waivers	22
8.11	Representation by Counsel	23
8.12	Headings	23
8.13	Counterparts; Signature Pages	23
8.14	Entire Agreement	23
8.15	Severability	23
8.16	Precedence	24
8.17	Public Statements	24
8.18	Interpretation	24

SCHEDULE 1 SERVICES, SERVICE TERM, SERVICE CHARGES AND CURRENCY

SCHEDULE 2 TRANSITION PLAN
SCHEDULE 3 WORKSTREAM LEADS

#### TRANSITION SERVICES AGREEMENT

This Transition Services Agreement (this "**Agreement**") is dated as of [  $\Box$  ] 2017, by and between Teva Pharmaceutical Industries Ltd., a company organized under the laws of Israel ("**Teva**") and CooperSurgical, Inc., a company organized under the laws of the State of Delaware ("**Buyer**"). Buyer and Teva are sometimes referred to in this Agreement collectively as the "**Parties**" and individually as a "**Party**".

WHEREAS, Buyer and Teva are parties to that certain Asset Purchase Agreement, dated as of [ ] 2017 (the "Purchase Agreement"), pursuant to which Buyer has acquired certain assets related to the Business from Teva;

WHEREAS, the Parties have agreed to the provision of certain services by Teva and its Affiliates to Buyer and its Affiliates for a transitional period following the Closing in order to ensure business continuity and to facilitate the separation of the operation of the Business from the operation of the business not related to the Business (the "Retained Business"): and

WHEREAS, Teva therefore agrees to provide, or to cause the provision by its Affiliates of the Services to Buyer or its Affiliates or other Recipients on the terms of this Agreement.

**NOW, THEREFORE**, in consideration of the above and of the promises and mutual covenants, agreements, and representations contained herein and in the Purchase Agreement and intending to be legally bound, the Parties agree as follows:

- 1. **DEFINITIONS.** Capitalized terms which are not defined herein shall have the meaning given in the Purchase Agreement. The following terms shall, unless the context otherwise requires, have the following meanings, respectively:
- 1.1 "Agreement" means this Transition Services Agreement;
- 1.2 "Business Day" shall mean any day that is not (a) a Saturday, (b) a Sunday or (c) any other day on which commercial banks are authorized or required by law to be closed in the City of New York;
- 1.3 **"Buyer"** shall have the meaning given in the introductory paragraph;
- 1.4 **"Buyer SteerCo Lead"** shall have the meaning given in Clause 8.6(b);
- 1.5 **"Consent"** shall have the meaning given in Clause 2.1(e);
- 1.6 "**Disclosing Party**" shall have the meaning given in Clause 8.3(a);
- 1.7 **"Dispute"** shall have the meaning given in Clause 8.7;
- 1.8 "Effective Date" means the Closing Date;
- 1.9 **"Erroneously Transferred Data"** shall have the meaning given in Clause 2.4(b);

- 1.10 **"Extended Term"** shall have the meaning given in Clause 7.1;
- 1.11 **"Extended Term Markup Percentage"** shall have the meaning given in <u>Schedule 4(c)</u>.
- "Facilities" means such facilities and sites as are owned or controlled by either Party or such Party's Affiliates and at which Services will be provided or received pursuant to this Agreement;
- 1.13 **"Force Majeure Event"** shall have the meaning given in Clause 4.1;
- 1.14 "Handling Fee Percentage" shall have the meaning set forth in Schedule 4(a).
- 1.15 "**Indemnitee**" shall have the meaning given in Clause 6.3(d);
- 1.16 **"Indemnitor**" shall have the meaning given in Clause 6.3(d);
- 1.17 "**Information**" shall have the meaning given in Clause 8.3(a);
- 1.18 "**Notice of Dispute**" shall have the meaning given in Clause 8.7;
- "Omitted Service" means a service that is not included in Schedule 1 or is not a Transferred Service which (i) was provided by the Seller or a Seller Affiliate to the Business in the 12 month period prior to the date of the Purchase Agreement, and (ii) is necessary to enable the Buyer to conduct the Business in the manner in which it was conducted by the Seller and Seller Affiliates during that 12 month period, and (iii) is a service, or a substantially similar service, that the Buyer cannot readily provide or obtain in a reasonably timely manner on commercially reasonable terms and (iv) the Seller or a Seller Affiliate is able to provide from its own resources;
- 1.20 "Parties" and "Party" shall have the meanings given in the introductory paragraph;
- 1.21 "Person" means an individual, a corporation, a general or limited partnership, a limited liability company, an association, a trust, any other legal entity or organization or governmental authority;
- 1.22 "Personal Data" means information that can reasonably be used to identify a person, such as their name, telephone number, physical address and email address;
- 1.23 **"Proposed Change"** shall have the meaning given in Clause 8.6(d);
- 1.24 "Provider" means whichever of Teva and/or any of its Affiliates is providing the relevant Service(s);
- 1.25 "Purchase Agreement" shall have the meaning given in the introductory paragraph;

- 1.26 "**Receiving Party**" shall have the meaning given in Clause 8.3(a);
- 1.27 "Recipient" means Buyer and/or any of its Affiliates designated by Buyer and not reasonably objected to by Provider to receive the Service(s) on Buyer's behalf;
- 1.28 "**Reference Period**" means the twelve (12) months prior to Closing;
- 1.29 **"Representatives"** shall have the meaning given in Clause 8.3(c);
- 1.30 "**Required Technology**" shall have the meaning given in Clause 2.4(a);
- 1.31 "**Retained Business**" shall have the meaning given in the introductory paragraph;
- 1.32 **"Service Charge"** means the charge per Service set forth in <u>Schedule 1</u>;
- 1.33 "Service Charge Percentage" shall have the meaning set forth in Schedule 4(b).
- 1.34 **"Service Term"** means the service term specified for each Service as set out in <u>Schedule 1</u>;
- 1.35 "Services" means the Services specified in Schedule 1;
- 1.36 "Services IP" shall have the meaning given in Clause 8.4(c);
- 1.37 **"SteerCo Leads"** shall have the meaning given in Clause 8.6(b);
- 1.38 "**Term**" shall have the meaning given in Clause 7.1;
- 1.39 **"Teva"** shall have the meaning given in the introductory paragraph;
- 1.40 "Teva SteerCo Lead" shall have the meaning given in Clause 8.6(b);
- 1.41 "Third Party" means any Person other than Teva or its Affiliates, or Buyer or its Affiliates;
- 1.42 "Third Party Contract" means any agreement with a Third Party relating to, or necessary for, the provision of a Service whether entered into before or after the date of this Agreement;
- 1.43 "Third Party Supplier" means any Third Party that is a party to a Third Party Contract;
- 1.44 "Transfer Regulations" means Council Directive 2001/23/EC or local legislation implementing the Council Directive, or similar legislation regarding automatic transfer of employees in jurisdictions outside Europe;
- 1.45 "Transferred Service" means a service or related activities agreed to be transferred by Seller or a Seller Affiliate to Buyer prior to Closing;

- 1.46 "Transition Plan" means a detailed plan prepared by Buyer and mutually agreed to by the Parties as set forth <a href="Schedule 2">Schedule 2</a>. The Transition Plan shall set forth plans for the transition of the Business and each Service hereunder, including plans to achieve the timely transition of each Service, key milestones, roles and responsibilities and apportionment of costs;
- 1.47 "TSA Steering Committee" shall have the meaning given in Clause 8.6(a);
- 1.48 "Work Product IP" shall have the meaning given in Clause 8.4(b);
- 1.49 "Working Hours" means 9am to 5pm on a Business Day;
- 1.50 **"Workstream"** means each of the individual functional workstreams, namely: [□]¹; and
- 1.51 "Workstream Lead" means the responsible functional leads appointed by a Party for each Workstream as identified in Schedule 3.

#### 2. THE SERVICES

### 2.1 <u>Provision of the Services</u>.

- (a) <u>Services</u>. In consideration of the payment of the Service Charges and other amounts set forth herein, Provider shall provide or cause to be provided the Services to Recipient in accordance with and subject to the terms and conditions set forth in this Agreement.
- (b) <u>Compliance with Applicable Law, etc.</u> Each Party shall comply in all material respects at all times with all applicable Laws and requirements and directions of applicable Governmental Authorities to the extent relevant to the provision or receipt (as applicable to such Party) of a Service.
- (c) <u>Changes to Services</u>. The Parties agree and acknowledge that the scope or characteristics of the Services may change in accordance with the procedure set out in Clause 8.6(d) during the Term as Recipient completes its transition from dependence on Provider for the Services to operational independence. Without limiting the foregoing, in the event that any of the Services are reasonably required to be modified in terms of content, manner of provision or otherwise (including the Service Charges therefor) due to applicable Law or the requirements and directions of applicable Governmental Authorities, including related to the Closing or the transactions contemplated by the Purchase Agreement, the Parties shall negotiate in good faith to implement such modifications and will enter into an amendment to this Agreement, including <u>Schedule 1</u>, to reflect such modifications.

Note to Draft: Workstreams to be completed once service schedules are finalized.

- (d) Service Charges. Subject to the other provisions of this Agreement, Recipient shall pay to Provider for each Service (i) the Service Charge for that Service; (ii) all amounts payable under Third Party Contracts to the extent applicable to the provision or receipt of the Services plus a handling fee of the Handling Fee Percentage of such amounts; (iii) all other costs, expenses and other amounts incurred by Provider in obtaining any equipment, hardware, software or services as part of the provision of the Services; and (iv) all other amounts agreed to by the Parties in writing. The Parties acknowledge that the Service Charges are based on the total costs incurred by Teva in providing such Service prior to both the Closing and the separation of the Business from the Retained Business. The Parties agree that in the event that Provider's total costs of providing all of the Services in the aggregate or any Service individually over the preceding three (3) months exceeds the aggregate Service Charge for Services or the individual Service Charge for such Service, as applicable, over such period by the Service Charge Percentage or more, a Party may refer such shortfall to the TSA Steering Committee for a prompt, good faith adjustment to the relevant Service Charge on a going-forward basis, along with a one-time true-up payment from Recipient to enable Provider to recoup its shortfall relating to the time period at issue.
- (e) <u>Consents.</u> It is the understanding of the Parties that, except as set forth in <u>Schedule 2.3</u> of the Purchase Agreement, any waivers, permits, consents or similar approvals from any third parties (including Third Party Suppliers) or Governmental Authorities (any such waiver, permit, consent or similar approval being, a "Consent") that are necessary for Provider to perform or Recipient to receive the Services have been obtained prior to the Effective Date. If any Consent has not been obtained prior to the Effective Date or has been so obtained but is later terminated or revoked by such Third Party or Governmental Authority, the Parties shall work together in good faith to obtain or achieve the reinstatement of such Consent or to agree upon a commercially reasonable alternative arrangement in respect of such Service to the maximum extent possible; provided, however, that, pending obtaining or achieving reinstatement of such Consent or agreement on such alternative arrangement, Provider shall (subject to Clause 2.1(b)), for a period not to exceed three (3) months, use commercially reasonable efforts to implement reasonable and lawful arrangements designed to provide Recipient with the benefits of such Service as if the appropriate Consent had been obtained; provided, further, that any reasonable costs and expenses incurred by either Party in connection with obtaining or achieving the reinstatement of such Consent, or the implementation of any such alternative arrangement with respect to any such Consent not obtained prior to the Effective Date shall be borne by Recipient.
- (f) Third Party Contracts. Recipient agrees to comply with the terms of all Third Party Contracts. In the event Provider passes on to Recipient the benefit of any remedies (including any applicable service level credits), warranties, representations, indemnities and undertakings given in such Third Party Contract and such pass through of benefits results in any costs to Provider, such costs shall be added to the Service Charge for such Service and itemized on the relevant invoice(s) in accordance with Clause 3.1(a).

- Transition of Services. Recipient acknowledges the transitional nature of the Services. Accordingly, as promptly as practicable following the date hereof, (g) Recipient agrees to use commercially reasonable efforts to make a transition of each Service to its own internal organization or to obtain alternative third-party sources to provide the Services and in each case in accordance with the Transition Plan. Provider and Recipient each agree to use commercially reasonable efforts to take or cause to be taken, all actions, and to do, or cause to be done, all things necessary, proper or advisable to consummate and make effective such transition of Services in accordance with this Agreement. If requested by Recipient, Provider shall, for a period not to exceed three (3) months, assist, and shall use commercially reasonable efforts to cause any third-party provider of Services to assist, Recipient in connection with the transition of the performance of Services by Provider to the performance of such Services by Recipient, which may include reasonable assistance with the transfer of records, migration of historical data, the transition of any such Service from the systems of Provider to the systems of Recipient and cooperation with and assistance to any third-party consultants engaged by Recipient in connection with such transition. To the extent that, at the expiry of the Term, Recipient will require an agreement with any third-party provider used by Provider to provide Services, Provider will, for a period not to exceed three (3) months, use commercially reasonable efforts to assist Recipient to obtain terms and conditions, including pricing terms, substantially similar to those received from such service providers by Provider or an Affiliate of Provider prior to the Effective Date for Recipient's benefit, taking into account the need to minimize the cost of such migration and the disruption to the ongoing business activities of the Parties and their Affiliates; provided, however, if Recipient is unable to obtain equivalent terms and conditions, such inability shall not require Provider to provide the Services following expiry of the Term. Recipient shall also be responsible for any reasonable out-of-pocket costs incurred by Provider in rendering all such assistance.
- (h) <u>Excused Performance</u>. To the extent that Provider is prevented from or delayed in providing or causing the provision of any Service by any act or failure to act of Recipient or any Affiliate of Recipient, Provider shall promptly notify Recipient's SteerCo Lead in writing in reasonable detail and shall not be liable for such prevention from or delay in the provision of the affected Service.

### 2.2 Omitted Service.

- (a) If within sixty (60) days after the date of this Agreement, the Buyer discovers any Omitted Service, the Buyer may notify the Seller of the Omitted Service and request that the Seller provide the Omitted Service as an additional Service under this Agreement. Such notice shall include Buyer's requested duration for each such Omitted Service.
- (b) Within seven Business Days of receiving that notice, the Seller shall notify the Buyer of the amount of the Service Charge and all other amounts contemplated in Clause 2.1(d), which shall be determined as contemplated in Clause 2.1(d), for Seller or a Seller Affiliate to provide the Omitted Service and Seller's proposed duration for each such

Omitted Service. If the Buyer does not respond to the Seller's notice within five Business Days of receiving it, the Buyer shall be deemed to have elected not to receive that service. If the Buyer notifies the Seller within that five Business Day period that it wants the Seller to provide the Omitted Service, then:

- (i) Schedule 1 shall be amended to add the Omitted Service and corresponding Service Charge and other fees and Service Term; and
- (ii) the Seller shall provide the Omitted Service to the Buyer as soon as reasonably practicable.
- (c) The Buyer shall pay the Seller or a Seller Affiliate for the Omitted Service the Service Charge and other amounts for that Omitted Service determined consistent with Clause 2.1(d), including the amounts quoted by the Seller in the Seller's notice.
- 2.3 <u>Cooperation</u>. Subject to the other provisions of this Agreement, Provider and Recipient agree to reasonably cooperate with each other in all matters relating to the provision and receipt of the Services. Subject to Clause 2.4, each Party shall, and shall cause its applicable Affiliates to, make available on a timely basis to the other Party all information and materials requested by the other Party to the extent reasonably necessary for the purposes of providing and receiving the Services in accordance with this Agreement and the Purchase Agreement.
- 2.4 <u>Facilities and Access to Facilities</u>. To the extent that the performance or receipt of Services requires either Party or its Affiliates, or on their behalf designees or subcontractors not reasonably objected to by the controlling Party, to have access to the controlling Party's Facilities during the Term:
  - (a) the Party which owns or controls such Facility shall permit or shall cause its relevant Affiliate which owns or controls such Facility to permit (for the purposes of this Clause 2.3 the "controlling Party") the other Party or its relevant Affiliate, designee or subcontractor (for the purposes of this Clause 2.3 the "accessing Party") to access and use the relevant Facility, to the extent reasonably necessary to provide or receive the Services in accordance with the terms of this Agreement (for the purposes of this Clause 2.3 "Access"), during Working Hours and at such other times as are reasonably required, to the extent permitted by the controlling Party's leases, to the extent affecting or relating to the Facilities and as communicated to the accessing Party. Notwithstanding the foregoing, where required by a Governmental Authority, each Party shall give the other Party or representatives of the relevant Governmental Authority access to the relevant premises in the manner and to the extent required by the relevant Governmental Authority;
  - (b) controlling Party may reasonably refuse to admit to its Facilities any employee or other personnel of accessing Party whose admission would, at the sole discretion of controlling Party cause (or be reasonably likely to cause) (i) a security and/or health and safety risk; or (ii) the controlling Party to be in breach of applicable Law; and

(c) accessing Party shall or shall cause its Affiliates to Access such Facilities in accordance with all applicable Laws, the directions of any Governmental Authority, the terms and conditions of the controlling Party's leases affecting or relating to the Facilities as the terms of any such leases are communicated in writing or have otherwise been made accessible to the accessing Party in advance to accessing Party, controlling Party's policies and procedures as such policies and procedures are communicated in writing or have otherwise been made accessible in advance to accessing Party from time to time and controlling Party's reasonable directions from time to time (including directions relating to security and health and safety) and shall keep the Facilities in good order and not commit or permit waste or damage to the same, shall not make any changes or improvements to the Facilities without controlling Party's prior written consent and shall permit controlling Party and its employees, agents and representatives to enter the Facilities at any time and for any purpose.

### 2.5 Security.

- (a) Access to Required Technology. To the extent that the performance of Services requires Provider to have access to Recipient's internal IT systems (including third-Person services, e-mail and access to computer networks, database and equipment) owned, licensed, leased or used by Recipient ("Required Technology"), Recipient shall provide reasonable access to such Required Technology in accordance with applicable Law and subject to the security, use, virus protection, disaster recovery and confidentiality policies and procedures (including physical security, network and system access, internet security, confidentiality and processing of Personal Data and security guidelines and procedures) in effect by Recipient with respect to its own internal IT systems as of the date hereof. Provider shall, and shall cause its personnel having access to the Required Technology to (i) comply with all of Recipient's security policies that are applicable to the provision of any Service or access to any Facility, networks and systems; (ii) not tamper with, compromise or circumvent any security or audit measures employed by Recipient whose Required Technology is being accessed; and (iii) ensure that only those users who are specifically authorized by Recipient gain access to the Required Technology. The rights of access to the Required Technology granted hereunder shall be restricted to user access only, and shall not include privileged or higher level access rights or rights to functionality. Other than as specifically permitted under this Agreement, no Person shall have any rights of access to the other Party's Required Technology or internal IT systems.
- (b) <u>Erroneous Data Transfer</u>. To the extent that Provider transfers, provides, or allows Recipient to access any data (including any Personal Data, test results, R&D laboratory information assets, financial information, clinical data, business and sales data, adverse event data, customer data, new product information, manufacturing information and documentation, investigation information, batch records, human resources information, government pricing, dossier information, regulatory information or information relating to data)

that was not intended to be transferred, provided or made accessible to Recipient either due to an error by Provider or because such data was not or could not reasonably be separated or otherwise removed from any data or database to which Recipient was being provided access in order to receive the Services ("Erroneously Transferred Data"), then following the earlier of (i) the date that Recipient receives written notice of such Erroneously Transferred Data; or (ii) the date it becomes reasonably apparent to Recipient that it has been transferred, received or been provided access to such Erroneously Transferred Data, Recipient shall (A) treat such Erroneously Transferred Data as Information for purposes of Clause 8.3; (B) not use or disclose such Erroneously Transferred Data for any purpose; and (C) promptly, at Provider's request, subject to any automatic electronic archiving system, delete such Erroneously Transferred Data and/or, at Provider's discretion, provide a copy of such Erroneously Transferred Data to Provider in a form reasonably satisfactory to Provider.

#### 2.6 <u>Service Quality</u>.

- (a) Service Level. Subject to the other provisions of this Agreement, Provider shall itself or through its subcontractors use commercially reasonable efforts to provide the Services with a standard of care and skill and in a manner and quality which is in no event inferior to the manner and quality in which services equivalent to such Services were provided to the Business or performed during the Reference Period, except as may otherwise be expressly provided in Schedule 1.
- (b) Independent Contractor Status. Provider shall act under this Agreement solely as an independent contractor and not as an agent, employee or joint venture counterparty of Recipient. All employees and representatives of Provider or its subcontractors providing the Services shall be under the direction, control and supervision of Provider (and not of Recipient), and Provider shall have the sole right to exercise all authority with respect to such employees and representatives and in no event shall such employees and representatives be deemed to be employees or agents of Recipient. Neither Party shall bind, or attempt to bind, the other Party hereto to any contract or the performance of any other obligation, or represent to any third Person that it is authorized to enter into any contract or binding obligation on behalf of the other Party hereto.
- (c) <u>Uses of Services</u>. Recipient shall not resell any Services to any Person (other than an Affiliate) whatsoever or permit the use of any Services by any Person (other than an Affiliate) other than in connection with Recipient's operation of the Business in substantially the same manner as conducted during the Reference Period
- (d) No Violation of Laws. Neither Provider nor any Third Party service providers shall be required to provide all or any part of any particular Service to the extent that providing such Service, according to written advice of counsel, would require Provider to violate any applicable Laws. In such case, Provider shall use its commercially reasonable efforts to arrange for an alternative method of delivering such Service that does not violate any applicable Laws and that effectuates the intent of this Agreement with respect to such Service to the maximum extent possible under applicable Law. Recipient shall bear the costs for such alternative methods.

- 2.7 <u>Subcontractors</u>. Teva may elect to appoint one or more Affiliates or Third Parties as a subcontractor for the purpose of performing some or all of the Services hereunder.
- 2.8 <u>No Legal, Tax or Other Advice</u>. Notwithstanding anything to the contrary herein or in <u>Schedule 1</u>, Recipient acknowledges that any input or information provided by Provider in connection with the Services does not constitute accounting, tax, legal, compliance or risk management advice.

#### 3. CHARGES, INVOICING AND PAYMENT

#### 3.1 Payment.

- (a) Recipient shall pay Provider the Service Charges and all other amounts set forth in Clause 2.1(d) in accordance with this Agreement for all Services actually provided pursuant to this Agreement. Except as otherwise expressly set forth in Schedule 1 (i) Provider shall invoice Recipient or its nominated Affiliate on a monthly basis in arrears for the Service Charges and all other applicable amounts attributable to Services provided during the preceding month, with such invoice being sent by Provider to Recipient no later than fifteen (15) Business Days after the first (1st) Business Day of the month immediately following such preceding month; (ii) each such invoice shall set forth in reasonable detail a description of such Services and the Service Charge and other applicable amounts therefor and shall be payable in the currency of such Service Charge as set forth on Schedule 1, by bank transfer within thirty (30) days following the date of receipt by Recipient or Recipient's nominated Affiliate; and (iii) subject to Clause 3.1(b), any amount not paid by the thirtieth (30th) day after Recipient or Recipient's nominated Affiliate's receipt of such invoice shall be subject to late charges at the rate of One-Year LIBOR (as published in the Wall Street Journal, Eastern Edition, on the due date) plus two percent (2%) from the due date until the date on which such amount is paid. Additionally, if any undisputed amount of an invoice is not paid within ten (10) Business Days after Recipient receives an additional notice from Provider that Provider will suspend the Services if Recipient fails to pay that undisputed amount, Provider may suspend the Services until that undisputed amount and applicable late charges are paid.
- (b) Recipient may not deduct, set-off, counterclaim or otherwise withhold any amount owed to it by Provider under this Agreement, the Purchase Agreement or any of the Ancillary Agreements or other agreement (on account of any obligation owed by Provider or any Affiliate of Provider, whether or not such obligation has been finally adjudicated, settled or otherwise agreed upon in writing) against the charges payable or against the expenses owed by Recipient to Provider pursuant to this Agreement; provided, however, that notwithstanding the foregoing, in the event Recipient disputes in good faith any amount on an invoice, Recipient shall notify Provider in writing within ten (10) Business Days after Recipient's nominated Affiliate's

receipt of such invoice and shall describe in detail the reason for disputing such amount, and will be entitled to withhold such amount during the pendency of the dispute. The provisions of Clause 8.7 shall apply with respect to any disputed amount. The Parties will use good faith efforts to resolve any dispute regarding payments owed by Recipient as soon as reasonably practicable. Upon resolution of the dispute, to the extent it is resolved that Recipient still owes Provider some or all of the amount withheld, it shall promptly pay such applicable amount to Provider, together with any applicable interest payment calculated in accordance with Clause 3.1(a). Notwithstanding the foregoing, Recipient shall timely pay all undisputed portions of each invoice in the manner set forth in Clause 3.1(a).

#### 3.2 Sales Taxes and Withholding Taxes.

- (a) Any Sales Taxes will be payable by Recipient to Provider in accordance with Clause 3.1. The Service Charges do not include Sales Taxes, which will be separately stated on the relevant and valid invoice to Recipient to the extent required by applicable Law. Provider shall be responsible for the remittance of all such Sales Taxes to the applicable Tax Authority to the extent required by applicable Law. Provider or Provider's nominated Affiliate shall timely prepare and file all tax returns required to be filed with any Tax Authority with respect to such Sales Taxes and, in the case of value-added taxes, timely provide Recipient with valid value-added tax invoices, in each case to the extent required by applicable Law. Recipient shall cooperate with Provider in the preparation and filing of any such tax returns and other documentation, in accordance with applicable Law. The Parties and their respective Affiliates shall cooperate to eliminate or minimize the amount of any such Sales Taxes imposed on the transactions contemplated in this Agreement. If Recipient provides Provider with a valid and applicable exemption certificate, or other equivalent valid documentation exempting Recipient from any such Sales Taxes, reasonably satisfactory to Provider, then Provider shall to the extent permitted under applicable Law not charge Recipient for such exempted Sales Taxes. Recipient is not responsible for any penalties or interest related to the failure of Provider to collect such Sales Taxes if not included on a valid invoice or the failure to remit such Sales Taxes to the proper tax authority. If Provider receives a refund of any Sales Taxes collected from Recipient, whether in the form of cash, credit or other similar offset, Provider shall credit or refund the amount to Recipient within a reasonable period of time.
- (b) Any and all payments made hereunder, or otherwise for Services provided pursuant hereto, shall be made without deduction of, for or on account of withholding Taxes, except to the extent required by applicable Law. If any applicable Law requires the deduction or withholding of any Tax from any such payments, then Recipient or Provider, as applicable, shall be entitled to withhold the required amounts and shall timely pay such withheld amounts over to the applicable Tax Authority in accordance with the requirements of applicable Law, and, except in the case of any Sales Tax (which amount shall continue to be due and payable to Provider), any amount so withheld and paid over shall be treated as having been paid to the other Party, and Recipient or Provider, as applicable, shall not be required to pay any additional amount as a result of or in respect of such withholding; provided, that prior to making any deduction or withholding from any payment under this Agreement, Recipient or Provider, as applicable, shall provide reasonable prior written notice to Recipient or

Provider, as applicable, of the amounts subject to deduction or withholding and provide to the other Party a reasonable opportunity to furnish forms, certificates or other items that would reduce or eliminate such deduction or withholding; provided, further, that the Parties shall reasonably cooperate and assist one another to minimize any such deduction or withholding. Each Party shall provide the other Party, as applicable, in a timely manner, with documentation evidencing payment to the applicable Tax Authority of any amounts so withheld and in the form and manner required by applicable Law to allow the relevant Party to recover the amount withheld to the extent allowable by applicable Law. If Recipient or Provider, as applicable, receives a refund of any withheld taxes, whether in the form of cash, credit or other similar offset, such Party shall credit or refund the amount to the other Party within a reasonable period of time.

#### 4. FORCE MAJEURE

- 4.1 <u>Excusing Performance</u>. Neither Party shall be liable for the failure to perform its obligations under this Agreement to the extent such failure is due to events beyond the reasonable control of the non-performing Party, including fires, floods, earthquakes, hurricanes, embargoes, shortages, epidemics, quarantines, war, acts of war (whether war be declared or not), terrorist acts, insurrections, riots, civil commotion, strikes, lockouts or other labor disturbances (whether involving the workforce of the non-performing Party or of any other Person), acts of government or regulatory authorities, or acts of God (a "Force Majeure Event").
- 4.2 Notice of Force Majeure Event. A Party claiming a right to be excused from performance under Clause 4.1 shall immediately notify the other Party in writing of the extent of its inability to perform, which notice shall specify the Force Majeure Event and the estimated likely period of time during which its performance will be affected.
- 4.3 Resumption. A non-performing Party as a result of a Force Majeure Event shall use commercially reasonable efforts, at its own expense, to eliminate the Force Majeure Event and to resume performance as soon as practicable.

#### 5. PERSONNEL

- 5.1 No Secondment. For the avoidance of doubt, Provider is not under any obligation to second or procure the secondment to Recipient of any employee or other personnel in connection with the provision of the Services. The Parties agree that such individuals providing Services are employees, contract employees or secondees of the Provider or its respective Affiliates.
- 5.2 <u>Application of the Transfer Regulations</u>. The Parties acknowledge and agree that they do not expect the Transfer Regulations to apply to the commencement, provision or termination of all or any of the Services of this Agreement.

#### 6. LIABILITY

- 6.1 Exclusions from Liability. EXCEPT TO PROVIDE INDEMNIFICATION UNDER CLAUSE 6.3 WITH RESPECT TO THIRD PARTY CLAIMS, AND EXCEPT IN CASES OF FRAUD OR WILLFUL MISCONDUCT BY A PARTY OR ITS AFFILIATES OR BREACH OF CLAUSE 8.3, IN NO EVENT SHALL EITHER PARTY OR THE AFFILIATES OF SUCH PARTY BE LIABLE TO THE OTHER FOR SPECIAL, INDIRECT, MULTIPLE, PUNITIVE, EXEMPLARY, INCIDENTAL OR CONSEQUENTIAL DAMAGES, OR FOR LOST PROFITS OR REVENUES OR INCOME, BUSINESS INTERRUPTIONS, DIMINUTION IN VALUE OR LOSS OF BUSINESS OPPORTUNITY ARISING UNDER OR IN CONNECTION WITH THIS AGREEMENT OR AS A RESULT OF ANY ACTIVITIES OR NON-PERFORMANCE HEREUNDER, REGARDLESS OF WHETHER ARISING FROM BREACH OF CONTRACT, WARRANTY, TORT, STRICT LIABILITY OR OTHERWISE, EVEN IF SUCH PARTY IS ADVISED OF THE POSSIBILITY OF SUCH LOSS OR DAMAGE OR IF SUCH LOSS OR DAMAGE COULD HAVE BEEN REASONABLY FORESEEN.
- 6.2 <u>Limitation of Liability.</u> TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LAW AND EXCEPT WITH RESPECT TO A CLAIM ARISING OUT OF FRAUD OR WILLFUL MISCONDUCT OR IN THE EVENT OF A BREACH OF A PARTY'S OBLIGATIONS UNDER CLAUSE 2.4 OR 8.3 OR FOR A PARTY'S INDEMNIFICATION OBLIGATIONS UNDER CLAUSES 6.3(a)(i) AND 6.3(b)(i) (WHICH, FOR CLARITY, IS LIMITED TO THIRD PARTY LOSSES), THE AGGREGATE LIABILITY OF EITHER PARTY UNDER THIS AGREEMENT IN CONNECTION WITH A CERTAIN SERVICE SHALL BE LIMITED TO AN AMOUNT EQUAL TO THE AMOUNTS PAID OR PAYABLE TO TEVA FOR SERVICE CHARGES UNDER THIS AGREEMENT DURING THE PRECEDING TWELVE (12) MONTHS FOR SUCH SERVICE.

#### 6.3 <u>Indemnities</u>.

(a) Recipient hereby agrees, at all times during the term of this Agreement and thereafter, to fully indemnify Provider and its Affiliates, and any of its and their respective shareholders, owners, officers, directors, employees, agents and representatives against, and defend and hold such Persons harmless from, any and all Losses claimed by a Third Party arising from or in connection with (i) the gross negligence, Fraud or willful misconduct by Recipient or any of its Affiliates or any of its and their respective shareholders, owners, officers, directors, employees, agents and representatives, in each case related to the receipt of Services under this Agreement; or (ii) any breach by Recipient of any of its covenants, agreements or obligations under this Agreement, in each case of the foregoing clauses (i) and (ii) subject to Clause 6.2 and except to the extent of those Losses which Provider has an obligation to indemnify pursuant to Clause 6.3(b), as to which Losses each Party shall indemnify the other to the extent of its liability for such Losses.

- (b) Provider hereby agrees at all times during the term of this Agreement and thereafter, to fully indemnify Recipient and its Affiliates, and any of its and their respective shareholders, owners, officers, directors, employees, agents and representatives against, and defend and hold such Persons harmless from, any and all Losses claimed by a Third Party arising from or in connection with (i) the gross negligence, Fraud or willful misconduct by Provider or any of its Affiliates, and any of its and their respective shareholders, owners, officers, directors, employees, agents and representatives, in each case related to the provision of or failure to provide any Services under this Agreement; or (ii) any breach by Provider of any of its covenants, agreements or obligations under this Agreement, in each case of the foregoing clauses (i) and (ii) subject to Clause 6.2 and except to the extent of those Losses which Recipient has an obligation to indemnify pursuant to Clause 6.3(a), as to which Losses each Party shall indemnify the other to the extent of its liability for such Losses.
- (c) No right of indemnification shall exist under the Purchase Agreement for claims arising out of the performance of this Agreement, it being the intent of the Parties that such claims shall be solely governed by the provisions of this Agreement.
- (d) Either Party (the "Indemnitee") intending to claim indemnification under this Clause 6.3 shall notify the other Party (the "Indemnitor") promptly in writing of any action, claim or liability in respect of which the Indemnitee believes it is entitled to claim indemnification, provided that the failure to give timely notice to the Indemnitor shall not release the Indemnitor from any liability to the indemnitee except to the extent the Indemnitor is actually and materially prejudiced thereby. The Indemnitor shall have the right, by notice to the indemnitee, to assume the defense of any such action or claim within the thirty (30) day period after the Indemnitor's receipt of notice of any action or claim with counsel of the Indemnitor's choice and at the sole cost of the Indemnitor. If the Indemnitor so assumes such defense, the indemnitee may participate therein through counsel of its choice, but at the sole cost of the indemnitee. The Party not assuming the defense of any such claim shall render all reasonable assistance to the Party assuming such defense, and all reasonable out-of-pocket costs of such assistance shall be for the account of the Indemnitor. No such claim shall be settled other than by the Party conducting the defense thereof, and then only with the consent of the other Party (not to be unreasonably withheld, delayed or conditioned); provided that the Indemnitee shall have no obligation to consent to any settlement of any such action or claim which imposes on the Indemnitor, and the Indemnitee shall have no right to withhold its consent to any settlement of any such action or claim to the payment of money by the Indemnitor or its insurer and includes a full release of claims with respect to the Indemnitee.
- Representations. Each Party represents and undertakes to the other Party that as of the Effective Date (a) it has the capacity and authority to enter into this Agreement; (b) the Persons entering into this Agreement on its behalf have been duly authorized to do so; and (c) this Agreement and the obligations created hereunder are binding upon it and enforceable against it in accordance with their terms (subject to applicable principles of equity) and do not and will not violate the terms of any other agreement, or any judgment or court order, to which it is bound.

6.5 <u>Exclusion of Warranties</u>. EXCEPT AS EXPRESSLY SET FORTH HEREIN, NEITHER PARTY MAKES ANY REPRESENTATIONS, STATEMENTS, COVENANTS OR WARRANTIES WITH RESPECT TO THE SERVICES, WHETHER EXPRESS OR IMPLIED, AND ALL IMPLIED WARRANTIES, INCLUDING THOSE RELATING TO MERCHANTABILITY AND FITNESS FOR A PARTICULAR PURPOSE, ARE HEREBY DISCLAIMED.

#### 7. TERM & TERMINATION

- 7.1 Term of Services. This Agreement and the provision of Services shall commence on the Effective Date and each Service shall terminate (subject to any earlier termination pursuant to Clause 7.2) automatically at 11:59 p.m. New York City time, on the last day of the applicable Service Term (the period between the Effective Date and the last to expire of the Service Terms and Extended Terms being the "Term"). If Recipient wishes to extend any Service Term, it shall provide written notice of such request to the TSA Steering Committee no later than thirty (30) days prior to the otherwise applicable expiration date for such Service Term, which request shall be considered by the TSA Steering Committee in accordance with Clause 8.6 (an "Extended Term"); provided that, for any Extended Term, Teva may increase the Service Charge for such Service by the Extended Term Markup Percentage. This Agreement shall expire at the end of the Term.
- 7.2 Termination. Except as otherwise expressly set forth in Schedule 1, this Agreement may be terminated as follows (a) Recipient may, without penalty, in its sole discretion terminate or reduce the scope of any Service or terminate this Agreement as a whole with or without cause, upon thirty (30) days' prior written notice of termination or reduction; (b) Provider may terminate, without prejudice to any other rights or remedies Provider may have, any one or more Services or this Agreement if Recipient fails to pay any undisputed invoices it owes under this Agreement when due and ten (10) days have elapsed following written notice from Provider of such non-payment without cure thereof; and (c) either Party may immediately terminate, without prejudice to any other rights or remedies such Party may have, any one or more Services or this Agreement if any of the following occurs (i) the other Party commits a material breach of this Agreement (except to the extent excused due to a Force Majeure Event pursuant to Clause 4, as applicable), and such breach is either not capable of being cured or is not cured within thirty (30) days after notice of breach is given by the non-breaching Party; (ii) if the other Party commences a voluntary case or other proceeding seeking liquidation, reorganization or other relief with respect to itself or its debts under any bankruptcy, insolvency or other similar Law now or hereafter in effect or seeks the appointment of a trustee, receiver, liquidator, custodian or other similar official of it or any substantial part of its property, or consents to any such relief or to the appointment of or taking possession by any such official in an involuntary case or other proceeding commenced against it, or makes a general assignment for the benefit of creditors or takes any corporate action to authorize any of the foregoing; or (iii) if any event equivalent or similar to any event in (ii) occurs under the Law of any jurisdiction.

- 7.3 Effect of Termination of a Service. Upon any termination or reduction of any Service pursuant to Clause 7.2 and (a) subject to any rights or obligations which have accrued prior to termination, neither Party shall have any further obligation to the other Party in respect of the part or parts of this Agreement which have been terminated; (b) any licenses or sub-licenses granted under Clause 8.4(d) shall immediately cease in relation to the terminated Services; (c) Provider shall promptly return to Recipient any of Recipient's equipment and materials containing Recipient's Information that are in Provider's equipment and, subject to any automatic electronic archiving system, materials (including any documentation) containing Provider's Information that are in Recipient's possession or control and that are not required for use in connection with any non-terminated Services; and (e) Recipient's obligation to pay Provider for such Service will cease or reduce commensurately, provided that Recipient shall pay for any non-cancellable costs of third parties engaged by Provider or other costs reasonably incurred by Provider as a result of the termination or reduction of the relevant Service prior to the expiry of any Service Term or Extended Term (provided that Provider shall use commercially reasonable efforts to minimize the existence and amount of any such liability or costs) in connection with such Service and in accordance with this Agreement. Except as set forth in this Clause 7.3, no such termination or reduction of any Service will in any way affect Provider's obligation to provide or make available any other Service shall not relieve Provider from any obligation hereunder to provide (or if applicable, to cause the provision of) any remaining Services.
- 7.4 <u>Effect of Termination of Agreement</u>. In the event of a termination of this Agreement, Provider shall be entitled to all undisputed outstanding amounts due from Recipient for the provision of Services rendered prior to the date of termination and, upon resolution of any disputed amounts in accordance with Clause 8.7 in favor of Provider, such disputed amounts.
- 5.5 Survival of Terms. The following terms will survive the termination or expiry of this Agreement for any reason (together with any other term that by its nature is intended to survive termination): Clauses 2.5(b) (Erroneous Data Transfer), 3.2 (Sales Taxes and Withholding Taxes), 6 (Liability), 7 (Term and Termination), 8.1 (Notices), 8.3 (Confidentiality), 8.4(a) (Retained Rights), 8.4(b) (Work Product IP), 8.4(c) (Services IP), 8.5 (Workstream Leads), 8.7 (Disputes), 8.8 (Governing Law), 8.9 (Jurisdiction), 8.12 (Headings), 8.10 (Amendments, Waivers), 8.14 (Entire Agreement), 8.15 (Severability), 8.16 (Precedence), 8.17 (Public Statements) and 8.18 (Interpretation).

#### 8. MISCELLANEOUS

8.1 Notices. All notices, requests and other communications to any Party hereunder shall be in writing (including electronic transmission) and shall be given:

#### If to Teva

Teva Pharmaceutical Industries Ltd.

425 Privet Road

Horsham, PA 19044

Attention: Ira Schreger, Sr. Vice President and General Counsel, Transactions

Telephone: 215-591-3000

E-mail: ira.schreger@tevapharm.com

with a copy (which shall not constitute notice) to:

Goodwin Procter LLP 100 Northern Avenue Boston, MA 02210

Attention: Kingsley Taft, H. David Henken and Gregg Katz Telephone: (617) 570-1222, (617) 570-1672 and (617) 570-1406

E-mail: ktaft@goodwinlaw.com, dhenken@goodwinlaw.com and gkatz@goodwinlaw.com

#### If to Buyer:

CooperSurgical, Inc. 75 Corporate Drive Trumbull, CT 06611

Attention: Robert D. Auerbach, M.D., FACOG, Executive Vice President, Chief Strategy Officer and Chief Medical Officer and Graceann Pisano, Corporate

Counsel

Telephone: 203 601 5200

E-mail: robert. auerbach@coopersurgical.com, graceann.pisano@coopersurgical.com and Rachel.augustine@coopersurgical.com

with a copy to (which shall not constitute notice):

Carter Ledyard & Milburn LLP

2 Wall Street

New York, NY 10005

Attention: David I. Karabell (212 238 8852) and Bryan Hall (212 238 8894)

Telephone: 212 732 3200

E-mail: karabell@clm.com and hall@clm.com

or such other address or e-mail address as such Party may hereafter specify for the purpose by notice to the other parties hereto. All such notices, requests and other communications shall be deemed received on the date of receipt by the recipient thereof if received prior to 5:00 p.m. on a Business Day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed to have been received on the next succeeding Business Day in the place of receipt.

8.2 <u>Assignment.</u> Neither Party may assign this Agreement or any of its rights or obligations hereunder except with the written consent of the other Party. Notwithstanding the foregoing provisions of this Clause 8.2, Buyer may assign this Agreement without Teva's written consent to an Affiliate and Teva may assign this Agreement without Buyer's prior written consent (a) in whole or on a Service-by-Service or Product-by-Product or Product-by-country basis to any of its Affiliates; or (b) in whole to a buyer of all or substantially all of its assets to which this Agreement relates (whether by merger, stock sale, asset sale or otherwise).

#### 8.3 Confidentiality.

- Prior to this Agreement and in the course of the provision or receipt of Services, each Party may have disclosed or may need to disclose or make accessible to the other (for purposes of this Clause 8.3, the Party disclosing or making accessible such information shall hereinafter be referred to as the "Disclosing Party", and the Party receiving such information shall hereinafter be referred to as the "Receiving Party") certain information which is either non-public, confidential or proprietary in nature (collectively, the "Information"); provided, that for the purposes of this Agreement, Information shall not include information that (i) becomes generally available to the public other than as a result of a disclosure by the Receiving Party or its Affiliates or contractors and through no violation of this Agreement or the Purchase Agreement; (ii) the Receiving Party can establish is available or known to the Receiving Party as of the date of this Agreement without an obligation to keep such information confidential; (iii) becomes lawfully available to the Receiving Party from a source other than the Disclosing Party and through no violation of this Agreement or the Purchase Agreement, provided, that such source is not known by the Receiving Party to be bound by a confidentiality agreement with or other obligation of secrecy to the Disclosing Party; or (iv) the Receiving Party can establish is independently developed by the Receiving Party without use of, or reference to, any other Information.
- (b) The Receiving Party shall hold in confidence and not disclose to any Third Party any Information received by it in connection with this Agreement from the Disclosing Party, and it shall use all Information received by it in connection with this Agreement from the Disclosing Party solely as necessary for the provision or receipt of the Services (and for no other purpose whatsoever) and it shall take the same degree of care with the Disclosing Party's Information as it uses with its own, but in no event less than a reasonable degree of care; provided, that Provider may disclose the Information (i) to those of its Affiliates or to third parties that provide Services or to any directors, members, officers, employees, agents and advisors (including, without limitation, attorneys, accountants, consultants and service providers) of any such Person, but only to the extent necessary for such Affiliates, third parties, directors, members, officers, employees, agents and advisors to carry out the Services or to enforce the rights of a Party hereunder, and provided they are bound by confidentiality obligations at least as stringent as those set forth herein; and (ii) to the extent required to be disclosed by Law or the rules of an applicable securities exchange; provided further, that in the

case of any disclosures made pursuant to sub-Clause (ii), to the extent legally permitted (as determined on the advice of counsel) and not otherwise prohibited by applicable Law, the Receiving Party shall, and shall direct its representatives to, promptly notify the Disclosing Party in writing so that the Disclosing Party may seek a protective order or other appropriate remedy or, in its sole discretion, waive compliance with this Clause 8.3. In the event no such protective order or other remedy is obtained, or that the Disclosing Party waives compliance with this Clause 8.3, the Receiving Party shall, without liability hereunder, furnish only that portion of the Information which is legally required and shall request assurance that the Information will be accorded confidential treatment.

(c) The Receiving Party shall inform any and all of its Affiliates, officers, directors, controlling persons, partners, employees, agents or representatives ("Representatives") that receive Information of the Disclosing Party of the confidential and proprietary nature of such Information and shall direct such Representatives to treat such Information as strictly proprietary and confidential. The Receiving Party shall establish procedures to ensure that the Information is properly protected and monitored for purposes of complying with this Clause 8.3. The Receiving Party agrees to be responsible for any breach of this Clause 8.3 by any of its Representatives.

#### 8.4 <u>Intellectual Property Rights</u>.

- (a) Retained Rights. Except as otherwise expressly provided in this Agreement, each of the Parties and their respective Affiliates shall retain all right, title and interest in and to (i) their respective IP Rights created, developed, written or authored by a Party or its Affiliate prior to the Effective Date; and (ii) any IP Right which is independently created, developed, written or authored by a Party or its Affiliate other than under this Agreement. Except as set forth in Clauses 8.4(a) to 8.4(d), this Agreement confers no right, license or interest whether directly or indirectly or by implication, estoppel or otherwise under any IP Rights of either Party.
- (b) Work Product IP. Subject to Clause 8.4(a) and the limited license granted in Clause 8.4(d), any IP Right created, developed, written or authored by Provider as a deliverable for a Service under this Agreement ("Work Product IP"), all right, title and interest throughout the world in and to all such IP Right shall be deemed to be a work-made-for-hire for Recipient and shall vest solely in Recipient unconditionally and immediately upon such IP Right having been created, developed, written or authored, unless the Parties agree otherwise in writing. To the extent title to any IP Right that is the subject of this Clause 8.4(b) vests, by operation of Law, in Provider, Provider hereby assigns, and causes the assignment by its Affiliate, to Recipient or its designated Affiliate all right, title and interest in such IP Right and agrees to provide such assistance and execute such documents as Recipient may reasonably request to vest in such Party all right, title and interest in such IP Right.

- (c) Services IP. Subject to Clause 8.4(a) and the limited license granted in Clause 8.4(d), in the event that any other IP Right is created, developed, written or authored by Provider or its Affiliate in connection with the performance of the Services and which is not intended to be a deliverable for a Service under this Agreement ("Services IP"), all right, title and interest throughout the world in and to all such IP Right shall vest solely in such Provider or its Affiliate unconditionally and immediately upon such IP Right having been created, developed, written or authored.
- (d) Grant of Rights. Buyer, on behalf of itself and its Affiliates, hereby grants to Teva and its Affiliates a non-exclusive, non-transferable (other than pursuant to Clause 8.2), non-sublicensable (except to its Affiliates and subcontractors for the purposes of providing the Services), royalty-free, worldwide license during the term of this Agreement to use all IP Rights owned or controlled by Buyer or its Affiliates (including all Transferred IP acquired by Buyer) to the extent necessary for Teva and its Affiliates to provide the applicable Service under this Agreement.
- 8.5 <u>Workstream Leads</u>. The initial points of contact for Provider and Recipient, with respect to any matters of day-to-day provisions of the Services, including attempting to resolve any issues that may arise during the performance of the Services, shall be the Workstream Leads for the relevant Service within such Workstream, who shall have the authority to handle such daily operational matters related to the Services within their Workstream.

#### 8.6 TSA Steering Committee.

- (a) Promptly following the Effective Date the Parties shall establish a steering committee (the "TSA Steering Committee"), with the primary objectives of (i) maintaining sufficient oversight over the provision of the Services under this Agreement and the standards to which such Services are provided; (ii) assuming overall responsibility for the handling of the effective transition of the Services to a relevant Party; (iii) keeping performance of the Services under continual review and resolving any issues with the performance of the Services as they arise; (iv) keeping the nature and amount of the Services under continual review and identifying any changes to Services that may be required by Recipient in accordance with Clause 2.1 and any Services which are no longer required by Recipient; and (v) undertaking such other tasks in relation to the Services as the Parties may from time to time agree.
- (b) The TSA Steering Committee shall comprise of a lead individual from each Party who shall represent each Party's interests in relation to the Services (each the "Teva SteerCo Lead" and the "Buyer SteerCo Lead" respectively, and together the "SteerCo Leads") and select members, as appropriate, from the Workstream Leads as set out in <a href="Schedule 3">Schedule 3</a>. As at the Effective Date, the Teva SteerCo Lead and Buyer SteerCo Lead are as follows:

 Teva SteerCo Lead
 Buyer SteerCo Lead

 □
 □

- The Teva SteerCo Lead and Buyer SteerCo Lead shall coordinate meetings of the TSA Steering Committee from time to time. The TSA Steering Committee shall meet on a monthly basis in such location as is reasonably convenient to each Party (including, for the avoidance of doubt, via audio or video conference) or otherwise as agreed between the Teva SteerCo Lead and Buyer SteerCo Lead. The first TSA Steering Committee meeting shall occur within two (2) weeks after the Effective Date and shall include a full review of the Services and the requirements of the Parties. The SteerCo Leads will escalate any proposed changes to the Services, the Service Charges or Service Term to the TSA Steering Committee for agreement by the TSA Steering Committee. The TSA Steering Committee shall endeavor to make all decisions by unanimous decision, provided that a decision between the Teva SteerCo Lead and the Buyer SteerCo Lead shall constitute a decision of the TSA Steering Committee. Either Party may from time to time and at its sole discretion, replace any one or more of its representatives on the TSA Steering Committee (including the Teva SteerCo Lead, the Buyer SteerCo Lead and/or the Workstream Leads) with an appropriate individual, by giving notice of replacement to the other Party, provided that the Parties shall use this right sparingly so as not to disrupt the work of the TSA Steering Committee.
- (d) Where Recipient wishes to: (i) increase the Service Term of existing Services; or (ii) increase the amount or scope of existing Services as permitted in accordance with Clause 2.1(c) (each a "**Proposed Change**"), it shall notify the Provider's SteerCo Lead of its request and such Proposed Change shall be discussed at the next convened TSA Steering Committee. If the TSA Steering Committee agrees that a Proposed Change is reasonably required to the Services, the TSA Steering Committee will direct any Proposed Change along with any proposed changes to the relevant Service Charges or Service Term to the SteerCo Leads for implementation.
- 8.7 <u>Disputes.</u> Other than with respect to any claim covered by Clause 6.3, any dispute, controversy or claim between the Parties arising out of, relating to or in connection with the Services, this Agreement, or the breach, termination, or validity thereof (a "**Dispute**"), shall be resolved by submitting such Dispute in writing first to the relevant Workstream Lead of each Party, and such Workstream Leads shall seek to resolve such Dispute through informal good faith negotiation. In the event that such Dispute is not resolved by the relevant Workstream Leads within five (5) Business Days after the submission of such Dispute to such Workstream Leads, the relevant Workstream Leads shall escalate such Dispute to the Teva SteerCo Lead and the Buyer SteerCo Lead for resolution. In the event that such Dispute is not resolved by the Teva SteerCo Lead and Buyer SteerCo Lead within five (5) Business Days of such escalation (or such longer time as the SteerCo Leads may agree), then the claiming Party may provide the other Party with a written notice (a "**Notice of Dispute**"), describing (a) the issues in dispute and such Party's position thereon; (b) a summary of the evidence and arguments supporting such Party's

positions; (c) a summary of the negotiations that have taken place to date; and (d) the name and title of the senior executives (which, in the case of Teva, will be a member of Teva's executive committee, and, in the case of Buyer, will be a senior executive of Buyer (i.e., direct report to the Chief Executive Officer) or their respective designees who will represent each Party. The senior executives or their respective designees designated in such Notice of Dispute shall meet in person or by telephone as often as reasonably necessary to resolve the Dispute and shall confer in a good faith effort to resolve the Dispute. In the event that such Dispute is not resolved by such senior executives or their respective designees within five (5) Business Days after receipt of the Notice of Dispute (or such longer time as such senior executives or their respective designees may agree), then either Party may pursue any remedy at law, equity, or otherwise.

- 8.8 Governing Law. This Agreement (and any claim or controversy arising out of or relating to this Agreement) shall be governed by and construed in accordance with the laws of the State of New York without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any jurisdiction other than the State of New York. Neither this Agreement nor any right or obligation of any of the Parties under this Agreement shall be governed by the U.N. Convention on Contracts for the International Sale of Goods, and the Parties expressly waive or disclaim, as the case may be, any right or obligation they may have under this Agreement pursuant to the U.N. Convention on Contracts for the International Sale of Goods.
- 8.9 <u>Jurisdiction</u>. Each Party hereto hereby irrevocably and unconditionally submits, for itself and its property, to the non-exclusive jurisdiction of any New York State court, or Federal court of the United States of America, sitting in Manhattan, New York City, New York, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement delivered in connection herewith or the transactions contemplated hereby or thereby or for recognition or enforcement of any judgment relating thereto, and each Party hereto hereby irrevocably and unconditionally (a) agrees not to commence any such action or proceeding except in such courts; (b) agrees that any claim in respect of any such action or proceeding may be heard and determined in such New York State court or, to the extent permitted by applicable Law, in such Federal court; (c) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such New York State or Federal court and (d) waives, to the fullest extent permitted by applicable Law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such New York State or Federal court. Each Party hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable Law. Each Party hereto irrevocably consents to service of process in the manner provided for notices in Clause 8.1. Nothing in this Agreement will affect the right of any Party hereto to serve process in any other manner permitted by applicable Law.
- 8.10 <u>Amendments; Waivers.</u> No amendment, modification or discharge of this Agreement, and no waiver hereunder, shall be valid or binding unless set forth in writing and duly executed by the Party against whom enforcement of the amendment, modification, discharge or waiver is sought. Any such waiver shall constitute a waiver only with

respect to the specific matter described in such writing and shall in no way impair the rights of the Party granting such waiver in any other respect or at any other time. Neither the waiver by any of the Parties hereto of a breach of or a default under any of the provisions of this Agreement, nor the failure by any of the Parties, on one or more occasions, to enforce any of the provisions of this Agreement or to exercise any right or privilege hereunder, shall be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. The rights and remedies herein provided are cumulative and none is exclusive of any other, or of any rights or remedies that any Party may otherwise have at law or in equity.

- 8.11 Representation by Counsel. Each Party represents and agrees with each other that it has been represented by or had the opportunity to be represented by, independent counsel of its own choosing, and that it has had the full right and opportunity to consult with its respective attorney(s), that to the extent, if any, that it desired, it availed itself of this right and opportunity, that it or its authorized officers (as the case may be) have carefully read and fully understand this Agreement and the Ancillary Agreements in their entirety and have had them fully explained to them by such Party's respective counsel, that each is fully aware of the contents thereof and their meaning, intent and legal effect, and that it or its authorized officer (as the case may be) is competent to execute this Agreement and has executed this Agreement free from coercion, duress or undue influence.
- 8.12 <u>Headings</u>. The headings of the sections and subsections of this Agreement are inserted for convenience only and shall not be deemed to constitute a part hereof.
- 8.13 <u>Counterparts; Signature Pages</u>. The Parties may execute this Agreement in one or more counterparts, each of which will be deemed an original and all of which, when taken together, will be deemed to constitute one and the same agreement. Any signature page hereto delivered by facsimile machine or by e-mail (including in portable document format (pdf), as a joint photographic experts group (jpg) file, or otherwise) shall be binding to the same extent as an original signature page, with regard to any agreement subject to the terms hereof or any amendment thereto and may be used in lieu of the original signatures for all purposes. Any Party that delivers such a signature page agrees to later deliver an original counterpart to any Party that requests it.
- 8.14 <u>Entire Agreement</u>. Without limiting the rights and obligations of the Parties under the Purchase Agreement and the other Ancillary Agreements, this Agreement and all schedules hereto and thereto, constitutes the full, complete, final and integrated agreement between the Parties relating to the subject matter hereof and supersedes all previous written or oral negotiations, commitments, agreements, transactions, or understandings with respect to the subject matter hereof.
- 8.15 <u>Severability</u>. If any provision, including any phrase, sentence, clause, section or subsection, of this Agreement is determined by a court of competent jurisdiction to be invalid, inoperative or unenforceable for any reason, such circumstances shall not have the effect of rendering such provision in question invalid, inoperative or unenforceable in any other case or circumstance, or of rendering any other provision herein contained invalid, inoperative or unenforceable to any extent whatsoever.

  Upon any such determination, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby are consummated as originally contemplated to the fullest extent possible.

- 8.16 <u>Precedence</u>. In the event of a conflict between a provision of this Agreement and a provision of the Purchase Agreement or an Ancillary Agreement, or a provision of a schedule hereto, the terms of this Agreement shall take precedence.
- 8.17 <u>Public Statements</u>. No party to this Agreement shall originate any publicity, news release or other similar public announcement, written or oral, whether relating to this Agreement or any documents or transactions contemplated hereby or the existence of any arrangement between the Parties, without the prior written consent of the other Party whether or not named in such publicity, news release or other similar public announcement, except to the extent permitted under the Purchase Agreement.
- 8.18 Interpretation. Unless expressly specified otherwise, whenever used in this Agreement, the terms "Clause" and "Schedule" refer to clauses and schedules of this Agreement. Whenever used in this Agreement, the terms "hereby," "hereof," "herein" and "hereunder" and words of similar import refer to this Agreement as a whole, including all articles, sections, schedules and exhibits hereto. Whenever used in this Agreement, the terms "include," "includes," and "including" mean "include, without limitation," "includes, without limitation," and "including, without limitation," respectively. Whenever the context of this Agreement permits, the masculine, feminine or neuter gender, and the singular or plural number, are each deemed to include the others. "Days" means calendar days unless otherwise specified. References in this Agreement to a particular Law or sections of a Law shall be deemed to refer any amended or replacement legislation thereto and to such sections or provisions as they may be amended after the date of this Agreement. The Parties have participated jointly in the negotiation and drafting of this Agreement and in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party (or any Affiliate thereof) by virtue of the authorship of any of the provisions of this Agreement.

[Signature Page Follows]

TEVA PHARMACEUTICAL INDUSTRIES LTD.	
By:	
Name:	=
Title:	
By:	
Name:	_
Title:	
COOPERSURGICAL, INC.	
By:	
Name:	-
Title:	

The Parties have caused this Agreement to be executed by their respective duly authorized officers as of the date first above written.

(Signature Page to Transition Services Agreement)

#### FORM OF TRADEMARK TRANSITION LICENSE AGREEMENT

This Agreement (this "Agreement") is made and entered into as of [ ], [2017] (the "Effective Date") by and among Teva Women's Health, LLC, a Delaware limited liability company, and Teva Pharmaceuticals USA, Inc., a Delaware corporation, (together, "Licensor") and CooperSurgical, Inc., a Delaware corporation ("Licensee").

#### RECITALS

WHEREAS, Teva Pharmaceutical Industries Ltd. ("<u>Teva</u>") and Licensee are parties to that certain Asset Purchase Agreement, dated September [•], 2017(the "<u>Purchase Agreement</u>"), pursuant to which Teva and its affiliates sold certain assets to Licensee related to the Products and Business (each as defined below);

WHEREAS, Licensor is an affiliate of Teva;

WHEREAS, prior to the Closing of the transactions under the Purchase Agreement, Licensor and its affiliates used the Licensed Marks (as defined below) in connection with the operation of the Business;

WHEREAS, Licensor desires to grant to Licensee a limited right and license to use and display the Licensed Marks in connection with the orderly transition of the Business from Teva and its affiliates to Licensee, and the orderly discontinuance of the use of the Licensed Marks in connection with the Business; and

WHEREAS, the execution and delivery of this Agreement is a condition to Closing under the Purchase Agreement.

#### AGREEMENT

In consideration of the premises above and in the Purchase Agreement, the mutual covenants and considerations set forth hereinafter and for other good and valuable consideration, the receipt and sufficiency of which is hereby acknowledged and confirmed, the parties hereto agree as follows:

**Section 1. Definitions.** Capitalized terms which are not defined herein shall have the meaning given in the Purchase Agreement. The following terms shall, unless the context otherwise requires, have the following meanings, respectively:

"Affiliate" shall have the meaning set forth in the Purchase Agreement.

"Business" shall have the meaning set forth in the Purchase Agreement.

"License" means the license granted under Section 2.1.

"Licensed Marks" means the trademark registrations, applications for trademark registration, and Internet domain name registrations, trademarks, service marks, names, symbols, designations listed on Exhibit A and any derivations thereof in use by Licensor or any of its

Affiliates in the Business during the twelve (12) month period prior to the Effective Date, together with any designs and/or logos in use by Licensor or its Affiliates in association with such trademarks in the Business during the twelve (12) month period prior to the Effective Date.

"Products" means the products set forth in Schedule 1.3 of the Purchase Agreement.

"Territory" means the United States.

"Term" means the term of this Agreement as set forth in Section 6.

"Trademark Guidelines" means the guidelines, standards and other requirements relating to use of the Licensed Marks communicated in writing to Licensee from time to time by Licensor, which revisions shall become effective upon publication or delivery of notice thereof to Licensee.

#### Section 2. License

- 2.1. Grant. Subject to the terms, conditions and provisions of this Agreement, Licensor hereby grants to Licensee a non-exclusive, nontransferable (except as provided in Section 7.2), nonsublicensable (except as provided in Section 2.2), royalty-free, fully paid-up, license in the Territory to use and display the Licensed Marks in connection with the operation of the Business during the Term, but only to the extent (a) the Licensed Marks are used in or displayed on the packaging, labeling or marketing materials printed on or immediately prior to the Effective Date or printed after the Effective Date to meet the requirements of the Business in the 6 months after the Effective Date, or electronic media, including websites, used immediately prior to the Effective Date, in connection with the Products, or (b) as necessary to receive the services under the Transition Services Agreement, in each case, in substantially the same manner in which they are in use by Licensor or its Affiliates in connection with the operation of the Business on or immediately prior to the Effective Date. For the avoidance of doubt, Licensee may use the Licensed Marks in connection or in combination with one or more other trademarks as part of the transition from using the Licensed Marks to such other marks designated by Licensee to identify the goods and service of the Business; provided that Licensee otherwise has the right (e.g., by ownership or license of such other trademarks) to legally use such other trademarks in connection or in combination with the Licensed Marks.
- 2.2. Limited Right to Sublicense. Licensee may sublicense any of the rights under the License to any of its Affiliates or to a third party upon prior written consent of Licensor (which consent shall not be unreasonably withheld), provided that the sublicense shall be set forth in a written agreement including, without limitation, specific acknowledgement that: (a) the sublicense is subject to all of the applicable terms, conditions and provisions of this Agreement; (b) in its exercise of the sublicense, the sublicensee shall comply with all of Licensee's applicable obligations under this Agreement; (c) the sublicensee has no right to grant any other person any sublicense any right under the License; and (d) the sublicensee shall automatically terminate (e.g., without any notice or other action by Licensee or Licensor) upon the earlier of the end of the Term or, if applicable, if and when the sublicensee ceases to be an Affiliate of Licensee. Upon granting any sublicense, Licensee shall give Licensor written notice thereof (including, without limitation, the name and principal place of business of the sublicensee and

the name, address and telephone number of the employee or other representative of the sublicensee responsible for the sublicensee's administration of and compliance with the sublicense agreement). Upon Licensor's request, Licensee shall deliver to Licensor a true, correct and complete copy of any sublicense agreement (including, without limitation, any and all amendments, supplements and modifications thereof), provided that Licensee may redact any financial terms from such sublicense agreement.

- 2.3. Proper Use of Licensed Marks. Licensee shall use the Licensed Marks only for the purpose set forth in Section 2.1, in substantially the same manner in which they were in use by Licensor and its Affiliates in connection with their operation of the Business on or immediately prior to the Effective Date and in accordance with this Agreement and the Trademark Guidelines in all material respects. Licensee shall ensure that any use of the Licensed Marks pursuant to the Licensee includes the trademark markings and notices (e.g., TM or ®) to the extent required and set forth in the Trademark Guidelines to preserve all of Licensor's rights (including, without limitation, all remedies for infringement) and interests with respect to the Licensed Marks. Licensee shall ensure that all Products offered by Licensee or any of its Affiliates under any of the Licensed Marks shall be of equal or better quality than the same Products or substantially similar products and services offered by Licensor and its Affiliates in the Business as conducted on the Effective Date. Licensee shall not use any of the Licensed Marks: (a) for any purpose other than as set forth in Section 2.1, including without limitation in connection with the marketing, sale or promotion of any products or services other than the Products; (b) in any manner that disparages Licensor, any of Licensor's Affiliates or any products or services offered by Licensor or any of its Affiliates; (c) in any manner that injures the goodwill associated with any of the Licensed Marks; or (d) in any manner that violates any law, regulation, rule, order or other requirement of any governmental authority having jurisdiction. Upon notice of or discovery of any noncompliance with the requirements of this Section 2.3, Licensee shall promptly notify Licensor of the noncompliance, remedy the noncompliance and notify Licensor when remedial action is taken.
- **2.4. Wind Down.** Licensee shall take commercially reasonable steps to wind down the use of the Licensed Marks in connection with the operation of the Business during the Term and completely discontinue use of the Licensed Marks by Licensee and its Affiliates as of the end of the Term.
- 2.5. Audit. If Licensor reasonably believes that Licensee's or Licensee's Affiliates use of the Licensed Marks is not in compliance with the requirements set forth in Section 2, then Licensor shall have the right to audit such uses of the Licensed Marks by Licensee and its Affiliates. Upon Licensor's reasonable advance notice, Licensee shall provide Licensor access from time to time during normal business hours to records and other documentation evidencing such use of the Licensed Marks by Licensee and its Affiliates (including, without limitation, all applicable advertising and promotional materials bearing such Licensed Marks); provided, however, that in no event shall Licensor exercise the rights granted under this Section 2.5 in a manner that interferes with the conduct of the Business in any material respect.
- **2.6. Nominative; Fair Use.** For the avoidance of doubt, nothing in this Agreement shall prevent Licensee and its Affiliates from, during the Term and after, using the Licensed Marks solely (a) in a non-trademark manner to describe the history of the Business, including Licensee's acquisition of the Business; (b) as required by applicable law; or (c) in a manner consistent with "trademark fair use" to the extent no consumer confusion is likely to result from such use.

#### Section 3. Ownership, Perfection, Maintenance and Enforcement

- 3.1. Ownership. Except as specifically set forth in the License, Licensor reserves all rights in the Licensed Marks. No additional rights are granted by implication, estoppel or otherwise. No title to or ownership of any of the Licensed Marks is transferred to Licensee under this Agreement or by Licensee's use of the Licensed Marks. All goodwill associated with the Licensed Marks shall inure to and belong exclusively to Licensor (including, without limitation, any goodwill that may accrue by virtue of Licensor's use of the Licensed Marks). Licensor reserves any and all rights to, and shall use commercially reasonable efforts to, preserve, apply for, register, prosecute, perfect, maintain, defend, enforce or take any other action with respect to any of the Licensed Marks so as not to diminish or materially impact Licensee's rights under the License. During the Term, Licensee shall take such action (including, without limitation, the execution and delivery of additional documents and the giving of testimony) as Licensor may reasonably request to effect, perfect, evidence, register, defend or enforce any of the Licensed Marks in any jurisdiction in which Licensee is operating the Business; provided that Licensor shall reimburse any out-of-pocket expenses reasonably incurred to take any such action requested by Licensor. During the Term, Licensor, at its own cost and expense, keep Licensee updated and provide Licensee with material information concerning all filings, office actions, correspondence and other documentation filed with, sent to, issued by, or received from any trademark, patent or copyright office or any other governmental agency in connection with any such activities to the extent such matters directly affect Licensee's rights under the License. During the Term, Licensee shall not contest the validity, enforceability, registration or Licensor's ownership of any of the Licensed Marks. Licensee shall not register, make any filing with any governmental or regulatory authority with respect to, or claim an
- 3.2. Third-Party Infringement. If Licensee becomes aware of any actual or suspected infringement or other unauthorized use of any of the Licensed Marks, then Licensee shall give Licensor prompt written notice thereof including, without limitation, a copy of any documentation in the possession or control of Licensee that may evidence such infringement or other unauthorized use. Licensee shall not take any other action with respect to the alleged infringement, misappropriation or other unauthorized use of any of the Licensed Marks without the prior written consent of Licensor.

#### **Section 4. Limited Warranties and Liabilities**

**4.1. Mutual Representations and Warranties.** Licensor and Licensee each hereby represent and warrant to the other party that: (a) it has full corporate power and authority to enter into this Agreement and to carry out its obligations hereunder; (b) this Agreement has been duly authorized by all necessary action on its part; and (c) it has not entered into any agreement with or granted any rights whatsoever, or is not otherwise subject to any obligations, that would be in conflict with its obligations under this Agreement.

- **4.2. Limited Warranty.** Licensor represents and warrants to Licensee that: (a) Licensor has the right to grant the License in and to the Licensed Marks; (b) neither Licensor nor any of its Affiliates have granted to any third party any license or other right that conflicts with the License; and (c) the execution, delivery and performance of this Agreement by Licensor does not, and Licensee's exercise of its rights under the License and other provisions of this Agreement, will not, infringe the rights of any third party under any trademark, or conflict with, or result in any breach of, require consent under, any agreement between Licensor and any third party, or result in, termination, cancellation, or modification of the rights granted to Licensee under the License or the other provisions of this Agreement. The sole and exclusive remedy for any breach of this warranty is set forth in Section 5.1.
- 4.3. Disclaimer of Warranties. EXCEPT AS EXPRESSLY SET FORTH IN SECTIONS 4.1 AND 4.2, EACH PARTY HEREBY DISCLAIMS ANY AND ALL REPRESENTATIONS AND WARRANTIES, EXPRESS OR IMPLIED, INCLUDING, WITHOUT LIMITATION, ANY EXPRESS OR IMPLIED WARRANTIES WITH REGARD TO: MERCHANTABILITY OR FITNESS FOR ANY PARTICULAR PURPOSE; WHETHER ANY OF THE LICENSED MARKS APPLY OR CAN BE USED OUTSIDE OF THE APPLICABLE JURISDICTION IN WHICH LICENSOR USED SUCH LICENSE MARKS AS OF THE EFFECTIVE DATE; OR WHETHER ANY OF THE LICENSED MARKS CAN BE USED OUTSIDE OF ANY JURISDICTION IN WHICH LICENSOR USED SUCH LICENSE MARKS AS OF THE EFFECTIVE DATE WITHOUT INFRINGING ANY TRADEMARK OR OTHER INTELLECTUAL PROPERTY RIGHT OF ANY THIRD PARTY IN SUCH JURISDICTION.
- 4.4. Exclusions from Liability. EXCEPT TO PROVIDE INDEMNIFICATION UNDER SECTION 5 WITH RESPECT TO THIRD PARTY CLAIMS, AND EXCEPT IN CASES OF FRAUD OR WILLFUL MISCONDUCT BY A PARTY OR ITS AFFILIATES OR BREACH OF SECTION 7.3 OR LICENSEE'S BREACH OF SECTIONs 2 OR 3.1, IN NO EVENT SHALL EITHER PARTY OR THE AFFILIATES OF SUCH PARTY BE LIABLE TO THE OTHER FOR SPECIAL, INDIRECT, MULTIPLE, PUNITIVE, EXEMPLARY, INCIDENTAL OR CONSEQUENTIAL DAMAGES, OR FOR LOST PROFITS OR REVENUES OR INCOME, BUSINESS INTERRUPTIONS, DIMINUTION IN VALUE OR LOSS OF BUSINESS OPPORTUNITY ARISING UNDER OR IN CONNECTION WITH THIS AGREEMENT OR AS A RESULT OF ANY ACTIVITIES OR NON-PERFORMANCE HEREUNDER, REGARDLESS OF WHETHER ARISING FROM BREACH OF CONTRACT, WARRANTY, TORT, STRICT LIABILITY OR OTHERWISE, EVEN IF SUCH PARTY IS ADVISED OF THE POSSIBILITY OF SUCH LOSS OR DAMAGE OR IF SUCH LOSS OR DAMAGE COULD HAVE BEEN REASONABLY FORESEEN.

#### **Section 5. Indemnification**

**5.1. By Licensee.** Licensee hereby agrees, at all times during the Term and thereafter, to fully indemnify Licensor and its Affiliates, and any of its and their respective shareholders, owners, officers, directors, employees, agents and representatives against, and defend and hold

such Persons harmless from, any and all Losses claimed by a third party arising from or in connection with a) the gross negligence, Fraud or willful misconduct by Licensee or any of its Affiliates or any of its and their respective shareholders, owners, officers, directors, employees, agents and representatives, in each case related to the licenses granted under this Agreement; or b) any breach by Licensee of any of its covenants, agreements or obligations under this Agreement, in each case of the foregoing clauses a) and b) except to the extent of those Losses which Licensor has an obligation to indemnify pursuant to Section 5.2, as to which Losses each party shall indemnify the other to the extent of its liability for such Losses.

- **5.2.** By Licensor. Licensor hereby agrees at all times during the Term and thereafter, to fully indemnify Licensee and its Affiliates, and any of its and their respective shareholders, owners, officers, directors, employees, agents and representatives against, and defend and hold such Persons harmless from, any and all Losses claimed by a third party arising from or in connection with a) the gross negligence, Fraud or willful misconduct by Licensor or any of its Affiliates, and any of its and their respective shareholders, owners, officers, directors, employees, agents and representatives, in each case related to the licenses granted under this Agreement; or b) any breach by Licensor of any of its covenants, agreements or obligations under this Agreement, in each case of the foregoing clauses a) and b) except to the extent of those Losses which Licensee has an obligation to indemnify pursuant to Section 5.1, as to which Losses each party shall indemnify the other to the extent of its liability for such Losses.
- **5.3.** No right of indemnification shall exist under the Purchase Agreement for claims arising out of the performance of this Agreement, it being the intent of the parties that such claims shall be solely governed by the provisions of this Agreement.
- **5.4. Notice and Cooperation.** Either party (the "Indemnitee") intending to claim indemnification under this Section 5 shall notify the other party (the "Indemnitor") promptly in writing of any action, claim or liability in respect of which the Indemnitee believes it is entitled to claim indemnification, provided that the failure to give timely notice to the Indemnitor shall not release the Indemnitor from any liability to the indemnitee except to the extent the Indemnitor is actually and materially prejudiced thereby. The Indemnitor shall have the right, by notice to the indemnitee, to assume the defense of any such action or claim within the thirty (30) day period after the Indemnitor's receipt of notice of any action or claim with counsel of the Indemnitor's choice and at the sole cost of the Indemnitor. If the Indemnitor so assumes such defense, the indemnitee may participate therein through counsel of its choice, but at the sole cost of the indemnitee. The party not assuming the defense of any such claim shall reasonable assistance to the party assuming such defense, and all reasonable out-of-pocket costs of such assistance shall be for the account of the Indemnitor. No such claim shall be settled other than by the party conducting the defense thereof, and then only with the consent of the other party (not to be unreasonably withheld, delayed or conditioned); provided that the Indemnitee shall have no obligation to consent to any settlement of any such action or claim which imposes on the Indemnitee any liability or obligation that cannot be assumed and performed in full by the Indemnitor, and the Indemnitee shall have no right to withhold its consent to any settlement of any such action or claim if the settlement involves only the payment of money by the Indemnitor or its insurer and includes a full release of claims with respect to the Indemnitee.

#### **Section 6. Term and Termination**

- **6.1. Term.** The Term of this Agreement shall commence on the Effective Date and, unless sooner terminated pursuant to Section 6.2 or 6.3 shall expire on the earlier to occur of (a) the one (1) year anniversary of the Effective Date and (b) the date on which all of the inventory of packaging and labeling included in the Transferred Assets for the Product that display the Licensed Marks and have been exhausted (the "<u>Term</u>").
- **6.2. Termination.** In the event of any breach of or default under this Agreement by Licensee or any of its Affiliates which is not cured within thirty (30) days after Licensor gives Licensee written notice of the breach or default, Licensor may terminate this Agreement by giving Licensee thirty (30) days written notice of termination, provided, however, that such notice period shall not extend beyond the expiration of the Term. Further, Licensor may terminate this Agreement by giving Licensee ninety (90) days written notice of termination in the event: (a) Licensee is adjudicated as bankrupt, becomes insolvent, makes an assignment for the benefit of creditors or proposes or makes an arrangement for the liquidation of its debts; or (b) Licensee or any of its Affiliates commences, directs or controls any legal action seeking to render any of the Licensed Marks invalid or unenforceable.
  - **6.3. Termination By Licensee.** Licensee may terminate this Agreement at any time by giving Licensor written notice of termination.
- **6.4. Effect of Termination.** Upon any expiration or termination of this Agreement, the following shall apply: (a) the License and any sublicenses granted under Section 2.2 shall immediately terminate; (b) Licensee shall immediately discontinue any and all use of the Licensed Marks; (c) Sections 2.6, 3.1, 4.3, 4.4, 5, 6 and 7; and (d) any and all rights and remedies for any breach of or default under this Agreement accruing to a party prior to expiration or termination of this Agreement shall survive.

#### Section 7. Miscellaneous

7.1. Notices. All notices, requests and other communications to any party hereunder shall be in writing (including electronic transmission) and shall be given:

If to Licensor

Teva Pharmaceuticals Ltd. 425 Privet Road Horsham, PA 19044 5 Basel Street, P.O. Box

Attention: Ira Schreger, Sr. Vice President and General Counsel, Transactions

Telephone: 215-591-3000

E-mail: ira.schreger@tevapharm.com

with a copy (which shall not constitute notice) to:

Goodwin Procter LLP 100 Northern Avenue Boston, MA 02210

Attention: Kingsley Taft, H. David Henken and Gregg Katz Telephone: (617) 570-1222, (617) 570-1672 and (617) 570-1406

E-mail: ktaft@goodwinlaw.com, dhenken@goodwinlaw.com and gkatz@goodwinlaw.com

#### If to Licensee:

CooperSurgical, Inc. 75 Corporate Drive Trumbull, CT 06611

Attention: Robert D. Auerbach, M.D. FACOG, Executive Vice President and

Chief Strategy and Medical Officer and Graceann Pisano, Corporate Counsel

Telephone: 203-601-5200

E-mail: robert.auerbach@coopersurgical.com and

rachel.augustine@coopersurgical.com and graceann.pisano@coopersurgical.com

### with a copy to (which shall not constitute notice):

The Cooper Companies, Inc. 6140 Stoneridge Mall Road, Suite 590 Pleasanton, California 94588

Attention: Randal Golden, Vice President and General Counsel

Telephone: 925-460-3660 Email: rgolden@cooperco.com

and

Carter Ledyard & Milburn LLP

2 Wall Street

New York, NY 10005

Attention: David I. Karabell and Bryan Hall

Telephone: 212-732-3200

E-mail:karabell@clm.com and hall@clm.com

or such other address or e-mail address as such party may hereafter specify for the purpose by notice to the other parties hereto. All such notices, requests and other communications shall be deemed received on the date of receipt by the recipient thereof if received prior to 5:00 p.m. on a Business Day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed to have been received on the next succeeding Business Day in the place of receipt.

**7.2.** <u>Assignment</u>. Neither party may assign this Agreement or any of its rights or obligations hereunder except with the written consent of the other party. Notwithstanding the foregoing provisions of this Section 7.2, Licensee may assign this Agreement without Licensor's written consent to an Affiliate and Licensor may assign this Agreement without Licensee's prior written consent (a) in whole or on a product-by-product or product-by-country basis to any of its Affiliates; or (b) in whole to a buyer of all or substantially all of its assets to which this Agreement relates (whether by merger, stock sale, asset sale or otherwise).

#### 7.3. Confidentiality.

- a) Prior to this Agreement and during the Term, each party may have disclosed or may need to disclose or make accessible to the other (for purposes of this Section 7.3, the party disclosing or making accessible such information shall hereinafter be referred to as the "**Receiving Party**") certain information which is either non-public, confidential or proprietary in nature (collectively, the "**Information**"); provided, that for the purposes of this Agreement, Information shall not include information that (i) becomes generally available to the public other than as a result of a disclosure by the Receiving Party or its Affiliates or contractors and through no violation of this Agreement or the Purchase Agreement; (ii) the Receiving Party can establish is available or known to the Receiving Party as of the date of this Agreement without an obligation to keep such information confidential; (iii) becomes lawfully available to the Receiving Party from a source other than the Disclosing Party and through no violation of this Agreement or the Purchase Agreement, provided, that such source is not known by the Receiving Party to be bound by a confidentiality agreement with or other obligation of secrecy to the Disclosing Party; or (iv) the Receiving Party can establish is independently developed by the Receiving Party without use of, or reference to, any other Information.
- b) The Receiving Party shall hold in confidence and not disclose to any third party any Information received by it in connection with this Agreement from the Disclosing Party solely as necessary to exercise its rights or fulfill its obligations under this Agreement (and for no other purpose whatsoever) and it shall take the same degree of care with the Disclosing Party's Information as it uses with its own, but in no event less than a reasonable degree of care; provided, that Licensor may disclose the Information (i) to those of its Affiliates or to third parties to the extent necessary to exercise its rights or fulfill its obligations under this Agreement, and provided they are bound by confidentiality obligations at least as stringent as those set forth herein; and (ii) to the extent required to be disclosed by Law or the rules of an applicable securities exchange; provided further, that in the case of any disclosures made pursuant to clause (ii), to the extent legally permitted (as determined on the advice of counsel) and not otherwise prohibited by applicable Law, the Receiving Party shall, and shall direct its representatives to, promptly notify the Disclosing Party in writing so that the Disclosing Party may seek a protective order or other appropriate remedy or, in its sole discretion, waive compliance with this Section 7.3. In the event no such protective order or other remedy is obtained, or that the Disclosing Party waives compliance with this Section 7.3, the Receiving Party shall, without liability hereunder, furnish only that portion of the Information which is legally required and shall request assurance that the Information will be accorded confidential treatment.
- c) The Receiving Party shall inform any and all of its Affiliates, officers, directors, controlling persons, partners, employees, agents or representatives ("Representatives") that receive Information of the Disclosing Party of the confidential and proprietary nature of such

Information and shall direct such Representatives to treat such Information as strictly proprietary and confidential. The Receiving Party shall establish procedures to ensure that the Information is properly protected and monitored for purposes of complying with this Section 7.3. The Receiving Party agrees to be responsible for any breach of this Section 7.3 by any of its Representatives.

- **7.4.** <u>Disputes.</u> Other than with respect to any claim covered by Section 5, any dispute, controversy or claim between the parties arising out of, relating to or in connection with the this Agreement or the breach, termination, or validity thereof (a "<u>Dispute</u>"), the claiming party may provide the other party with a written notice (a "<u>Notice of Dispute</u>"), describing (a) the issues in dispute and such party's position thereon; (b) a summary of the evidence and arguments supporting such party's positions; (c) a summary of the negotiations that have taken place to date; and (d) the name and title of the senior executives (which, in the case of Licensor, will be a member of Licensor's executive committee, and, in the case of Licensee, will be a senior executive of Licensee (i.e., direct report to the Chief Executive Officer) or their respective designees who will represent each party. The senior executives or their respective designees designated in such Notice of Dispute shall meet in person or by telephone as often as reasonably necessary to resolve the Dispute and shall confer in a good faith effort to resolve the Dispute. In the event that such Dispute is not resolved by such senior executives or their respective designees within five (5) Business Days after receipt of the Notice of Dispute (or such longer time as such senior executives or their respective designees may agree), then either party may pursue any remedy at law, equity, or otherwise.
- 7.5. Governing Law. This Agreement (and any claim or controversy arising out of or relating to this Agreement) shall be governed by and construed in accordance with the laws of the State of New York without giving effect to any choice or conflict of law provision or rule that would cause the application of the laws of any jurisdiction other than the State of New York. Neither this Agreement nor any right or obligation of any of the parties under this Agreement shall be governed by the U.N. Convention on Contracts for the International Sale of Goods, and the parties expressly waive or disclaim, as the case may be, any right or obligation they may have under this Agreement pursuant to the U.N. Convention on Contracts for the International Sale of Goods.
- 7.6. <u>Jurisdiction</u>. Each party hereto hereby irrevocably and unconditionally submits, for itself and its property, to the non-exclusive jurisdiction of any New York State court, or Federal court of the United States of America, sitting in Manhattan, New York City, New York, and any appellate court from any thereof, in any action or proceeding arising out of or relating to this Agreement delivered in connection herewith or the transactions contemplated hereby or thereby or for recognition or enforcement of any judgment relating thereto, and each party hereto hereby irrevocably and unconditionally (a) agrees not to commence any such action or proceeding except in such courts; (b) agrees that any claim in respect of any such action or proceeding may be heard and determined in such New York State court or, to the extent permitted by applicable Law, in such Federal court; (c) waives, to the fullest extent it may legally and effectively do so, any objection which it may now or hereafter have to the laying of venue of any such action or proceeding in any such New York State or Federal court and (d) waives, to the fullest extent permitted by applicable Law, the defense of an inconvenient forum to the maintenance of such action or proceeding in any such New York State or Federal court. Each

party hereto agrees that a final judgment in any such action or proceeding shall be conclusive and may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by applicable Law. Each party hereto irrevocably consents to service of process in the manner provided for notices in Section 7.1. Nothing in this Agreement will affect the right of any party hereto to serve process in any other manner permitted by applicable Law.

- 7.7. Amendments. Waivers. No amendment, modification or discharge of this Agreement, and no waiver hereunder, shall be valid or binding unless set forth in writing and duly executed by the party against whom enforcement of the amendment, modification, discharge or waiver is sought. Any such waiver shall constitute a waiver only with respect to the specific matter described in such writing and shall in no way impair the rights of the party granting such waiver in any other respect or at any other time. Neither the waiver by any of the parties hereto of a breach of or a default under any of the provisions of this Agreement, nor the failure by any of the parties, on one or more occasions, to enforce any of the provisions of this Agreement or to exercise any right or privilege hereunder, shall be construed as a waiver of any other breach or default of a similar nature, or as a waiver of any of such provisions, rights or privileges hereunder. The rights and remedies herein provided are cumulative and none is exclusive of any other, or of any rights or remedies that any party may otherwise have at law or in equity.
- **7.8.** Representation by Counsel. Each party represents and agrees with each other that it has been represented by or had the opportunity to be represented by, independent counsel of its own choosing, and that it has had the full right and opportunity to consult with its respective attorney(s), that to the extent, if any, that it desired, it availed itself of this right and opportunity, that it or its authorized officers (as the case may be) have carefully read and fully understand this Agreement in their entirety and have had them fully explained to them by such party's respective counsel, that each is fully aware of the contents thereof and their meaning, intent and legal effect, and that it or its authorized officer (as the case may be) is competent to execute this Agreement and has executed this Agreement free from coercion, duress or undue influence.
  - 7.9. Headings. The headings of the sections and subsections of this Agreement are inserted for convenience only and shall not be deemed to constitute a part hereof.
- **7.10.** Counterparts. Signature Pages. The parties may execute this Agreement in one or more counterparts, each of which will be deemed an original and all of which, when taken together, will be deemed to constitute one and the same agreement. Any signature page hereto delivered by facsimile machine or by e-mail (including in portable document format (pdf), as a joint photographic experts group (jpg) file, or otherwise) shall be binding to the same extent as an original signature page, with regard to any agreement subject to the terms hereof or any amendment thereto and may be used in lieu of the original signatures for all purposes. Any party that delivers such a signature page agrees to later deliver an original counterpart to any party that requests it.
- **7.11.** Entire Agreement. Without limiting the rights and obligations of the parties under the Purchase Agreement and the other Ancillary Agreements, this Agreement and all schedules hereto and thereto, constitutes the full, complete, final and integrated agreement between the parties relating to the subject matter hereof and supersedes all previous written or oral negotiations, commitments, agreements, transactions, or understandings with respect to the subject matter hereof.

- **7.12.** Severability. If any provision, including any phrase, sentence, clause, section or subsection, of this Agreement is determined by a court of competent jurisdiction to be invalid, inoperative or unenforceable for any reason, such circumstances shall not have the effect of rendering such provision in question invalid, inoperative or unenforceable in any other case or circumstance, or of rendering any other provision herein contained invalid, inoperative or unenforceable to any extent whatsoever. Upon any such determination, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner in order that the transactions contemplated hereby be consummated as originally contemplated to the fullest extent possible.
- **7.13.** <u>Precedence.</u> In the event of a conflict between a provision of this Agreement and a provision of the Purchase Agreement or an Ancillary Agreement, or a provision of a schedule hereto, the terms of this Agreement shall take precedence.
- **7.14.** Public Statements. No party to this Agreement shall originate any publicity, news release or other similar public announcement, written or oral, whether relating to this Agreement or any documents or transactions contemplated hereby or the existence of any arrangement between the parties, without the prior written consent of the other party whether or not named in such publicity, news release or other similar public announcement, except to the extent permitted under the Purchase Agreement.
- 7.15. Interpretation. Unless expressly specified otherwise, whenever used in this Agreement, the terms "Section" and "Exhibit" refer to section and exhibits of this Agreement. Whenever used in this Agreement, the terms "hereby," "hereof," "herein" and "hereunder" and words of similar import refer to this Agreement as a whole, including all articles, sections, schedules and exhibits hereto. Whenever used in this Agreement, the terms "include," "includes" and "including" mean "include, without limitation," "includes, without limitation" and "including, without limitation," respectively. Whenever the context of this Agreement permits, the masculine, feminine or neuter gender, and the singular or plural number, are each deemed to include the others. "Days" means calendar days unless otherwise specified. References in this Agreement to a particular Law or sections of a Law shall be deemed to refer any amended or replacement legislation thereto and to such sections or provisions as they may be amended after the date of this Agreement. The parties have participated jointly in the negotiation and drafting of this Agreement and in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties and no presumption or burden of proof shall arise favoring or disfavoring any party (or any Affiliate thereof) by virtue of the authorship of any of the provisions of this Agreement.

**IN WITNESS WHEREOF**, the parties have caused this Agreement to be executed by their authorized representatives. Licensor: Licensee: TEVA WOMEN'S HEALTH, LLC COOPERSURGICAL, INC. By Ву \_\_\_\_ Name: Name: Title: Title: By Name: Title: Licensor: TEVA PHARMACEUTICALS USA, INC. Ву Name: Title: Ву Name: Title: 13

### **EXHIBIT A**

### **Licensed Marks**

TEVA wordmark
TEVA WOMEN'S HEALTH wordmark
TEVA PHARMACEUTICALS USA wordmark





TEVA WOMEN'S HEALTH ACCESS SOLUTIONS





**EXHIBIT E** 

GUARANTEE ("Guarantee"), dated as of September 11, 2017, given by The Cooper Companies, Inc., a Delaware corporation ("TCC"), in favor of Teva Pharmaceutical Industries Ltd. ("Teva").

#### **PREAMBLE**

On the date of this Guarantee, CooperSurgical, Inc., a Delaware corporation ("<u>CSI</u>"), which is a wholly-owned indirect subsidiary of TCC, entered into an Asset Purchase Agreement between CSI and Teva (in its current form and as it may be amended in accordance with its terms, the "<u>APA</u>") in which CSI agreed to purchase certain assets related to Teva's Paragard intrauterine copper contraceptive device business, on the terms and subject to the conditions set out in the APA (the "<u>Transaction</u>"). As an inducement for Teva to enter into the APA, TCC is willing to guarantee CSI's obligations under the APA on the terms and conditions set out in this Guarantee.

#### ACCORDINGLY, TCC hereby agrees as follows:

- 1. <u>Guarantee</u>. As an inducement for Teva to enter into the APA, TCC, for itself and for its successors and assigns, hereby guarantees, on the terms and conditions set out in this Guarantee, the payment and performance when due of all liabilities, covenants and agreements of CSI contained in, or arising under, the APA (the "<u>Guaranteed Obligations</u>"). If CSI fails to make any payment or perform any covenant or agreement that it is obligated to pay or perform under the Guaranteed Obligations, TCC shall promptly and fully make, or cause to be made, that payment or perform, or cause to be performed, that covenant in place of CSI.
- 2. <u>Scope</u>. This Guarantee applies to the Guaranteed Obligations, and not to any other obligations that CSI or any other Person may otherwise have to Teva. "<u>Person</u>" shall be construed broadly and means an individual or an entity of any kind.
- 3. <u>Defenses</u>. TCC shall not have any liability under this Guarantee or under the APA, under any legal theory, whether at law, in equity, in contract, in tort or otherwise, other than as explicitly set out in this Guarantee or in the APA, as applicable. TCC shall be entitled to assert, as defenses to TCC's obligations under this Guarantee, any and all defenses that CSI may have under the APA or any other agreement entered into by CSI and Teva in connection with the Transaction.
- 4. <u>Nature of Guarantee</u>. This is a guarantee of payment and performance, not of collection, and Teva shall not be required to pursue any remedies it may have against CSI under the APA.
- 5. <u>TCC Waivers</u>. Without waiving or limiting any provision of the APA, TCC hereby waives (a) notice of acceptance of this Guarantee by any Person, (b) notice of the amount of indebtedness or extent of performance owing under this Guarantee, (c) notice of demand for payment, notice of default or non-payment or non-performance, presentment, protest, notice of protest, and notice of dishonor, (d) any defense based on any change in the corporate existence, structure or ownership of TCC or CSI, or any insolvency bankruptcy, reorganization, liquidation or similar proceeding of TCC or CSI or affecting any of their assets. TCC acknowledges that it will receive substantial direct and indirect benefits from the consummation of the Transaction, and that it is knowingly granting the waivers set out in this <u>Section 5</u> in contemplation of those benefits.

- 6. <u>Representations and Warranties</u>. TCC hereby represents and warrants to Teva that TCC has duly executed and delivered this Guarantee, and that this Guarantee constitutes the valid and binding obligation of TCC, enforceable against TCC in accordance with its terms, except as enforcement may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws affecting creditors' rights generally or by general equitable principles.
- 7. Expiration. This Guarantee shall expire, and TCC shall have no further obligation under this Guarantee, on the date that is 12 months after the Closing Date (as that term is defined in the APA) (the "Expiration Date"), unless Teva asserts, on or before the Expiration Date, a claim under this Guarantee alleging that Guaranteed Obligations are due and owing from TCC (a "Claim"), in which case this Guarantee shall expire, and TCC shall have no further obligation under this Guarantee, on the later of (a) the Expiration Date and (b) the date on which that Claim is finally resolved and any Guaranteed Obligations due and owing from TCC under the final resolution of that Claim have been paid, performed or otherwise satisfied in full.
  - 8. Notices. All notices, requests and other communications to any party to this Guarantee shall be in writing (including electronic transmission) and shall be given:

#### If to TCC:

The Cooper Companies, Inc. 6140 Stoneridge Mall Road, Suite 590 Pleasanton, California 94588

Attention: Randal Golden, Vice President and General Counsel

Telephone: 925-460-3660 Email: rgolden@cooperco.com

with a copy (which shall not constitute notice) to:

Carter Ledyard & Milburn LLP 2 Wall Street New York, NY 10005 Attention: David I. Karabell and Bryan Hall

Telephone: 212-732-3200

E-mail: karabell@clm.com and hall@clm.com

#### If to Teva:

Teva Pharmaceutical Industries Ltd. 425 Privet Road Horsham, PA 19044

Attention: Ira Schreger, Sr. Vice President and General Counsel, Transactions

Telephone: 215-591-3000

E-mail: ira.schreger@tevapharm.com

with a copy (which shall not constitute notice) to:

Goodwin Procter LLP 100 Northern Avenue Boston, MA 02210

Attention: Kingsley Taft, H. David Henken and Gregg Katz Telephone: (617) 570-1222, (617) 570-1672 and (617) 570-1406

E-mail: ktaft@goodwinlaw.com, dhenken@goodwinlaw.com and gkatz@goodwinlaw.com

or such other address or e-mail address as that party may hereafter specify for the purpose by notice to the other party. All such notices, requests and other communications shall be deemed received on the date of receipt by the recipient thereof if received prior to 5:00 p.m. on a Business Day in the place of receipt. Otherwise, any such notice, request or communication shall be deemed to have been received on the next succeeding Business Day in the place of receipt. "Business Day" means any day that is not (a) a Saturday, (b) a Sunday or (c) any other day on which commercial banks are authorized or required by law to be closed in the City of New York.

- 9. <u>Governing Law</u>. This Guarantee shall be governed in all respects, including as to validity, interpretation and effect, by the laws of the State of New York, without giving effect to any choice or conflict of law provision or rule, whether in the State of New York or any other jurisdiction, that would cause the laws of any jurisdiction other than the State of New York to apply.
- 10. <u>Jurisdiction and Venue</u>. THE NEW YORK STATE AND UNITED STATES FEDERAL COURTS SITTING IN NEW YORK COUNTY, NEW YORK SHALL HAVE EXCLUSIVE JURISDICTION OVER ALL ACTIONS, SUITS AND PROCEEDINGS ARISING OUT OF OR RELATING TO THIS GUARANTEE OR ANY DEALINGS BETWEEN THE PARTIES TO THIS GUARANTEE RELATING TO THE SUBJECT MATTER OF THIS GUARANTEE, INCLUDING CONTRACT CLAIMS, TORT CLAIMS AND ALL OTHER STATUTORY AND COMMON LAW CLAIMS, AND EACH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY (A) SUBMITS, FOR ITSELF AND ITS PROPERTY, TO THE JURISDICTION OF ANY SUCH COURT IN ANY SUCH ACTION OR PROCEEDING OR FOR RECOGNITION OF ANY JUDGMENT AND (B) WAIVES ANY OBJECTION TO THE LAYING OF VENUE OF, AND ANY DEFENSE BASED ON AN INCONVENIENT FORUM IN, ANY SUCH ACTION OR PROCEEDING IN ANY SUCH COURT.
- 11. <u>Jury Trial Waiver</u>. EACH PARTY IRREVOCABLY WAIVES ALL RIGHT TO TRIAL BY JURY IN ANY ACTION, SUIT OR PROCEEDING BASED UPON OR ARISING OUT OF THIS GUARANTEE OR ANY DEALINGS BETWEEN THE PARTIES TO THIS GUARANTEE RELATING TO THE SUBJECT MATTER OF THIS GUARANTEE. THE SCOPE OF THIS WAIVER IS INTENDED TO BE ALL-ENCOMPASSING OF ANY AND ALL DISPUTES THAT MAY BE FILED IN ANY COURT AND THAT RELATE TO THE SUBJECT MATTER OF THIS GUARANTEE, INCLUDING CONTRACT CLAIMS, TORT CLAIMS AND ALL OTHER STATUTORY AND COMMON LAW CLAIMS. EACH PARTY

HAS REVIEWED THIS WAIVER WITH ITS LEGAL COUNSEL, AND KNOWINGLY AND VOLUNTARILY WAIVES ITS JURY TRIAL RIGHTS FOLLOWING CONSULTATION WITH THAT LEGAL COUNSEL. IN THE EVENT OF LITIGATION, THIS GUARANTEE MAY BE FILED AS A WRITTEN CONSENT TO A TRIAL BY THE COURT.

- 12. <u>Amendments and Waivers</u>. No amendment of or waiver of any provision of this Guarantee shall be of any force or effect unless it is in writing and signed, in the case of an amendment, by both TCC and Teva or, in the case of waiver, by the party or each of the parties against whom the waiver is to be effective.
- 13. <u>Headings</u>. The section headings contained in this Guarantee are for convenience of reference only and shall not in any way affect the meaning or interpretation of any provision of this Guarantee.
- 14. <u>Successors and Assigns</u>. Neither this Guarantee nor any rights or obligations under this Guarantee shall be assigned by either party, in whole or in part, including by operation of law, without the prior written consent of the other party, and any purported assignment without that consent shall be void and of no effect. Subject to the immediately preceding sentence, this Guarantee shall be binding on, and inure to the benefit of, the successors and permitted assigns of the parties.
  - 15. No Third Party Beneficiaries. This Guarantee shall not confer any rights or remedies other than the parties to this Guarantee and their successors and permitted assigns.
- 16. <u>Counterparts</u>; <u>Facsimile and Electronic Signatures</u>. This Guarantee may be executed in 2 or more counterparts, each of which shall be deemed to be an original and all of which together shall constitute one and the same instrument. Facsimile and electronic counterpart signatures to this Guarantee shall be valid and binding.

\* \* \*

IN WITNESS WHEREOF, TCC has executed and delivered this Guarantee as of the date first written above.

### AGREED AND ACCEPTED:

### TEVA PHARMACEUTICAL INDUSTRIES LTD.

By: /s/ Yitzhak Peterburg
Name: Yitzhak Peterburg
Title: Chief Executive Officer

By: /s/ Dror Bashan
Name: Dror Bashan

Title: SVP M&A and Global Alliances

### THE COOPER COMPANIES, INC.

By: /s/ Randal Golden

Name: Randal Golden

Title: Vice President and General Counsel





CONTACT:

Kim Duncan Vice President, Investor Relations The Cooper Companies, Inc. <u>ir@cooperco.com</u> 6140 Stoneridge Mall Road Suite 590 Pleasanton, CA 94588 925-460-3663 www.coopercos.com

## THE COOPER COMPANIES ANNOUNCES DEFINITIVE AGREEMENT TO ACQUIRE PARAGARD® IUD FROM TEVA

PLEASANTON, Calif., September 11, 2017 – The Cooper Companies, Inc. (NYSE: COO) today announced it has entered into a definitive asset purchase agreement with Teva Pharmaceutical Industries Ltd. to acquire the global rights and business of its PARAGARD® Intrauterine Device (IUD) in a \$1.1 billion cash transaction. The boards of directors of both Cooper and Teva have unanimously approved the transaction and closing is expected to occur prior to calendar year-end 2017 subject to the satisfaction of customary closing conditions including antitrust clearance.

PARAGARD is currently sold only in the U.S. and had revenues of approximately \$168 million for the trailing twelve-month period ending June 30, 2017. Excluding acquisition and integration related expenses and deal-related amortization, the transaction is expected to be accretive to Cooper's gross and operating margins, and approximately \$0.70 to \$0.75 accretive to earnings per share in year one. The acquired business includes Teva's manufacturing facility in Buffalo, NY, which exclusively produces the PARAGARD ILD.

Cooper estimates the U.S. IUD market is approximately \$1 billion with expected growth in the mid-single digits. Approximately 16% of the U.S. market is believed to be non-hormonal IUDs with PARAGARD being the only approved product in this space.

Commenting on the transaction, Robert S. Weiss, Cooper's president and chief executive officer said, "We are extremely pleased to announce this acquisition which will expand CooperSurgical's business in a large and growing segment of the contraceptive device market. CooperSurgical will now offer the only IUD on the U.S. market that is hormone-free, long-lasting, and reversible. This is also a perfect strategic fit as it leverages CooperSurgical's long-standing, strong relationships with gynecologists."

Cooper has arranged a commitment for a bridge loan of \$1.1 billion to support funding of the acquisition. Cooper's financial advisor for the transaction is Guggenheim Securities with Carter Ledyard & Milburn LLP acting as legal advisor.

### **About PARAGARD**

PARAGARD is an intrauterine device (IUD) that is inserted into a woman's uterus for long-term birth control (contraception). PARAGARD's T-shaped plastic frame has copper wire coiled around the stem and two copper sleeves along the arms that continuously release copper into the uterus. Together the plastic

T frame and copper produce a sterile inflammatory reaction in the uterus that is toxic to sperm, which helps prevent fertilization. PARAGARD is the only copper IUD available in the U.S. It is over 99% effective and is FDA approved for contraception for up to ten years after insertion. PARAGARD is regulated as a pharmaceutical device that is inserted and later removed by an in-office health care provider. The cost of the IUD is covered by insurance in most cases.

#### **About The Cooper Companies**

The Cooper Companies, Inc. ("Cooper") is a global medical device company publicly traded on the NYSE (NYSE:COO). Cooper is dedicated to being A Quality of Life Company™ with a focus on delivering shareholder value. Cooper operates through two business units, CooperVision and CooperSurgical. CooperVision brings a refreshing perspective on vision care with a commitment to developing a wide range of high-quality products for contact lens wearers and providing focused practitioner support. CooperSurgical is committed to advancing the health of families with its diversified portfolio of products and services focusing on women's health, fertility and diagnostics. Headquartered in Pleasanton, CA, Cooper has more than 10,000 employees with products sold in over 100 countries. For more information, please visit <a href="https://www.coopercos.com">www.coopercos.com</a>.

#### **About Teva**

Teva Pharmaceutical Industries Ltd. (NYSE and TASE: TEVA) is a leading global pharmaceutical company that delivers high-quality, patient-centric healthcare solutions used by approximately 200 million patients in over 60 markets every day. Headquartered in Israel, Teva is the world's largest generic medicines producer, leveraging its portfolio of more than 1,800 molecules to produce a wide range of generic products in nearly every therapeutic area. In specialty medicines, Teva has the world-leading innovative treatment for multiple sclerosis as well as late-stage development programs for other disorders of the central nervous system, including movement disorders, migraine, pain and neurodegenerative conditions, as well as a broad portfolio of respiratory products. Teva is leveraging its generics and specialty capabilities in order to seek new ways of addressing unmet patient needs by combining drug development with devices, services and technologies. Teva's net revenues in 2016 were \$21.9 billion. For more information, visit <a href="https://www.tevapharm.com">www.tevapharm.com</a>.

#### **Conference Call and Webcast**

The Company will host a conference call today at 5:00 PM ET to discuss this transaction. The live dial-in number for the call is 855-643-4430 (U.S.) / 707-294-1332 (International). The participant passcode for the call is "Cooper". A simultaneous webcast of the call will be available through the "Investor Relations" section of The Cooper Companies' website at http://investor.coopercos.com and a transcript of the call will be archived on this site for a minimum of 12 months. A recording of the call will be available beginning at 8:00 PM ET on September 11, 2017 through September 18, 2017. To hear this recording, dial 855-859-2056 (U.S.) / 404-537-3406 (International) and enter code 266737.

For presentation slides on the acquisition, visit the Investor Relations section of the Company's website at www.cooperco.com.

#### Forward-Looking Statements

This earnings release contains "forward-looking statements" as defined by the Private Securities Litigation Reform Act of 1995. Statements relating to guidance, plans, prospects, goals, strategies, future actions, events or performance and other statements that are other than statements of historical fact, including all statements regarding the acquisition of Teva's PARAGARD including financial position, market position, product development and business strategy, expected cost synergies, expected timing and benefits of the transaction, difficulties in integration entities or operations, as well as estimates of our and PARAGARD's future expenses, sales and diluted earnings per share are forward looking. In addition, all statements regarding anticipated growth in our revenue. To identify these statements look for words like "believes," "expects," "may," "will," "should," "could," "seeks," "intends," "plans," "estimates" or "anticipates" and similar words or phrases. Forward-looking statements necessarily depend on assumptions, data or methods that may be incorrect or imprecise and are subject to risks and uncertainties.

Among the factors that could cause our actual results and future actions to differ materially from those described in forward-looking statements are: adverse changes in the global or regional general business, political and economic conditions, including the impact of continuing uncertainty and instability of certain countries that could adversely affect our global markets, and the potential adverse economic impact and related uncertainty caused by these items, including but not limited to, the United Kingdom's election to withdraw from the European Union; foreign currency exchange rate and interest rate fluctuations including the risk of fluctuations in the value of foreign currencies or interest rates that would decrease our revenues and/or earnings; our indebtedness and associated interest expense could adversely affect our financial health, prevent us from fulfilling our debt obligations or limit our ability to borrow additional funds; changes in tax laws or their interpretation and changes in statutory tax rates, including but not limited to, the United States and other countries with proposed changes to tax laws, some of which may affect our taxation of earnings recognized in foreign jurisdictions and/or negatively impact our effective tax rate; acquisition-related adverse effects including the failure to successfully obtain the anticipated revenues, margins and earnings benefits of the Teva PARAGARD acquisition, integration delays or costs and the requirement to record significant adjustments to the preliminary fair value of assets acquired and liabilities assumed within the measurement period, required regulatory approvals for the Teva PARAGARD acquisition not being obtained or being delayed or subject to conditions that are not anticipated, adverse impacts of changes to accounting controls and reporting procedures, contingent liabilities or indemnification obligations, increased leverage and lack of access to available financing (including financing for the acquisition or

refinancing of debt owed by us on a timely basis and on reasonable terms); a major disruption in the operations of our manufacturing, research and development or distribution facilities, due to technological problems, including any related to integration of the Teva PARAGARD acquisition, natural disasters, system upgrades or other causes; a major disruption in the operations of our manufacturing, accounting and financial reporting, research and development or distribution facilities due to technological problems, including any related to our information systems maintenance, enhancements or new system deployments, integrations or upgrades; disruptions in supplies of raw materials, including but not limited to, components used to manufacture our silicone hydrogel lenses; new U.S. and foreign government laws and regulations, and changes in existing laws, regulations and enforcement guidance, which affect areas of our operations including, but not limited to, those affecting the health care industry including the contact lens industry and the medical device industry; compliance costs and potential liability in connection with U.S. and foreign laws and health care regulations pertaining to privacy and security of third party information, including but not limited to product recalls, warning letters, and data security breaches; legal costs, insurance expenses, settlement costs and the risk of an adverse decision, prohibitive injunction or settlement related to product liability, patent infringement or other litigation; limitations on sales following product introductions due to poor market acceptance; new competitors, product innovations or technologies, including but not limited to, technological advances by competitors, new products and patents attained by competitors, and competitors' expansion through acquisitions; reduced sales, loss of customers and costs/expenses related to recalls; failure to receive, or delays in receiving, U.S. or foreign regulatory approvals for products; failure of our customers and end users to obtain adequate coverage and reimbursement from third-party payors for our products and services; the requirement to provide for a significant liability or to write off, or accelerate depreciation on, a significant asset, including goodwill, and idle manufacturing facilities and equipment; the success of our research and development activities and other start-up projects; dilution to earnings per share from the Teva PARAGARD acquisition or other acquisitions or issuing stock; changes in accounting principles or estimates; environmental risks; and other events described in our Securities and Exchange Commission filings, including the "Business" and "Risk Factors" sections in the Company's Annual Report on Form 10-K for the fiscal year ended October 31, 2016, as such Risk Factors may be updated in quarterly filings.

We caution investors that forward-looking statements reflect our analysis only on their stated date. We disclaim any intent to update them except as required by law.

COO-G

Source: The Cooper Companies, Inc.



A Quality of Life Company™

CooperVision®





**Acquisition of PARAGARD® IUD from Teva** 

September 11, 2017



## **Forward Looking Statements**

#### Forward-Looking Statements

This earnings release contains "forward-looking statements" as defined by the Private Securities Litigation Reform Act of 1995. Statements relating to guidance, plans, prospects, goals, strategies, future actions, events or performance and other statements which are other than statements of historical fact, including our 2017 Guidance and all statements regarding acquisitions including the acquired companies' financial position, market position, product development and business strategy, expected cost synergies, expected timing and benefits of the transaction, difficulties in integrating entities or operations, as well as estimates of our and the acquired entities' future expenses, sales and diluted earnings per share are forward looking. In addition, all statements regarding anticipated growth in our revenue, anticipated effects of any product recalls, anticipated market conditions, planned product launches and expected results of operations and integration of any acquisition are forward-looking. To identify these statements look for words like "believes," "expects," "may," "will," "should," "could," "seeks," "plans," "estimates" or "anticipates" and similar words or phrases. Forward-looking statements necessarily depend on assumptions, data or methods that may be incorrect or imprecise and are subject to risks and uncertainties.

Among the factors that could cause our actual results and future actions to differ materially from those described in forward-looking statements are: adverse changes in the global or regional general business, political and economic conditions, including the impact of continuing uncertainty and instability of certain countries that could adversely affect our global markets, and the potential adverse economic impact and related uncertainty caused by these items, including but not limited to, the United Kingdom's election to withdraw from the European Union; foreign currency exchange rate and interest rate fluctuations including the risk of fluctuations in the value of foreign currencies or interest rates that would decrease our revenues and/or earnings; our indebtedness and associated interest expense could adversely affect our financial health, prevent us from fulfilling our debt obligations or limit our ability to borrow additional funds; changes in tax laws or their interpretation and changes in statutory tax rates, including but not limited to, United States and other countries with proposed changes to tax laws, some of which may affect our taxation of earnings recognized in foreign jurisdictions and/or negatively impact our effective tax rate; acquisition-related adverse effects including the failure to successfully obtain the anticipated revenues, margins and earnings benefits of acquisitions, integration delays or costs and the requirement to record significant adjustments to the preliminary fair value of assets acquired and liabilities assumed within the measurement period, required regulatory approvals for an acquisition not being obtained or being delayed or subject to conditions that are not anticipated, adverse impacts of changes to accounting controls and reporting procedures, contingent liabilities or indemnification obligations, increased leverage and lack of access to available financing (including financing for the acquisition or refinancing of debt owed by us on a timely basis and on reasonable terms); a major disruption in the operations of our manufacturing, research and development or distribution facilities, due to technological problems, including any related to integration of acquisitions, natural disasters, system upgrades or other causes; a major disruption in the operations of our manufacturing, accounting and financial reporting, research and development or distribution facilities due to technological problems, including any related to our information systems maintenance, enhancements or new system deployments, integrations or upgrades; disruptions in supplies of raw materials, including but not limited to, components used to manufacture our silicone hydrogel lenses; new U.S. and foreign government laws and regulations, and changes in existing laws, regulations and enforcement guidance, which affect areas of our operations including, but not limited to, those affecting the health care industry including the contact lens industry and the medical device industry; compliance costs and potential liability in connection with U.S. and foreign laws and health care regulations pertaining to privacy and security of third party information, including but not limited to product recalls, warning letters, and data security breaches; legal costs, insurance expenses, settlement costs and the risk of an adverse decision, prohibitive injunction or settlement related to product liability, patent infringement or other litigation; limitations on sales following product introductions due to poor market acceptance; new competitors, product innovations or technologies, including but not limited to, technological advances by competitors, new products and patents attained by competitors, and competitors' expansion through acquisitions; reduced sales, loss of customers and costs/expenses related to recalls; failure to receive, or delays in receiving, U.S. or foreign regulatory approvals for products; failure of our customers and end users to obtain adequate coverage and reimbursement from third party payors for our products and services; the requirement to provide for a significant liability or to write off, or accelerate depreciation on, a significant asset, including goodwill, and idle manufacturing facilities and equipment; the success of our research and development activities and other start-up projects; dilution to earnings per share from acquisitions or issuing stock; changes in accounting principles or estimates; environmental risks; and other events described in our Securities and Exchange Commission filings, including the "Business" and "Risk Factors" sections in the Company's Annual Report on Form 10-K for the fiscal year ended October 31, 2016, as such Risk Factors may be updated in quarterly filings.

We caution investors that forward-looking statements reflect our analysis only on their stated date. We disclaim any intent to update them except as required by law.





## **Transaction Highlights**

- Acquiring the Global Rights and Business of the PARAGARD Intrauterine Device (IUD) from Teva Pharmaceutical Industries Ltd., for \$1.1B cash
  - Transaction expected to close by calendar year end 2017<sup>1</sup>
- Delivers compelling financial benefits
  - TTM revenue of approx. \$168 million as of 6/30/17, mid-single-digit growth expected
  - Expected to be immediately accretive to gross and operating margins
  - Expected to be accretive to Non-GAAP EPS by approx. \$0.70 to \$0.75 in Year 1<sup>2</sup>
  - Strong cash flow generation
  - ROIC exceeds cost of capital in Year 1
- \$1.1 billion Bridge Loan commitment received to support financing. Permanent financing expected in the form of a long term debt facility.
- Pro-forma net debt to adjusted EBITDA expected to be approx. 2.6x at first quarter-end post closing

Note:

 $1. \quad \textit{Subject to customary closing conditions including antitrust clearance} \\$ 

Excludes acquisition and integration related expenses and deal-related amortization
 September 11, 2017

<u>Coper</u>



## Strategic Highlights

- Great market IUDs represent a large and growing segment of the contraceptive market
  - IUD use has grown considerably in recent years in the U.S. but still lags behind the rest of the world
  - The American Congress of Obstetricians and Gynecologists (ACOG) strongly supports the expansion of IUD contraceptive utilization
- Great strategic fit broadens and strengthens CSI's gynecological portfolio
  - Product sold directly to gynecologists where CSI has a long history of calling efforts with strong relationships
  - The device nature of insertion and removal fit exceptionally well with CSI's proceduresavvy sales force
  - PARAGARD's established user base in large healthcare systems creates an opportunity for system purchases across CSI's portfolio
  - Premier, high volume product will enhance cross-selling of other CSI products in the gynecologist office

Source: Management estimates

September 11, 2017





### **PARAGARD Overview**

- PARAGARD is the only hormone-free, long lasting, reversible contraceptive option available in the U.S.
  - FDA approved in 1984, 10 years of contraceptive use, over 99% effective
  - 100% of sales in U.S. (acquiring global rights and will evaluate OUS launch)
  - Established U.S. market share of approx. 16%
- Operational/Other
  - Single manufacturing facility in Buffalo, NY dedicated to making PARAGARD
  - Distributes primarily through one distributor
  - Approx. 45 employees (2/3 commercial, 1/3 manufacturing)
  - Company website: paragard.com



### **IUD Market Overview**

### > Total U.S. IUD Market: approx. \$1 billion

- Non-Hormonal: approx. 16% of market (PARAGARD is the only option available, 10-year indication)
- Hormonal: approx. 84% of market
  - Bayer: market leader with approx. 97% share
    - Mirena: FDA approved in 2000 (5-year indication)
  - Allergan: approx. 3% share, FDA approval of *Liletta* in 2015 (4-year indication)

### Comparison of Non-Hormonal and Hormonal

 Advantages of PARAGARD (non-hormonal, 10-year indication for use) offset by hormonal IUDs reducing or eliminating menstrual cycles



### IUD Penetration and Growth Rates

- Estimated at 10-12% penetration in U.S. (Europe: 15%, Asia: 30%)
- IUD units growing 3-4% per year in U.S.

©öper



# Regulatory and Potential Future Competitive Product

- ➤ IUD's are regulated as a pharmaceutical product in the U.S. and thus must go through a much more onerous approval process than devices
  - FDA guidance on approval study typically requires (1) 10,000 women months of use for one year approval and (2) 200 women per year of claimed use
- Potential future competitive product Copper IUD¹
  - IND holder: FHI 360
  - Status recruiting participants for a phase III study
    - Estimated study completion date November 2021
    - Initial end-point of study is contraceptive effect at 3-years
      - Upon completion of study, would be submitted to FDA in an application for a 3-year indication of use (PARAGARD has 10-year indication of use)
      - 200 Patients will need to be followed for each additional year to support efficacy beyond the initial 3-years of the study

Note:

Information from Clinical Trials.gov search for copper IUD September 11, 2017

September 11, 2017





## **ACA** and Contraceptives

### Private Insurance

- Must cover at least one type of all 18 FDA-approved contraceptive methods for women as prescribed without cost sharing
  - Must cover the copper IUD (PARAGARD) and at least one hormonal IUD at no cost to policy holders

### Medicaid

- Federal law requires Medicaid programs to cover family planning services/supplies without cost sharing
  - Women who qualify for Medicaid under the ACA's expansion of the program must receive the same coverage as Private Insurance

### Uninsured

 The federal Title X National Family Planning Program funds a network of clinics to provide family planning care to low-income and uninsured women at reduced or no cost



